Single-Stage Three-Phase Differential Inverter

Utilizing Single-Ended Primary-Inductor

Converters

By

Ahmed Shawky Mohamed Hussein



A thesis submitted for the degree of Doctor of Philosophy Graduate School of Engineering Department of Electrical and Mechanical Engineering Nagoya Institute of Technology Nagoya-Japan March 2021

Single-Stage Three-Phase Differential Inverter Utilizing Single-Ended Primary-Inductor Converters

A thesis submitted to Graduate School of Engineering Department of Electrical and Mechanical Engineering Nagoya Institute of Technology For the degree of Doctor of Philosophy

By

Ahmed Shawky Mohamed Hussein

Supervisor

Prof. Takaharu Takeshita

Department of Electrical and Mechanical Engineering Nagoya Institute of Technology

To Whom IT MAY CONCERN

We hereby certify that this is a typical copy of the original doctor thesis of Mr. Ahmed Shawky Mohamed Hussein

Thesis Title:

Single-Stage Three-Phase Differential Inverter Utilizing Single-Ended Primary-Inductor Converters

(SEPIC コンバータを用いたシングルステージ三相差動インバータ)

Official seal of the institute

Declaration of Originality

I, at this moment, certify that the final developed research reported in this thesis and the thesis itself was composed and originated entirely by myself in the Graduate School of Engineering, Department of Electrical and Mechanical Engineering at the Nagoya Institute of Technology.

Ahmed Shawky Mohamed Hussein

Dedication

To my beloved father, To my dear mother, To my sweetheart wife, Omnia, To my darlings, Hana and Youssef, To my beloved sister, Heba, To my beloved Brothers, Hussein and Mohamed, To the spirit of my dear brother Abdel-Rahman, To all members of my family, Those people's love is always the source of my motivation.

Abstract

In recent years, the world is moving toward Renewable Energy Sources (RES) to produce clean and efficient energy and reduce global warming introduced by CO2 emissions. RES provides power to vast numbers of residential, commercial, and industrial customers using the utility grid's transmission lines. Among many exciting RES types, Photovoltaic (PV) has proven its competencies for many fundamental reasons. Grid-connected applications are becoming more popular due to fast PV penetration and new architectures, such as PV module architectures. This success has been stimulated by a considerable cut in PV installed price of PV modules, storage batteries, inverters, and other controllers.

DC-AC inverters are among the most critical parts of the PV module architectures. They are vital players for successful operation by converting the Direct Current (DC) power of the PV to Alternating Current (AC) power needed by the grid. Many categories of inverters with different ratings were proposed in the last three decades. Single-phase inverters are conventional in many applications, such as centralized, string, and micro-inverters. However, this solution has many severe problems, such as the double-line frequency component and its electrolyte capacitors. This issue disturbs the normal connection between the PV modules and the grid. Nowadays, three-phase micro-inverters are the best topology to solve double-line frequency. The balance between DC and AC power exists, and no electrolyte capacitors are required.

This project surveys many three-phase inverters topologies for grid-connected applications. Between dozens, single-stage differential-based inverters have been selected. Differential inverters replace every leg of traditional VSI inverter with a DC-DC converter. This configuration gives many properties such as single-stage, modularity, and isolation using isolated DC-DC converters. One DC-DC converter's connection per phase provides the inverter's uniform operation and enhances power density. It facilitates the power extension (scalability) by adding parallel converters without adding control complexity. They have step-up/down voltage gain with bi-directional power ability by utilizing the appropriate DC-DC converters.

This project contemplates a single-stage three-phase isolated differential inverter utilizing Single-Ended Primary-Inductor Converters (SEPIC). The SEPIC includes two power

switches, one inductor, a single High-Frequency Transformer (HFT), and two small Film capacitors. This low number of components produces a compacted and cheap inverter. According to differential characteristics, each SEPIC converter possesses one-third of the total power. Therefore, small components with low power ratings are utilized. The SEPIC converter has the lowest energy storage of passive elements. Finally, it provides voltage isolation and continuous input current with no extra components.

The project develops a simple control strategy for injecting active power at the utility grid. It is depend on the d-q synchronous frame and achieves little Total Harmonic Distortion (THD). The Negative Sequence Harmonic Component (NSHC) is a common issue in differential inverters and generating from the circulating currents between the DC-DC converters. That is because the alternating output voltage is greater than the input DC voltage. A NSHC detection circuit has been proposed by adding a three-stage Low Pass Filters (LPF) to mitigate this issue. The proposed controller's feasibility with NSHC compensation is proven by using a real-time Power Simulator (PSIM). Finally, a simple DSP board has been programmed for the proposed control because it has a small computational burden.

The project also develops a three-phase Modular Differential Inverter (MDI) by utilizing three SEPIC converters per phase. It processes a triple power by using identical nine SEPIC converters. A mismatch of $\pm 20\%$ in each SEPIC converter parameters has been investigated to confirm the inverter performance and its modular feature. The proposed MDI utilizes the proposed control strategy without extra circuits. Finally, the proposed inverter utilizes Silicon Carbide (SiC) MOSFET devices for better operation at a high switching frequency. A laboratory prototype validates the proposed inverter with 200V grid-side voltage, 100V DC, 1.6kW, and 50 kHz switching frequency.

Acknowledgements

In the name of Almighty ALLAH, the Most Gracious and The Most Merciful. I would like to explicit my profound thankfulness and recognition to ALLAH for granting me the blessings to do this project. With the help and permission of ALLAH, I achieved many things, including this thesis. Indeed, without the relief of ALLAH, nothing is carried out. Many most profound blessings for Prophet Mohammed (Peace be upon him) because he guides us for ALLAH. We ask ALLAH to help us to follow his message and get forgiveness in the hereafter.

I would like to express my deep respect to my advisor, Prof. Takaharu Takeshita. For more than three years, he continuously supports me in promoting inflectional research in a focused way. It was a real pleasure for me to have such a motivated and authentic person as an advisor. He makes me more confident by telling me that it is my work, and I have to lead everything in my own thesis. His office door was always open, and I will never forget the hours of deductive discussion we had throughout three years.

The second blessing is for my present and old colleagues in Takeshita-Kitagawa Laboratory for their help in providing an excellent research environment. These blessings go first to Dr. Suzuki Kazuma for helping me in my research with valuable discussion in addition to my daily life in Japan. Throughout my Ph.D. work, I had the opportunity to collaborate with others, such as Nishime Shinichi, Hamaguchi Takumi, Oshima Takuya, Matsura Ryohei, Muku Koki, Motoyama Hiromasa, Hirose Naoki, Fukuzaki Tomoyasu, and Mitani Kohei for helping me in research and experimental step-up of my circuit. Also, I cannot forget Yamanaka Suguru, Takei Kenta, Budo Kohei, and Song Chen for many helpful things in research and daily life issues. Moreover, Special thanks to my Egyptian colleagues at Nagoya Institute of Technology, Dr. Mahmoud Abdallah, Hamdy Radwan, and Ahmed Ismail, for their intensive help in research and daily life. Furthermore, Special thanks to Prof. Kitagawa Wataru and Tamano Fumiyo for helping me many times.

In contrast, I want to thank my professors and colleagues at my home university (Aswan University). At the risk of not remembering some of them, I would like to especially thank Prof. Mohamed Orabi, Dr. Emad Mohamed Ahmed, Dr. Mahrous Ahmed, Prof. El-Sayed Hasaneen, Dr. Mohamed Ali, Dr. Mohamed Saad, Dr. Yasser Nour, Dr. Mohtar Ali, Dr.

Eltaib Abdeen, Youssef kandeel, Mahmoud Nassary, Khaled Ahmed, Ahmed Hassan, Ahmed Shier, Amjad Samir, Mohamed Sediek, Fatma Hilmy, Eman Gomaa, Ali Selim, Ahmed Salem, Abulkasim Bakeer, Mohamed Zaery and Tarek Sayed for their friendship, support, and deep intellectual conversations that influenced my ideas and contributed to my research journey.

I would like to express my sincere blessing to the Muslim community in Nagoya, especially El-Shiekh Ahmed Zaid, Dr. Moaz Elsayed Elshir, Ismail Haque, Muhanad Khalid, Dr. Ashraf Abdel Haleem, and Obaidullah Safie for their valuable discussion during praying times at the Islamic Center of Nagoya, Japan.

There is someone who always stood by me, who still lionized my victories and heartened me through hard times, someone whose encouragement, love, and sacrifice were everlasting. Thank you, my wife, Omnia, from the basement of my heart.

I would like to thank my parents, brothers, and sister for all of their unconditional love and support during the long years of my learning.

Finally, and the most importantly, a deep appreciation for my little daughter Hana and my darling son Youssef. They brought many happy moments for my spirit and gave a different meaning to my message in this life.

Ahmed Shawky Mohamed Hussein 15-12-2020 Nagoya, Japan

Contents

Abstract	i
Acknowledgments	iii
Contents	v
List of Figures	viii
List of Tables	xii
Acronyms and abbreviations	xiii

1	Introduction and Literature Review	1
	1.1 Overview	1
	1.2 Traditional and Recent PV Architectures	4
	1.3 Micro-inverter topologies	8
	1.3.1 Double-line Frequency component	9
	1.3.2 Limited power scale	9
	1.3.3. Why Three-Phase Micro-inverters	10
	1.4 Standards, and requirements of three-phase micro-inverters	11
	1.4.1 Required Features of PV Modules (DC Source)	12
	1.4.2 Required Features of Grid Side	13
	1.4.3 Required Internal Features of Three-Phase Micro-inverters	14
	1.5 Thesis Objectives	17
	1.6 Thesis Outline	18
2	Classification of Three-phase Micro-Inverter Topologies	20
	2.1 Introduction	20
	2.2 Two-Stage VSI inverters	22
	2.3 Single-Stage Z Source Inverters	24
	2.4 Single-Stage Switched-Capacitor Inverters	29
	2.5 Single-Stage Differential Inverters	30
	2.6 Summary of the proposed classification	34
3	Proposed SEPIC-Based Differential Inverter	37
	3.1 Introduction	37

	3.2 Proposed SEPIC-Based Differential Inverter (SEPIC-BDI)	38
	3.2.1 Proposed PWM modulation Scheme of SEPIC-BDI	39
	3.2.2 Operational Principals of Proposed SEPIC Converters in	42
	Differential Inverter	
	3.2.3 Mathematical Analysis of Proposed SEPIC-BDI	45
	3.3 Proposed Mathematical Model for NSHC of Differential Inverters	47
	3.4 Design Example of the Proposed SEPIC-BDI	52
	3.4.1 The Parameters Selection and the Design Criteria	53
	3.4.2 Small-Signal Model of the Proposed SEPIC Converters	58
	3.4.3 Proposed Control Technique	61
	3.5 Results of the Proposed SEPIC-BDI	63
	3.5.1 Simulation Results	63
	3.5.2 Experimental Results	65
4	SEPIC-BDI: Improved Control with High Performance	70
	4.1 Introduction	70
	4.2 Proposed Control Technique (Control-I)	71
	4.3 System Results of Proposed Control Technique (Control-I)	74
	4.4 Proposed Control Technique (Control-II)	80
	4.5 System Results of Proposed Control Technique (Control-II)	81
	4.6 Loss Analysis of the Proposed Inverter	84
	4.6.1 MOSFET Devices	84
	4.6.2 Input Inductor	86
	4.6.3 HFT	86
	4.6.4 Coupling and Output Capacitors	87
	4.7 Efficiency Profile of the Proposed Inverter	87
5	Madular Differential Invertor: Analysis and Experimentation	80
5	5.1 Introduction	80
	5.2 Proposed Modular Differential Inverter MDI	07 80
	5.2 Proposed Modular Differential Inverter MDI	07
	5.4 Design Example of MDI based on Proposed SEDIC Modules	74 06
	5.4 1 Simulation Desults	90 07
	5.4.2 Experimental Popults	7/ 00
	5.5 Discussion Comparison and Evaluation	77 102
	Jiscussion, Comparison and Evaluation	103

6	Conclusions	107
	6.1 Contribution of This Work	107
	6.2 Suggestions for Future Work	109
Re	ferences	110
Lis	t of publications	125

	٠	٠	
V	1	1	

List of Figures

- 1.1: World net electricity generation (TWh) and its expectation until 2040.
- **1.2:** Percentage of RES share at Japan using many types from 2014 to 2018.
- 1.3: Price of residential grid-connected PV systems of 2000-2020.
- 1.4: Traditional PV system architectures.
- 1.5: Advanced PV module architectures.
- **1.6:** AC smart PV module.
- **1.7:** Comparison between single-phase micro-inverter with decoupling leg and three-phase micro-inverter.
- **1.8:** Possible implementations of three-phase inverters.
- **1.9:** The critical parts of the three-phase inverters and their detailed features.
- 1.10: Objectives of the Thesis.
- 2.1: Three-phase VSI equipped with LFT for grid-connected applications.
- 2.2: Schematic power circuit of three-phase CSI.
- 2.3: Three-phase two-stage VSI proposed in [67].
- 2.4: Three-phase two-stage VSI proposed in [68].
- 2.5: Three-phase two-stage VSI proposed in [70].
- 2.6: Component counts of three-phase two-stage VSI inverter topologies.
- 2.7: Multi-input three-phase two-stage VSI [71].
- **2.8:** Classical three-phase Z source inverter.
- 2.9: Structure of three-phase quasi Z-source invertor (ZSI) topology.
- 2.10: Power circuit of L-ZSI [77].
- 2.11: Power circuit of trans-ZSI [78].
- 2.12: Power circuit of voltage fed isolated ZSI [79].
- 2.13: Power circuit of three-phase four-leg Quasi ZSI [80].
- 2.14: Analysis of components' count utilized in three-phase Z-source micro-inverter topologies.
- 2.15: Switched-Capacitor based Inverter proposed in [81].
- 2.16: Switched-Capacitor based multi-level Inverter proposed in [82].

- 2.17: Switched-Capacitor-inductor Inverter proposed in [83].
- 2.18: Schematic power circuit of three-phase differential inverter topology.
- 2.19: Non-isolated three-phase differential inverter topologies.
- 2.20: Isolated three-phase differential inverter topologies.
- 2.21: Components count utilized in three-phase differential inverters topologies [85–90].
- 3.1: Proposed single-stage three-phase grid-connected isolated SEPIC-BDI.
- 3.2: Continuous Modulation Scheme CMS of differential inverters.
- 3.3: Duty cycle calculation at CMS and MCMS schemes for differential inverters.
- **3.4:** Operational waveforms of the proposed single-stage three-phase isolated grid-tied SEPIC-BDI.
- **3-5:** Schematic of proposed SEPIC converter and its operational states for line and switching-frequency.
- **3.6:** Mathematical representation of duty cycle, output voltage, output current and input current of the SEPIC-BDI.
- 3.7: PCP board implementation of one SEPIC converter utilized at the proposed SEPIC-BDI.
- 3.8: Photographs of implemented passive elements of proposed SEPIC-BDI.
- 3-9: AC equivalent circuit of the SEPIC converter utilized at proposed SEPIC-BDI.
- **3.10:** Bode plot of the open-loop transfer function of the SEPIC converter at different duty cycles ($d_x = 0.1 0.8$ from bottom to above).
- 3.11: Proposed control of the SEPIC-BDI.
- 3.12: Simulation results of grid phase voltage, output grid current and control parameters.
- 3.13: Simulation results of duty cycle, input inductor currents and output voltages.
- 3.14: Simulation results of coupling capacitor voltage, HFT current and switches currents.
- 3.15: Experimental Setup of the proposed single-stage SEPIC-BDI.
- **3.16:** Experimental results of grid voltage, output grid current and output voltages of SEPIC-BDI.
- 3.17: Experimental results of input currents, DC voltage and DC current of SEPIC-BDI.
- **3.18:** FFT Harmonic Spectrum of Grid current before and after NSHC for proposed SEPIC-BDI.

- 4.1: Open Loop transfer function of the bode blot of isolated SEPIC-BDI.
- **4.2:** Proposed control diagram (Control-I) with NSHC compensation circuit of isolated SEPIC-BDI.
- 4.3: Bode diagram of closed-loop of modified grid-current control (Control-I).
- **4.4:** Simulation results of proposed SEPIC-BDI at case 1.
- 4.5: Experimental results of proposed SEPIC-BDI at case 1.
- 4.6: FFT Harmonic Spectrum of grid-current at case 1.
- 4.7: IEC-61000 standards compared with FFT Spectrum of experimental grid-current at case 1.
- **4.8:** Simulation results of proposed SEPIC-BDI at case 2.
- **4.9:** Experimental results of proposed SEPIC-BDI at case 2.
- **4.10:** FFT Harmonic Spectrum of grid-current at case 2.
- 4.11: IEC-61000 standards compared with FFT Spectrum of experimental grid-current at case 2.
- **4.12:** Proposed control diagram (Control-II) with NSHC compensation circuit of isolated SEPIC-BDI.
- 4.13: Experimental results of proposed differential inverter.
- 4.14: Duty cycle and Static gain M of proposed differential inverter.
- 4.15: Actual and reference d-q values of proposed differential inverter.
- **4.16:** FFT analysis of the differential inverter's injected grid-current at different input DC voltages.
- 4.17: THD analysis of differential inverter's injected grid-current for internal power range.
- **4.18:** Efficiency of the SEPIC-BDI at different output power and Vdc = 100V.
- 5.1: Main Circuit of proposed single-stage three-phase MDI.
- 5.2: Voltage gain of boost and buck-boost Modules and MDI static gain at duty cycle range.
- 5.3: Duty cycle, output voltage, voltage and current stress of boost and buck-boost DC-DC Modules at one cycle of line-frequency.
- 5.4: Input current of boost and buck-boost Modules at differential inverter on one cycle of linefrequency.
- 5.5: Flowchart of the design steps for the proposed SEPIC module utilized in MDI.
- 5.6: Grid voltage, duty cycles, injected grid-current, reference and actual current of MDI.

- 5.7: Duty cycles, output voltages, input current of SEPIC modules in phase u with $\pm 20\%$.
- **5.8:** Zoomed View of Fig. 5.7 at high duty cycle of SEPIC modules connected in phase u.
- **5.9:** Experimental voltage and current stress of utilized MOSFET switches at proposed differential inverter.
- **5.10:** Experimental voltage and current stress of utilized inductor and HFT at proposed differential inverter.

List of Tables

- 2.1: Performance characteristics of two-stage VSI topologies.
- 2.2: Performance comparison for selected topologies of Z-source micro-inverters.
- **2.3:** Performance comparison of different DC-DC converters utilized at differential inverters.
- 2.4: Performance voltage stress for selected topologies of differential inverters.
- **3.1:** Voltage gain of different basic DC-DC converters.
- **3.2:** Isolated SEPIC converter components with its operational voltages, current and ripple components.
- 3.3: The designated parameters of the proposed isolated SEPIC-BDI.
- 3.4: Experimental Parameters of proposed Isolated SEPIC-BDI.
- **3.5:** Coefficients of the open-loop transfer function of SEPIC converter utilized at proposed SEPIC-BDI.
- 4.1: Low-order harmonic components percentages of proposed inverter at case 1.
- 4.2: Low-order harmonic components percentages of proposed inverter at case 2.
- 4.3: Control parameters for the proposed Control-II.
- 4.4: Calculated average voltages and currents of the SEPIC converter.
- 4.5: Calculated Power loss distribution of one SEPIC converter at Proposed Inverter topology.
- 5.1: Circuit parameters of MDI.
- **5.2:** The maximum mismatch in PSIM at different mismatch cases at $d_u = 0.8$.
- 5.3: Comparison between proposed inverter and similar topologies.
- 5.4: Comparison between proposed inverter and similar control techniques.
- 5.5: Comparison between the three proposed control techniques.

Acronyms and Abbreviations

DC	Direct Current
AC	Alternating Current
RMS	Root-Mean Square
EIA	Energy Information Association
TWH	Tera Watt Hour
Co2	Carbon Dioxide
RES	Renewable Energy Sources
PV	Photo Voltaic
EPIA	European Photovoltaic Industry Association
VRE	Variable Renewable Energy
IOT	Internet of Things
EV	Electric Vehicles
DG	Distributed Generation
MW	Mega Watt
KW	Kilo Watt
VSI	Voltage Source Inverter
MPP	Maximum Power Point
MPPT	Maximum Power Point Tracking
DMPPT	Distributed Maximum Power Point Tracking
MIC	Module Integrated Converter
MII	Module Integrated Inverter
CMV	Common Mode Voltage
EMI	Electro-Magnetic Interference
HFT	High Frequency Transformer
LFT	Low Frequency Transformer
THD	Total Harmonic Distortion
SiC	Silicon-Carbide
GaN	Gallium-Nitride
ZVS	Zero-Voltage Switching
ZCS	Zero-Current Switching
ZVZCS	Zero-Voltage Zero-Current Switching
PWM	Pulse Width Modulation
SPWM	Sinusoidal Pulse Width Modulation

ZSI	Z-Source Inverter
ST	Shoot Through
SCVD	Switched-Capacitor Voltage-Doubler
ANPC	Active-Neutral-Point-Clamped
SEPIC	Single-Ended Primary-Inductor Converter
NSHC	Negative Sequence Harmonic Component
CMS	Continuous Modulation Scheme
DMC	Discontinuous Modulation Scheme
MCMS	Modified Continuous Modulation Scheme
POL	Point-of-Load Converter
PFC	Power Factor Correction
PI	Proportional Integral
PID	Proportional Integral Derivative
PR	Proportional Resonance
LPF	Low Pass Filter
BPF	Band Pass Filter
PCB	Printed Circuit Board
PSIM	Power Simulator
RHP	Right-Half Plane
LHP	Left-Half Plane
MDI	Modular Differential Inverter
IPOP	Input-Parallel Output-Parallel
DSP	Digital Signal Processing
FFT	Fast Fourier Transform
UPS	Uninterruptible Power Supply

Chapter 1: Introduction and Literature Review

1.1 Overview

Many technological applications have been introduced and enhanced in daily world environments, from Nano-technologies to massive industrial technologies to simplify human life. These applications are very hungry for energy, especially the electrical energy type, which is considered the easy-facilitated source of current world energy. According to international energy reports, specifically from Energy Information Administration (EIA), as depicted in Fig. 1.1, the world net electricity generation increases from 21.5 to 26.3 TWh between 2012 and 2020. It will increase to 69% by 2040 due to the fastest consumption of many end-users. Also, it considered the steadily grown energy demand for many decades [1].

Historical and current electricity production systems are predominated by fossil fuels such as coal, oil, and gas in addition to nuclear power, as illustrated in Fig. 1.1. Unfortunately, fossil fuels are running out very fast due to the excessive use of several societies [2]. Also, the produced carbon dioxide (CO2) introduced by fossil fuel consumption presents a high percentage of greenhouse gas emissions and have a severe environmental impact. It increases the earth's temperature and imposes many other challenges, which are considered essential in the debate about world climate change [3]. Nuclear power is hazardous and needs a very high level of security. This oversized issue interprets the noticeable world transition from this type of energy to alternatives having a low risk.



Figure 1.1: World net electricity generation (TWh) and its expectation until 2040.

Although the energy sources mentioned above have been used for a long time, previously adverse effects need a significant and concerted transition to a promising alternative. Political and economic communities all over the world highlight and follow this trend. Renewable Energy Sources (RES) are considered an excellent alternative energy sources, as feasible in Fig. 1.1, because of its unlimited resourcing and being eco-friendly [4]. RES sources include biomass, wind, solar, tidal, and geothermal. Although RES development is still expensive, its electricity production has increased rapidly. The maximum global capacity for RES-based power plants, residential plants, and other installations raised from 1223 GW to 2532 GW between 2010 and 2019 [5]. It is foreseen that the generation capacity estimation of RES will rise to 60% by 2040 per global energy outlook developed by the European Photovoltaic Industry Association (EPIA) [5].

The RES share in developed countries like Japan is fascinating. Fig. 1.2 illustrates the total RES generation, including self-consumption, using different types in the period 2014:2018. Between 2017 and 2018, it boosted from 16.4% to 17.4%. The japan policy aims to increase this rate to 24% by 2030 [6]. That is why fossil fuel percentage has decreased from 81% to 78% in the same period, which is still high and needs consequential decisions. Solar Photovoltaic (PV) power generation has grown three times, specifically from 1.8% to 6.5%. Furthermore, Variable Renewable Energy (VRE), comprising Solar PV and wind power, stepped up to 7.2% in 2018, with only 6.3% in 2017. It is worth noticing that biomass and wind power also improved electricity generation by 2.2% and 0.7%, respectively.

On the other hand, RES types such as Hydro and geothermal power still have constant power generation capacity. Hydropower is restricted now and reached its saturation point because of the limited locations on rivers and waterfalls to build new stations and generate more power. Furthermore, building additional dams harms climate change all over the world.

By area, 30% of Japanese RES-based generation capacity comes from Tohoku Electric Power and Hokuriku Electric Power. In addition to Shikoku electric power, Kyushu Electric Power raised the VRE portion to 12% in 2018. In the whole world, the RES share of power generation has risen annually by more than 30% in many developed countries, especially in Europe. For the USA, RES generation capacity exceeded 11% in 2018.

Finally, Germany targeting one-hundred percent of its electricity generation from RES in 2030[7].



Figure 1.2: Percentage of RES share at Japan using many types from 2014 to 2018.

It is worth noticing that, among many types of RES, PV solar power units are earning more concern due to (a) clean and pollution-free technology, (b) great return on investment, and (c) abundant availability. Its promising features have augmented, in these days, by extensive research efforts that decrease components' cost, maximize power conversion efficiency, and boost overall reliability and stability. Moreover, the research efforts drive the strength of the market and industry. The projection of world electricity generation based on PV systems will increase to 300% in the next 20 years, according to the EPIA outlook 2017 [5]. Moreover, globally installed PV capacity increased by 16.6 GW in 2010 with a 131% increase from the year prior. Nearly seven times the amount (2.4 GW) was established in 2007 [8]. According to the EPIA, the leading country of cumulative installed PV capacity is Germany [5].

All this success is stimulated by a considerable cut in PV installed price of PV modules, storage batteries, inverters, and other controllers [9]. The price of installed residential and commercial PV systems decreased from 12 to $4.3 \notin$ /W during 2000–2014. The PV module prices fell from 4 to $1.02 \notin$ /W for the same period and will decrease to about 0.03\$/W in the next five years [7]. Storage elements that are incorporated in standalone systems improved the overall system performance by solving unpredictable outputs of PV sources [8–10]. The current edge applications, like the Internet of Things (IoT), distributed generations (DGs), and electric vehicles (EVs), force more improvements and cause

rising concerns [11, 14].

On the other hand, the grid-connected PV architectures are more convenient than standalone systems due to cost reduction and higher capacity, as shown in Fig. 1.3. Compared to standalone systems, the residential systems' price decreased to 9, 8.5, and 8.5 (W for the USA, Japan, and Germany. The storage feature, with its additive cost, is not required [15]. In 2018, the price decreased to unprecedented values less than 4 (W. The lowest price is for Germany with 1.2 (W. Moreover, it was 2.3 and 3.3 (W for Japan and the USA, respectively. The noticeable higher prices for PV systems in Japan and the USA in the 2007:2010 and 2014:2016 periods respectively result from local currency exchange, not the advancements in PV systems. For example, at this time, the federal tax of USA customers compounded to 30%. Fortunately, after this period, the PV prices went down again as it has for a long time. Motivated by this, much research has sophisticated to decrease prices by investigating and discussing grid-connected PV architectures' merits, especially the crucial part, the inverter [16].



Figure 1.3: Price of residential grid-connected PV systems of 2000-2020.

1.2 Traditional and Recent PV Architectures

PV's power plants provide power to vast numbers of residential, commercial, and industrial customers by using the utility grid. Up to 2009, most grid-connected PV architectures have built at a small level. They have linked to the distribution utility grid. The established PV system in Florida connected to transmission level (230 kV) draws the world's attention to a higher power scale [17]. This PV architecture is known as

centralized PV architecture, as shown in Fig. 1.4-a. The PV modules are linked in series to construct a PV string with sufficient output voltage. Many PV strings are paralleled to attain the desired power levels. A DC-AC centralized inverter is linked to transform PV DC power to appropriate AC power. This inverter rating in the Mega Watt (MW) scale and uses a three-phase grid. Moreover, classical Voltage Source Inverter (VSI) topologies were used for many years in this architecture because the voltage gain is unnecessary [18]. However, the PV array's partial shading conditions disturb the Maximum Power Point Tracking (MPPT) control and diminish the total DC voltage for the PV architecture. Nonregular disturbance at tracking the actual Maximum Power Point (MPP) is possible when partially shadowing is existed. It is introducing different local MPPs and mismatches between PV modules [19]. This architecture also needs high-voltage DC cables among the modules and the inverter, which can undergo a single failure point and make the whole PV system unable to operate. Also, this requires string diodes to mitigate circulating current between strings. The reduced DC voltage was solved by integrating a DC-DC boost converter among the centralized inverter and PV to step-up the voltage. However, the other mentioned drawbacks still existed and may violate the performance of this architecture.

String architectures utilize a Distributed Maximum Power Point Tracking (DMPPT) concept to enhance overall performance. This concept was performed by separating the PV array into different strings. It effectively deviated the string diodes. An individual MPPT controller and inverter are then applied to every string, as shown in Fig. 1.4-b. Therefore, partial shading and module mismatching problems were significantly reduced in this configuration [19].



Figure 1.4: Traditional PV system architectures.

By doing this, the PV architecture efficiency increases about 1–3% larger than the central inverter [20] because the losses generated from string diodes and string mismatch losses have mitigated. Commonly, the power capacity for one string is 5kW. The modularity of the system facilitates system expansion by adding more strings. Moreover, inverter per string enhance system reliability and increase flexibility. However, the total produced mismatch losses between PV modules at every string is not critically detected. Therefore, they cannot be removed in this PV architecture [16–20].

In addition to partial shading conditions and introduced mismatch losses in traditional architectures, its purpose is limited to energy harvesting's high power scale. Also, it needs customers to live close to centralized power plants to save power transmission costs. It adds new challenges because some customers live far away, which needs long transmission lines and introduces high power loss. In contrast, the DG system has smaller, decentralized sources that generate electricity much closer to those who use it. There are many producers, and even though they produce less individually (kW scale), they are all connected to the grid. Several technologies form the backbone of a DG system [21].

As the PV generation efficiency and power quality are a fundamental issue for PV architectures, many trials and improvements have been made to remove traditional architectures' main drawbacks. Therefore, the PV module architecture, which has recently been proposed, is the optimal solution. It completely mitigates the mismatch losses between modules by using DMPPT per module, as shown in Fig. 1.5 [22]. It has the following features:

- \checkmark Low cost powered by mass production.
- ✓ High modularity and simple design.
- ✓ Easy integration at DG systems
- ✓ Minimum installation time due to easy plug-and-play feature.
- \checkmark Low mismatch losses.
- ✓ High safety without hazardous high voltage DC cables.

This architecture has two different configurations: MIC and MII [23–27]. Module Integrated Converter (MIC) utilizes a DC-DC converter having an internal MPPT controller for every module. The required high DC bus voltage is then provided by connecting many MICs in series. Then the interface between the PV side and utility grid

is physically implemented by a string inverter. The exciting feature of this configuration is the similarity with the string PV architecture. It uses the string inverter but adds a DC optimizer for every module. It is worth noticing that MIC is popularly called DC optimizer in much literature works [25]. It completely disconnects between the PV modules and reconnects them via DC optimizers on the DC bus. It completely decouples the mismatched modules, and the shaded modules inject less current than unshaded ones. Form the cost point of view, this architecture is significant to replace the traditional string inverters without redesigning and integrating the DC optimizers.



The second configuration performs the individual PV module power directly by Module Integrated Inverters (MII). Here, an inverter with its MPPT controller per PV module removes the DC bus and decreases previous configuration costs, leading to flexibility and scalability. This configuration is known as AC PV modules in many research and industrial societies because it can attach MII in the PV module's backward, as shown in Fig. 1.6. It produces a PV module with AC output power [23].



Figure 1.6: AC smart PV module.

(7)

These days, it is clear that the DMPPT concept achieved in the recent PV modules adds many advantages. The reduction of PV modules for each inverter diminishes the mismatch losses from one side. It increases system modularity and plug-and-play operation for PV architecture expansion or periodic maintenance on the other side [27]. This promising feature makes the researchers move further than this by developing PV sub-modules and PV cell architectures [82-29]. It is useful for low-power applications, rather than electricity harvestings, such as portable devices and IoT applications. These applications, in general, are exposed to many severe levels of partial shading.

However, specifically for energy harvesting, optimization between the number of PV modules and inverters is necessary because it is tightly reflected upon the overall system cost. Moreover, it depends on the level of partial shading conditions. The PV architecture's feasibility depends on two interconnected factors: the total cost and partial shading level. When the partial shading level is small, that may happen in many conditions, the mismatch losses between two closet PV modules are negligible. This situation draws the attention to connect one or two or even three modules per inverter to make a significant cost-based architecture. To make a flexible solution, researchers and designers proposed micro-inverters in many research pieces of literature [30]. Technically, it worth mentioning that the micro-inverter becomes more general and flexible than the MII. It's because the former is independent of the PV module ratings and has utilized several PV modules. On the other hand, MII has utilized only one PV module, making it restricted to specific scenarios.

1.3 Micro-inverter topologies

As mentioned above, Micro-inverters are considered the central fatal processing units in recent PV module architectures due to its high feasibility for performance and cost [30]. Single-phase inverter topologies are dominant in this direction. They have witnessed extraordinary market success, such as Solar Bridge, SMA, Enecsys, and Enphase inverters [31-32]. Moreover, single-phase inverters' voltage gain capability using isolated and non-isolated converter topologies has been discussed and classified [33-37]. Different connection configurations of single-phase micro-inverters have been also introduced to improve the overall efficiency and grid features [30]. However, single-phase inverter

topologies has two crucial technical drawbacks.

1.3.1 Double-line Frequency Component

In this topology, the DC power of one or more PV modules supplies the inverter's DC side. It achieves the maximum power operation by using MPPT controllers. However, a double-line frequency harmonic component is introducing at the inverter's input. The source of this component is the discrepancy among the PV modules' instant DC power, and the AC delivered single-phase sinusoidal power. This component is the main issue to develop the smooth operation of MPPT controllers and inverter control. The straightforward solution is to integrate a high storage element between PV modules and inverter for decoupling this component. A bulky electrolyte capacitor forms this balance by filtering the resulting double-line frequency form the inverter input.

Unfortunately, this capacitor is the central defect because it has a short lifetime. The lifetime of capacitors is inversely proportional to its capacitance. To get the same decoupling power capability, increasing the voltage at the electrolyte capacitor location decreases the required capacitance and allowing small values. This technique replaces the electrolyte capacitor with a reliable film one, such as film capacitors. However, using a dedicated film capacitor with a long life needs an additional circuit called a decoupling circuit to increase the voltage [33].

For Example, Fig. 1.7 shows a single-phase VSI inverter with an additional leg and small capacitor. In addition to a small film capacitor, this leg is responsible for decoupling the double-line frequency component from the DC side [38]. There are many types of decoupling circuits. It depends on the decoupling circuit location such as the PV side, DC link side, and AC side. Also, it was discussed and covered by several literature reviews [33, 34]. However, this circuit adds more challenges for control and the general performance of single-phase micro-inverter topologies. Also, all their trials introduce additional components as well as additive costs.

1.3.2 Limited power scale

All accomplished single-phase inverter topologies have been limited to small-scale single-phase (110 and 120 AC voltage) residential and commercial PV installations [39-40]. With time, customers getting more significant energy loads with more powerful

energy consumption. Large air conditioners and pool heaters are some of the famous examples. Also, electric vehicles have become more familiar for many customers and need high-power-rated inverters for quick charging. For this reason, many homes will need a three-phase configuration by providing three live wires instead of one wire in a conventional single-phase configuration. Although three multiple single-phase inverters are recently used to perform three-phase power, it has many complications, especially the power balance.



Figure 1.7: Comparison between single-phase micro-inverter with decoupling leg and three-phase micro-inverter.

1.3.3. Why Three-Phase Micro-inverters

The previous drawbacks draw attention to three-phase inverter topologies. As illustrated in Fig. 1.7, the single-phase having the decoupling circuit has the same component counts as a three-phase micro-inverter. Moreover, the third leg processes additional power in the three-phase micro-inverter instead of the decoupling function in the single-phase one. This solution avoids the expense of less reliability (electrolyte capacitors) and more system complexity (decoupling circuits) [33-34]. Furthermore, three-phase micro-inverters based on identical modular three legs (photo 2) have more straightforward control and are the most preferably used ones. Three-phase micro-inverters also have the following advantages [39-40]:

- ✓ The electrolyte capacitors' presence provides low voltage ripple components on the PV side, DC link voltage, and the grid where the double line frequency component is disappeared.
- \checkmark This exciting feature opens the way for small and low-cost film capacitors to damp

only the high switching frequency harmonics and expand the PV module architecture's life and reliability.

- ✓ The low voltage ripple enhances the tracking efficiency of the MPPT controller.
- ✓ Multiple units are not required due to the inherent three-phase balancing feature.
- Reduction of cost using a single three-phase micro-inverter contrary to three singlephase units in high power purposes.
- ✓ It is naturally controlled through a three-phase central power system operation having symmetrical operation between all phases.
- ✓ For homes, the single-phase appliances are feeding on one active phase. Large loads such as electric cars are connected at all three-phases.

1.4 Standards, and requirements of three-phase micro-inverters

Fig. 1.8 shows the possible implementation of three-phase inverters at residential PV systems. The system's main parts are the DC side (PV or Battery) and the AC side (utility grid). Both of them have specific standards and requirements to process power simultaneously. Besides, the utilized micro-inverters should have some specific performance points such as small size, low cost, and high efficiency.

It is worth noticing that, as shown in Fig. 1.8-a, there are two independent three-phase inverters. The first process the power of PV modules. It needs high voltage gain and efficiency and unidirectional power capability. The second three-phase inverter, for the battery bank, processes the power in both directions.

The interest of this configuration is the flexibility and low power ratings of both inverters. It raises the grid voltage during the discharging state of the battery and decreases commonly happening inverter tripping. It is useful only when the exported battery power more considerable than the PV modules' power. Moreover, the battery saving across all phases and mitigate phases unbalance.

Fig. 1.8-b shows the second configuration, which is cheaper than the first one. It uses a bidirectional DC-DC converter instead of the battery inverter. Therefore, the power connection is shifting to the PV DC side. However, this configuration increases the rating of solar PV inverter and needs bidirectional power processing capability. Per EIA standards, the PV modules, battery, and the grid's feature requirements are concerns
during the inverter topology selection and the design process for both configurations [1]. The required functions for each system part are summarized below.



a- First configuration of three-phase inverters



b- Second configuration of three-phase invertersFigure 1.8: Possible implementations of three-phase inverters.

1.4.1 Required Features of PV Modules (DC Source)

In general, the PV modules are nonlinear energy sources with a V-I curve highly dependent on the irradiance and temperature. For energy harvesting optimization, The PV modules require to be operated at MPP with high tracking efficiency and zero leakage current by using MPPT controllers. Many MPPT controllers have been developed [40-44]. Whatever the presented micro-inverter topologies, the MPPT optimization degree depends on the following factors:

- ✓ High tracking efficiency: This is accomplished using micro-inverter topologies with continuous input current and has minimal input voltage and current ripple. There is an inverse relationship between MPPT tracking efficiency and input PV module ripples.
- Robust dynamic response: This factor depends on the control performance rather than the selected topology. This important factor can be easily implemented using simple

inverter topologies that possess a plain control technique.

 Low processing power and low execution time: This defines the chosen control boards for the selected inverter topology and the computational burden, defining the system cost and reliability.

In contrast, Low voltage PV module architectures employing transformer-less microinverter topologies are exposed to grounding issues [45]. That is why the distance between customer and PV modules is even closer. Using high-frequency switching and in the presence of the parasitic capacitance among the PV module cells and the ground introduces a leakage current which cannot be neglected.

This leakage current produces Common-Mode Voltage (CMV) on the PV modules resulting in many severe personal safety concerns. It accumulates the architecture losses, minimizes the grid-connected current quality, and stimulates severe radiated Electro-Magnetic Interference (EMI) [46].

There are two main concerns related to CMV, the common-mode loop path, and the source of CMV. However, many papers studied CMV reduction in non-isolated topologies of micro-inverters [30, 47, and 48]. The CMV has not been mitigated at most because the common-mode is highly dependent on switching patterns, inverter operation, and design. Therefore, High-Frequency Transformers (HFT) integration can cut the common mode loop path. Therefore, isolated topologies of micro-inverters are very reliable in reducing CMV [49].

1.4.2 Required Features of Grid Side

The rapid growth of the grid-tied PV-based micro-inverters and its influence on the grid performance must have addressed. Many national and international standards, such as IEEE 519-2014 and IEC 61727, are certified to evaluate micro-inverter performances. One of the most critical characteristics of micro-inverters is the injected power semblance. This semblance has been assessed by the Total Harmonic Distortion (THD) of the grid current. THD is disturbed through the harmonic distortion of the synchronized voltage on the grid and the harmonic emission on the PV side. The maximum value of the allowed THD is 5%. On the other hand, the inverter power and THD relationship must be considered for full optimization [50]. The THD usually increases when the inverter operates in light load [51].

The mismatch of output AC voltage components produced by micro-inverters can introduce a massive DC current injection. Well-designed micro-inverters must prevent this current and its destructive effects under satisfied limits, especially for transformer-less topologies. The DC current injection limit in IEEE 1547 standard is specified to be lower than 0.5% of the injected grid current [52].

1.4.3 Required Internal Features of Three-Phase Micro-inverters

First, the module PV architectures must have a small capacitor between the PV modules and the selected three-phase inverters to support PV characteristics. Moreover, it provides smooth DC voltages with tiny voltage ripples. However, some inverter topologies introduced reflected harmonics to the PV side due to its switching pattern and its inherent characteristics. These harmonics, especially the low-order harmonics, violates the system performance and needs additional capacitors in between. Three-phase topologies such as flying-capacitor multi-level and some Z-source inverter topologies are examples of these types of topologies. However, the integrated bulky capacitors used by these kinds of topologies have a small lifetime. The combined issues such as heat dissipation, humidity, and overcurrent deteriorated these capacitors' whole performance. Therefore, these topologies are not good options in PV module architectures.

Second, the step-up voltage gain in grid-tied micro-inverters is necessary because most commercial PV modules' DC voltage is lower than 40 V. Traditionally, cascaded two-stage or multi-stage micro-inverter topologies are proposed to solve this issue [19, 20]. The first stage is responsible for boosting the PV voltage using boost DC-DC converter topologies. Switched-inductor and switched-capacitor converters have proposed to generate high gain DC-DC boost converters [53]. The Voltage multipliers and the coupled inductor based boost converters are other topologies that have also been proposed to widen the voltage range and improve flexibility [54]. Typically, the second stage is an inverter to invert the DC voltage to a high-quality AC voltage. The main advantage of the two-stage structure is the flexibility of the controller design. However, the inverter's efficiency and reliability are relatively low due to multiple power processing and the high components count, which introduces increased system cost and complexity.

Recent advancements of single-stage inverter structures are preferred. They are highly

used to reduce the overall cost and complexity [55]. These single-stage structures not only perform voltage boosting, but they also can perform AC conversion. Unfortunately, these single-stage systems suffer from complicated control algorithms, sometimes requiring software rather than the hardware implementation.

Third, voltage isolation in the selected micro-inverter topology is essential to cut the CMV under acceptable limits. For safety purposes, a Line Frequency Transformer (LFT) is integrated among inverter and utility grid at traditional PV architectures. LFT is bulky, costly, and may pose efficiency challenges, especially for advanced PV architectures [56]. Therefore, compact HFT is considered a mandatory part of micro-inverters to achieve voltage isolation and provide additional voltage gain. HFTs perform bidirectional power isolation between PV modules and the utility grid, improving the overall system's reliability and providing more security for humans in the PV field. However, microinverters that utilize HFT have low efficiency and higher cost than transformer-less topologies [20, 39]. Therefore, using state-of-the-art technologies to integrate HFT into the next generation of micro-inverters are highly recommended considering these factors. Fourth, as discussed before, the micro-inverters rating is between MII inverter and string inverters (100W-5kW, for example). So adding another PV module means more power. So if the selected inverter topology is capable of further extension or scalability, this will enhance the performance with less complexity. If not, the parallel connection of many inverters is the solution.

Fifth, increasing the micro-inverter's efficiency is vital because it has prominent and remarkable license access to PV industrial markets. Micro-inverters' efficiency depends on the characteristics of their inherent parts, such as semiconductor switches, passive elements, and switching strategy. It has recently shown that using advanced semiconductor technologies such as SiC and GaN switches can guarantee good performance and high efficiency [57–58]. Furthermore, the developing technologies in the passive elements are considered an added value for the inverter efficiency. Concerning the switching strategy, using high-frequency switching techniques and optimizing switching losses and conduction losses ratios improve the overall efficiency [59]. Also, soft switching methodologies such as Zero-Voltage-Switching (ZVS) and Zero-Current-Switching (ZCS) contribute significantly to improve inverter efficiency [60-61].

Finally, to push the market of PV-based three-phase micro-inverters, each part of the inverter should be investigated to reduce the overall cost. The micro-inverter cost in a PV system accounts for about 10% of the total cost [16]. The following factors positively define the overall cost of a micro-inverter:

- \checkmark The number of components and their ratings.
- ✓ Low power loss and low heat dissipation will diminish heat sinks, which decrease the overall cost.
- ✓ Operating switching frequency because it determines the size and cost of passive elements.
- ✓ Power density improves the consumed Printed Circuit Board (PCB) area and its packaging technology.
- Type of the implemented controller: analog or digital, its effective MPP tracking, and the number of used loops can improve the overall system efficiency and related cost.



Figure 1.9: The critical parts of the three-phase inverters and their detailed features.

The field test data shows that the gap between micro-inverters' lifetime and the PV module is still large. Therefore, improving the reliability of PV inverters is a must. The micro-inverters failure effects on the overall PV system performance should be mitigated with modular and redundant topologies. Finally, Fig. 1.9 shows the list of all previously mentioned requirements.

According to previous discussions, three-phase micro-inverters' published topologies are presented and evaluated based on a new classification in this thesis. The simple topologies

that achieve a high percentage of previously developed standards and requirements are revised in detail among dozens of existing inverters. The developed classification has been used to identify the merits and demerits of the classified inverter topologies, according to Fig. 1.9.

A recommendation has then been given by selecting the differential inverters. This thesis also made a complete overview of the presented differential based inverters to find the best one. According to the previous overview, the Single-Ended Primary-Inductor Converter (SEPIC) carries many advantages that promote a well-functioning three-phase inverter. For this reason, differential inverter using SEPIC converters is proposed in this thesis for micro-inverter application. Furthermore, this thesis presents a novel grid-current control for achieving low THD and small computational burden. Besides, analysis and experimentation of a modular differential inverter, depend on many SEPIC converters, have also been proposed.

1.5 Thesis Objectives

Thesis objectives are shown in Fig. 1.10 and can be summarized as follows:

- Low-order Harmonic Components in the proposed SEPIC-based differential inverter; First, an in-depth mathematical model of negative-sequence harmonic component has been developed to know its source. Moreover, the effect of it on the output gridcurrent and DC side current has been investigated. Then, the parameters selection of the proposed SEPIC-based differential inverter is designed based on the developed model.
- Grid-current Control, in this thesis, two-loop based d-q synchronous fame gridcurrent control has been proposed. It has a simple structure and is easily validated at traditional DSP boards. In the first loop, PID, Type-II, and PI compensators are designed and implemented to achieve grid-current control and inject pure active power. The high-order of the SEPIC converters' open-loop transfer function needs rigid compensators for a stable and reliable performance. Therefore, better performance for the proposed inverter is achieved. The second loop utilizes a negative-sequence harmonic component detection circuit to mitigate it. This circuit is based on three-stage Low-Pass Filters. The performance of both control loops is

validated in simulations and experiments.

• A modular differential inverter is also proposed in this thesis for high power applications purposes. It incorporates many of the proposed SEPIC converters. The practical issues of differential inverter such as variable duty cycle, low order harmonics, and circulating power are presented during the design process. A mismatch in integrated SEPIC converters is also investigated to confirm the system performance. Moreover, a laboratory prototype of a simple modular differential inverter is provided to prove the authenticity of the proposed modular differential inverter.

1.6 Thesis Outline

The arrangement of this thesis is as follows:

Chapter 2 provides a well-supported detailed classification for the best three-phase inverter topologies between many dozens of published inverters. Single-input and multi-input inverter topologies using single-stage and two-stage conversion are evaluated and revised in detail. Differential inverters, Z source inverters, and Switched-capacitor inverters are an example of single-stage inverter topologies. Two-stage VSI is also discussed, especially for low power applications. In contrast, modular multi-level inverters and multi-input VSI inverter topologies are also presented. The component counts and voltage and current stress of screened topologies are compared. The final recommendation of using differential inverters in the rest of the thesis is selected and accompanied by micro-inverter application merits.

Chapter 3 presents a comprehensive analysis of the proposed SEPIC-based differential inverter. It includes the principal operation, modulation techniques, and differential inverters issues such as variable duty cycle, low-order harmonics, and circulating currents. A novel mathematical model is proposed to support this analysis. Then, the main advantages of the SEPIC converters is discussed. After that, a grid-current control is developed, where A PID compensator is designed to enhance the inverter's stability and compensate for the utilized SEPIC converters' phase-delay. A negative-sequence harmonic component detection circuit is introduced in the second loop of the developed control. The design details of passive elements and switching devices are also covered.

At final, the provided experimental, and simulation results prove the robustness of the presented control.



Figure 1.10: Objectives of the Thesis.

Chapter 4 presents a modified grid-current control, which is rigid for the proposed SEPIC-based differential inverter. The PID compensator is replaced with a Type-II compensator. This compensator is better due to its capability to eliminate the high switching harmonics of the proposed inverter. Therefore, it has successfully decreased the low-order harmonics and THD and showed its accuracy by satisfying the recent grid codes (<5%). Finally, a PI compensator is presented for the proposed inverter. The control burden is decreased and validates the usage of inexpensive DSP boards.

Chapter 5 presents the analysis and design of modular differential inverter based on nine SEPIC converters to process high power (three times of proposed one in previous chapters). The validation of the proposed inverter with mismatch parameters at SEPIC converters is investigated using numerical simulations.

Chapter 6 covers the thesis summary, influential contributions, and future ideas of the proposed inverter.

Chapter 2: Classification of Three-phase Micro-Inverter Topologies

2.1 Introduction

Traditional three-phase VSI topology is one of the most famous topologies which convert the PV power to pure sinusoidal AC power. In industrial markets of medium-voltage adjustable speed drives, the VSI design is reasonably efficient, has acceptable reliability with smooth dynamic response, and can run motors without de-rating. These promising features make it an excellent candidate to process the power between RES and grid using simple three legs, as shown in Fig. 2.1 [56]. It controls active and reactive power and injects AC grid-current with low THD.

The grid requirements, such as voltage isolation, are developed by adding LFT between RES and grid. For many years, this inverter was used in RES, especially in large-scale applications [56] because of the distinct advantages such as high efficiency, simple control, and easy implementation. However, in small scale RES connected to the grid distribution level (<5 kW), the new PV module architectures have independent PV modules with low input voltages [64]. Moreover, the partial shading scenarios force this voltage to be time-variant [63-65]. This characteristic of the new PV architectures introduces a voltage-gain limitation because the VSI AC output voltage is less than the PV DC voltage. The considerable size of LFT has additional limitations to inverter size and cost. Finally, VSI does not have continuous input current property.



Figure 2.1: Three-phase VSI equipped with LFT for grid-connected applications.

Otherwise, the three-phase Current Source Inverter (CSI) provides a step-up output voltage with continual input-current, as shown in Fig. 2.2. PV modules supplied power behaved like a current source and is compatible with PV's inherent characteristics. Compared to VSI, the current of the CSI is controlled and limited by using input inductor [66]. The utilized switches are subjected to lower peak current due to the constant power supply. The CSI's negative link voltage capability enables the inverter to handle reactive or regenerative loads. However, it needs unidirectional switches composed of two switches and two diodes per leg and introduces additional components. The used DC inductor is relatively large to exhibit current source features. The large DC inductor also slows the current response. The load must be balanced; otherwise, the input current has high input harmonic components, making the input inductor size increase. Due to this disadvantage, the three-phase CSI is mostly used in three-phase high power RES applications.



Figure 2.2: Schematic power circuit of three-phase CSI.

Although the traditional VSI and CSI represent an apparent success in many RES applications, previously mentioned properties have introduced many limitations. These limitations complicate their application in recent low-power PV modules architectures and their promising micro-inverter topologies. The voltage gain of both is still inapplicable. Galvanic isolation is still problematic. Adding another inverter with a parallel connection to the old one is the only solution for power extension or scalability. This issue needs a redesign of PV architecture and complicates its wirings.

Motivated by the problem statement mentioned above, many inverter topologies were proposed in the literature as candidate solutions. In between, the interesting topologies for three-phase micro-inverters at PV module architectures have been classified as follows.

- Two-Stage VSI inverters.
- Single-Stage Z Source Inverters.
- Single-Stage Switched-Capacitor Inverters.
- Single-Stage Differential-Based Inverters.

2.2 Two-Stage VSI inverters

Three-phase VSI has a voltage step-down property. They are adding a boosting stage before VSI to provide the voltage step-up ability. It has a DC-DC boost or other boost-derived converters like buck-boost, isolated boost, isolated buck-boost, or full-bridge converters. In [67], a conventional boost converter is integrated with the three-phase VSI. The boost converter performs MPPT control and feed DC power, as demonstrated in Fig. 2.3. The presented system uses an adaptive DC link voltage by adjusting the reference DC link voltage based on the Common Point of Interconnection (CPI) voltage. The adaptive DC link voltage control helps in the reduction of switching losses. A feed-forward loop was also utilized to improve the dynamic behavior.



Figure 2.3: Three-phase two-stage VSI proposed in [67].

In [68], a two-stage three-phase grid-tied inverter was presented, as illustrated in Fig. 2.4. The presented ZVS technique improved inverter efficiency. The boosting stage is an LLC resonant DC-DC full-bridge converter with center point iterations MPPT algorithm. Full-bridge LLC resonant converter can sustain ZVS or ZCS operation for the full range of input voltages, improving the inverter efficiency. Moreover, the phase skipping control improve inverter efficiency under light load conditions. A triple loop control of ZVS is

also presented to guarantee the inverter's stability in PV applications [69]. Fig. 2.5 shows the two-stage three-phase VSI inverter proposed in [70]. The boosting stage is an isolated full-bridge DC-DC converter. The second stage is a three-phase VSI inverter. They used a soft switching technique based on Zero Voltage Zero Current Switching (ZVZCS) mechanism.



Figure 2.4: Three-phase two-stage VSI proposed in [68].



Figure 2.5: Three-phase two-stage VSI proposed in [70].

As shown in Table. 2.1, most previous topologies can perform MPPT with high tracking efficiency due to continuous input current. For topology V1, the DC link capacitor is only about $35\mu F$, so that the film capacitor has been utilized. For topologies V2 and V3, an electrolytic-less capacitor was still needed. For grid side, the reported THD is 5% for all. However, this topology needs DC link voltage control in addition to grid-current control. At the inverter topology, the CMV is reduced in isolated DC-DC converters. The component counts of these topologies is illustrated in Fig. 2.6. Although, V3 has the lowest number of components, overall component counts is still large.

On other hand, a multi-input three-phase VSI micro-inverter, shown in Fig. 2.7, is proposed [71]. It utilizes fewer switches per leg and reduces the component count and overall cost. The number of switches is reduced by one in each leg compared to traditional two-stage VSI. This topology's exciting feature is the power extension by adding PV and

its leg to the developed inverter. Moreover, each PV module stage's power extension is accomplished by using gate signals on the nearest two switches to the inductor. The middle switch in each leg is utilized for inverter operation. However, this topology still implement electrolyte capacitors. Moreover, it lacks voltage isolation.

Tuble 2.1. 1 enormanoe enaracteristics of two suge voi topologies.								
Topology		Power				Control	Switchi	THD
Fig./Re f	Boosting Stage	Rating kW	CMV	η	MPPT		ng	%
2.3/67	V3 Boost	6.5	High	NR	INC	PWM	Hard	5
2.4/68	V1 LLC	0.4	Low	95.5%	СРІ	PWM + ZVS	Soft	5
2.5/70	V2 Full Bridge	~1	Low	91.5%	-	PWM + ZVZCS	Soft	5

Table 2.1: Performance characteristics of two-stage VSI topologies.



Figure 2.6: Component counts of three-phase two-stage VSI inverter topologies.

2.3 Single-Stage Z Source Inverters

Impedance Z-source Inverter (ZSI) is a single-input single-stage inverter with many interesting features. The ZSI can limit the VSI and current source topologies' problematic issues and provide a step-up voltage conversion solution. The classical ZSI has an extra circuit called the Z-network, as illustrated in Fig. 2.8. This network improves the inverter's features by increasing buck/boost conversion due to Shoot Through (ST) property. ST also eliminates the dead time required in conventional VSI topologies [72]. However, the classical ZSI topology has low voltage gain and high switching losses due to high voltage

stress. In the presented studies [73-74], different modulation techniques were proposed to increase the voltage gain without adding extra elements in ZSI power stage, enhancing the boosting capability over a complete range operation.



Figure 2.7: Multi-input three-phase two-stage VSI [71].



Figure 2.8: Classical three-phase Z source inverter.

In [75], high gain was achieved by defining the right choice of active vectors and ST states' placement. In this way, the voltage gain can be increased at high modulation indices. Also, with the inclusion of ST states, the number of switching is less, which results in low switching losses and low voltage stress, and the small size of the reactive components. Many topologies have been derived from classical ZSI (Z1) for micro-inverter applications, such as:

- 1. Z2-quasi ZSI (qZSI).
- 2. Z3- Inductor-based ZSI (L-ZSI) switched inductor (SL-ZSI).
- 3. Z4-Switched-inductor quasi ZSI (SL-qZSI).

- 4. Z5-Transformer based ZSI (Trans-ZSI).
- 5. Z6-Inductor-based ZSI (L-ZSI).

The qZSI topology, shown in Fig. 2.9, has the advantage of continuous input current, making it suitable for MPPT implementation of PV applications with high tracking efficiency. In [76], grid-current control depend on Finite Control Set Model Predictive Control (FCS-MPC) with reduced computation time had proposed.



Figure 2.9: Structure of three-phase quasi Z-source invertor (ZSI) topology.

A modified ZSI (L-ZSI), involving only diodes and inductors for boosting the inverter's ratio, was proposed in [77] and is shown in Fig. 2.10. The L-ZSI is a modification on the SL-ZSI using fewer components. The proposed topology provides inrush input current suppression because no current flows to the start-up's circuit. It also provides a continuous input current with the common ground between PV modules and inverter.



The trans-Z-source inverter topology utilizes the Z-network, a transformer, and a capacitor having reduced components count, as proposed [78]. Its circuit diagram is illustrated in Fig. 2.11. The voltage gain and stress are significantly improved using a transformer's turn ratio larger than one. Isolated Z-source inverter using HFTs is

presented in [79] for low voltage applications (Fig. 2.12). This isolated topology introduces extra components because the utilization of HFTs requires an additional power conversion stage.

On the other hand, four-leg quasi-ZSI was proposed in [80], as illustrated in Fig. 2.13. There, a Model Predictive Control (MPC) technique was used to overcome the drawbacks of the classical ZSI. MPC was used to control the load current and quasi ZS network capacitor voltage with high accuracy, leading to a fast response. The proposed controller handles each phase current independently. As a result, the proposed quasi ZS four-leg inverter provides fault-tolerant capability; for example, if one leg fails, the others can work usually. Finally, a three-phase four-leg inverter was employed to ensure the PV module architecture's reliable operation unbalanced load conditions.



Figure 2.12: Power circuit of voltage fed isolated ZSI [79].

Table. 2.2 contrasts the previously discussed three-phase ZSI topologies, which are effectively utilized in micro-inverters. It is worth noticing that, for the high tracking efficiency of MPPT controllers, the continuous input Z-source micro-inverter topologies are preferred. CMV is highly reduced in isolated Z-source topologies, where, in non-isolated topologies, modified PWM is required. For high voltage gain, L-ZSI [77] and voltage fed trans-ZSI [78] are recommended. Fig. 2.14 presents the number of

components of each topology. Classical ZSI, L-ZSI, and QZSI have the lowest number of components, which decrease the overall cost. Table 2.3 shows the voltage stress of each topology. For a turn ratio equal to 1, all Z-source micro-inverter topologies have equal switch stress except L-ZSI. The same findings for capacitor stress except for Z3, which has no capacitors making it more reliable. However, the reported THD is 11% and 8.7% in [75] and [80]. Moreover, Isolated ZSI [79] has a high number of components. The Z network's capacitor was huge and requires an electrolytic capacitor. Finally, for qZSI topology and trans-Z-source topologies [76-78], this capacitor is still 25% larger than the electrolytic-less capacitor.



Figure 2.13: Power circuit of three-phase four-leg Quasi ZSI [80].

Topology		Power			Gain		Switc	THD
Fig. Ref.	Solution topology	Rating kW	CM V	η	$M * B * \frac{V_{dc}}{2}$	Control	hing	%
(8) [75]	Z1 classical	0.26	No	NA	$B = \frac{1}{1 - 2D}$ $D < 0.5$	SVM	Hard	11
(9) [76]	Z2 qZSI	0.45	No	NA	$B = \frac{1}{1 - 2D}$ $D < 0.5$	FCS-MPC	Hard	NA
(10) [77]	Z3 L-ZSI	~0.15	No	NA	$B = \frac{1+D}{1-D}$ $D < 1$	Maximum Boost control	Hard	NA
(11) [78]	Z4 Tran-ZSI	NR	No	NA	$B = \frac{1}{1 - (1 + n)D}$ $D < 0.5$ $1 - (1 + n)D \neq 0$	Constant Boost control	Hard	NA
(12) [79]	Z5 I-ZSI	2.88	Yes	NA	$B = \frac{n}{1 - 2D}$ D < 0.5	Constant B 3 rd inject	Hard	NA
(13) [80]	Z6 QZSI	~0.4	No	NA	$B = \frac{1}{1 - 2D}$ $D < 0.5$	MPC	Hard	8.7

 Table 2.2: Performance comparison for selected topologies of Z-source micro-inverters.



Figure 2.14: Analysis of components' count utilized in three-phase Z-source micro-inverter topologies.

2.4 Single-Stage Switched-Capacitor Inverters

In this topology, a switched-capacitor cell or circuit is used instead of the boost converters at two-stage VSI topologies. The principal operation depends on the capacitors charging in parallel then discharges the capacitors in series to magnify the input DC voltage. In [81], a Switched-Capacitor-Voltage-Doubler (SCVD) network is proposed. It includes two capacitors, two diodes, and two switches. Like boost converters, it is put between the DC and conventional VSI, as shown in Fig. 2.15. Space vector PWM is used to drive the additional two switches to reduce the CMV. In [82], a three-phase inverter utilizes a seven-level Active-Neutral-Point Clamped (ANPC) inverter instead of conventional VSI is presented. The switched-capacitor technique, which provides voltage step-up, is also developed, as illustrated in Fig. 2.16. ANPC multi-level inverter performs output voltage with a staircase voltage waveform, partially following a sine wave template. This technique leads to a low THD with high voltage levels and low voltage stress.

In [82], a single-stage boosting VSI topology is proposed. The voltage step-up is achieved using a magnetic integrated switching inductor-capacitor network. It includes an inductor, two diodes, and two-identical switching inductor-capacitor cells. The size of the magnetic circuit is reduced by magnetic coupling, as illustrated in Fig. 2.17. However, these topologies of switched-capacitor cells have many components, especially at the three-phase systems, and lack galvanic isolation and modular construction [83]. In addition to VSI, extra two switches and two diodes are added in topologies [81] and [82], respectively. Moreover, topology in [83] has 27 switches. Furthermore, it has many issues related to

the capacitors' mismatch and self-voltage balancing [84].



Figure 2.15: Switched-Capacitor based Inverter proposed in [81].



Figure 2.16: Switched-Capacitor based multi-level Inverter proposed in [82].



Figure 2.17: Switched-Capacitor-inductor Inverter proposed in [83].

2.5 Single-Stage Differential Inverters

Three-phase differential micro-inverters are single-input single-stage step-up/down topologies. They have high efficiency with a low number of components, hence yielding acceptable total cost and size. Moreover, these topologies can provide bi-directional power transfer. This topology's main drawback is the voltage stress resulting from the variable duty cycle operation and the circulating current produced from the differential

connection of converters. The schematic circuit of the differential inverter is illustrated in Fig. 2.18.



Figure 2.18: Schematic power circuit of three-phase differential inverter topology.

It composed of three DC-DC converters connected in parallel at the input and differentially at the output side. The three-phase output is produced using sinusoidal pulse width modulation (SPWM) with a 120 phase shift. This inverter can step-up/down the voltage using suitable DC-DC converters. The output voltages have both DC and AC components. The DC component values are equal for all phases, which introduce the voltage stress. It is canceled by the differential connection, while the AC component appearing on each converter's output voltage is combined to create the sinusoidal AC voltage of the inverter. The diverse characteristics of the DC-DC converters give rise to many differential inverters, as previously shown [85–110]. Fig. 2.19 and Fig. 2-20 show the various types of the non-isolated and isolated differential three-phase inverter topologies. The selected topologies are mentioned as D1 to D6. The PV modules with its MPPT are easily implemented with high tracking efficiency for DC-DC converters, which have non-pulsating input currents. For example, boost, SEPIC and Cuk converter topologies.

For CMV elimination, isolated DC-DC converters like Flyback, SEPIC, and Cuk converters are better. Otherwise, CMV elimination can be achieved by using proper modulation techniques. In buck-boost and boost converters, high voltage is effectively possible using a high duty ratio. In isolated topologies, the voltage gain is attained by the turn ratio of HFT in addition to the converter's duty ratio.



Figure 2.19: Non-isolated three-phase differential inverter topologies.

Table 2.3 details the comparison between different DC-DC converters, which are utilized in differential inverters. It is worth mentioning that all presented topologies are in the range of micro-inverters applications (0-5kW). The reported THD for most topologies is under 5%. As mentioned in chapter 1, this value is compatible with current grid standards. Both linear and nonlinear control methods were implemented entirely in many literature work pieces, reflecting the control function's simplicity. As mentioned before, the output of differential inverters has two different components; DC and AC. DC component is canceled on the grid side, but it increases the stress on the inverter switches. Therefore, Table 2.4 illustrates the voltage stress comparison for the selected topologies. It is worth mentioning that the buck-boost converter's topology is better from the component count and total voltage stress point of view. Between different isolated buck-boost converters, flyback and SEPIC converter is effective. Fig. 2.21 shows the component counts of the discussed topologies. The Full-bridge converter has many components. On the other hand, the four-switch SEPIC differential inverter topology has the lowest components count.



a- D5-isolated differential Cuk inverter [89]



b- D6-isolated differential full bridge inverter [90] **Figure 2.20:** Isolated three-phase differential inverter topologies.

Topology		Power	THD				Mode	
Ref.	Solution DC-DC	Rating kW	1HD %	CMV	η	Control	scheme	Switchin g
[85]	D1 (Boost)	1–1.5	6–7	High	NA	Sliding mode	ССМ	Hard
[87]	D3 5 buck-boost	2.5	1.8	Low	95	3 control loops	ССМ	Hard
[88]	D4 SEPIC	2	NR	High	NA	DI sliding mode	ССМ	Hard
[89]	D5 Cuk	0.5	6	Low	93	PR + SLM FF control	CMS DMS	Hard
[90]	D6 Full bridge	NR	NR	Low	89	CSVPWM	ССМ	Soft

Table 2.3: Performance comparison of different DC-DC converters utilized at differential inverters.

,	Гороlоду	Switch A	Switch B	Capacitor	Capacitor
Ref.	Solution	Switch A	Switch D	А	В
[85]	D1 Boost	$V_{dc} + 2V_{grid}$	$V_{dc} + 2V_{grid}$		2V _{grid}
[86]	D2 Cuk	$V_{dc} + 2V_{grid}$	$V_{dc} + 2V_{grid}$	$V_{dc} + 2V_{grid}$	2V _{grid}
[87]	D3 5 buck-boosts	$V_{dc} + 2V_{grid}$	$V_{dc} + 2V_{grid}$	V _{dc}	2V _{grid}
[88]	D4 SEPIC	$V_{dc} + 2V_{grid}$	$V_{dc} + 2V_{grid}$	V _{dc}	$2V_{grid}$
[89]	D5 Cuk	$\frac{nV_{dc}}{+ 2V_{grid}}$	V_{dc}/n + $2V_{grid}$	$\frac{nV_{dc}}{+ 2V_{grid}}$	2V _{grid}
[90]	D6 Full bridge	$\sum V_{dc}$	$\sum 2V_{grid}$		2V _{grid}

Table 2.4: Performance voltage stress for selected topologies of differential inverters.



Figure 2.21: Components count utilized in three-phase differential inverters topologies [85–90].

2.6 Summary of the proposed classification

According to the previous discussion of possible best topologies for three-phase microinverters application at PV module architectures, the following statements are concluded:

- 1. The best topology at two-stage VSI topologies is LLC resonant DC-DC converter based micro-inverter.
- 2. L-ZSI is the superior topology at Z-source micro-inverters.
- 3. The best topologies of differential micro-inverters are HFT-based SEPIC and Cuk differential topologies.

In this thesis, the differential based inverters has been selected due to the following features [85-115].

✓ It provides two distinct features in one single processing stage; voltage boosting and

voltage inversion.

- ✓ Differential inverter topology having three DC-DC converters provides stepup/down voltage gain using their instantaneous duty cycle. Fortunately, many DC-DC converter topologies capable of operating on a wide range of duty cycles.
- ✓ The three DC-DC converters' symmetrical operation is implemented easily in singlephase, three-phase, or even poly-phase differential inverters. It just needs one or more DC-DC converter per phase.
- ✓ Isolated DC-DC converters, such as the SEPIC, flyback, and Cuk converters, have a small HFT. Therefore, the proposed differential inverter provides galvanic isolation and reduces leakage current, CMV, and Electro-Magnetic Interference (EMI).
- ✓ The HFT of isolated DC-DC converters is small because it works on the switching frequency. Furthermore, it can easily contribute to voltage boosting/bucking of the proposed differential inverter.
- ✓ In three-phase differential inverters, each DC-DC converter processes one-third of the inverter power and uses DC-DC converters with small passive elements, small rating, low energy storage, and high power density at low cost.
- ✓ This topology has bi-directional power capability due to DC-DC converters' symmetrical operation using the switches, which facilitates the same control.
- Simple two-switch-based DC-DC converters promote a low switches count (only six) and low-number of driver circuits that simplify the control structure.
- ✓ The redundancy of DC-DC modules topologies gives many opportunities for control robustness, low components counts, and overall system reliability.
- The research society has intensive work on DC-DC converter topologies, which get a reliable and efficient differential inverter.
- ✓ Unpretentious PWM modulation techniques such as space vector, unipolar, and bipolar modulations are easily implemented due to differential inverters' straightforward control functions.
- The simplicity of PWM circuits enables the control implementation of conventional DSP control boards. It decreases the complexity and inverter cost.
- ✓ Some DC-DC converters' continuous input current provides adjustable output voltage without electrolyte capacitor, especially for PV and fuel cell applications.

✓ DC-DC converters connection at differential inverters (input parallel, output differential series, or output parallel) has a modular configuration. It facilitates power extension by adding more modular DC-DC converters.

Chapter 3: Proposed SEPIC-Based Differential Inverter

3.1 Introduction

Currently, PV-based distribution systems have undergone numerous changes in their architectures. With more PV generation modules to the grid and the increasing use of the electric vehicles and its energy storage systems, it is requisite to adjust the power conversion between the various PV sources and the grid. It is imperative to use highly efficient and compact electronic inverters. For this purpose, this architecture requires an inverter with voltage step-up, which provides high efficiency and improves reliability and power density. However, besides modular construction, bidirectional power flow, and galvanic isolation requirements, these characteristics limit many possible existed inverter topologies.

According to the author's knowledge, differential inverter topologies achieve many of the requirements mentioned above efficiently due to their unique construction [85-110]. It appears as a star topology for the PV module architectures and its interesting inverter topologies. First, the differential inverter is a single-stage power circuit between many possible topologies. Second, it has a modular operation, galvanic isolation and bidirectional energy flow due to its simple construction established on isolated DC-DC converters.

A three-phase single-stage SEPIC-Based Differential Inverter SEPIC-BDI is proposed in this chapter. For galvanic isolation, it has three small HFT-based SEPIC converters. This chapter presents the differential inverter's principal operation, its SEPIC converters, and the PWM modulation. The proposed Modified Modulation Scheme (MCMS) provides static linearization for the duty cycle. It produces a grid output current free form low-order harmonics.

Furthermore, the Negative Sequence Harmonic Component (NSHC) is analyzed in detail. For this purpose, a novel mathematical model is presented. It investigates the NSHC source and its associations with the differential inverters characteristics, such as variable duty cycle, DC-DC converter unipolar operation, and circulating currents. The design process, analysis, and experimentation of the presented inverter are expressed. It includes the parameters selection, its design criteria, small-signal modeling, and the proposed control technique. Finally, the proposed inverter's simulation and experimental results have been provided to validate its performance and workableness.

3.2 Proposed SEPIC-Based Differential Inverter (SEPIC-BDI)

Fig. 3.1 depicts the power schematic of the proposed single-stage three-phase isolated SEPIC-BDI. It utilizes three identical bidirectional isolated DC-DC SEPIC converters (x = u, v, w). It is worth noticing that x indicates the SEPIC converter and its related phase, connecting it with the grid, as illustrated in Fig. 3.1. All three SEPIC converters are supplied from the same input DC source at the input side (input parallel connection) and star-connected at the grid side. Every SEPIC converter has two MOSFET devices (main switch S_{mx} and synchronous switch S_{rx}), one input inductor L_x , one HFT with magnetizing inductance L_{mx} , and two film capacitors (coupling capacitor C_x and output capacitor C_{ox}). This low number of components diminishes the proposed inverter size and enhances the overall power density. Moreover, the utilized film capacitors have a long life-time and enhance inverter reliability. SEPIC converters' uniform operation enables inverter modularity with plug-and-play activity and can be extended to higher power without additional control circuits. The differential inverter uses bidirectional SEPIC converters to produce AC output voltage greater or smaller than the input DC voltage. This feature is essential in RES to the grid with wide range operating conditions.

Since the three SEPIC converters process the differential inverter power, each SEPIC converter's power rating equals one-third of the total inverter power. This exciting feature facilitates using small active and passive components with lower power ratings. It also reduces the size of the power stage. Moreover, the switches operate over one-half of the line-frequency cycle, decreasing the inverter power losses and improving inverter efficiency.

Between different DC-DC converter topologies [111-130], the SEPIC converter is selected in the proposed inverter because it has the following advantages:

 DC-DC SEPIC is a step-up/down converter. It's extremely utilized in many applications like RES [119, 122 and 130], Point of Load (POL) converters [113], DGs [117,120, and 124], and Power Factor Correction (PFC) [115, 127 and 128].

- 2. SEPIC converter provide galvanic isolation using small HFT [116].
- 3. It's a non-inverting buck-boost circuit and simplify the implantation of utilized HFT.
- 4. The input inductor of SEPIC converter validate continuous input current and prevent issues related to short circuit current [115, 124 and 130].
- 5. The input inductor of SEPIC converter eliminates the large electrolyte input capacitor integrated between the PV modules and the inverter. This capacitor is used to decrease the switching ripples and reflected harmonics as discussed in Ch. 2.
- 6. The SEPIC converter has the lowest energy storage passive elements that enhance power density between many isolated buck-boost converter topologies [87, and 91].



Figure 3.1: Proposed single-stage three-phase grid-connected isolated SEPIC-BDI.

3.2.1 Proposed PWM modulation Scheme of SEPIC-BDI

The unipolar operation of most DC-DC converters generates DC voltage even at a variable duty cycle. Since the differential inverter is constructed from three DC-DC converters, the PWM modulation should generate a sinusoidal output voltage imposed on a DC voltage component. This voltage can be issued using sinusoidal duty cycle d_x changing over the utilized DC-DC converter's operating range. The other two DC-DC converters' duty cycle is displaced with 120°, which in turn produces three-phase output voltages on the output terminals. The DC voltage components of all DC-DC converters are ideally equal. Therefore, they cancel each other at the output terminals due to the differential connection. Finally, the AC voltage components of all converters appear on the grid terminals and inject pure alternating currents.

This concept of modulation is called Continues Modulation Scheme (CMS). It is straightforward with quick implementation because the PWM of the conventional DC-DC converter circuits does not change. The PWM of DC-DC converters utilized at different applications such as POL converters and PFC circuits have the same configuration. In these applications, once the desired output voltage is expressed, the duty cycle becomes steady. It only changes at transient conditions and control compensation for the desired output voltage. For the differential inverter case, the duty cycles change continuously over the grid line-frequency cycle.

CMS PWM modulation, illustrated in Fig. 3.2, has been used in many differential inverter topologies [85, 88, 89, 91-103, and 108]. It drives the switches of the utilized DC-DC converters by a sinusoidal duty cycle modulated on the grid line-interval. It is given as $T_L = 1/F_{line}$. Where F_{line} represents the grid line-frequency. The utilized DC-DC converter switches must operate at a high switching-frequency to get higher power density. This is accomplished by obtaining a triangle signal operated at switching interval $T_s = 1/F_{sw}$. Here F_{sw} presents the switching frequency, as illustrated in Fig. 3.2. Therefore, the gate signals of the utilized DC-DC converter switches generate from comparing the variable duty cycle (operated at F_{line}) with the saw-tooth signal (operated at F_{sw}). The originality of the CMS PWM backs to the conventional VSI modulation. The research society uses a sinusoidal waveform compared with the inverter modulation index. That is why CMS utilizes a variable sinusoidal duty cycle. However, this modulation is better for

step-down or bucking topology.

However, using this modulation with buck-boost DC-DC converters introduces a mismatch between the actual output voltage and its reference signal [86-87, 91]. The source of this mismatch is the dynamic behavior of the passive elements (such as inductors and capacitors), in the DC-DC converters, toward this modulation. Technically, this mismatch produce low-order harmonics such as third-order, fourth-order, and fifth-order components on the output terminals of DC-DC converters.

DC-DC converters that have similar characteristics to VSI. This modulation gives good

results with differential inverters based on buck converters. The VSI inverter is a voltage

To solve this issue, additional controller loops at third, fourth, and fifth order compensators can be designed to mitigate its effect form the grid-current. However, this

solution is hard to implement because it disturb the stability of the inverter by limiting the overall bandwidth of the proposed control. Therefore, many linearization strategies have been developed in many works to mitigate the source of this mismatch. In [99-100], a Linearization strategy, between the static gain of differential inverter and its output voltage, is proposed for the boost and buck-boost converters based differential inverter.



Figure 3.2: Continuous Modulation Scheme CMS of differential inverters.

Converter Topology	Voltage gain $\frac{v_{ox}}{v_{dc}}$
DC-DC Buck Converter	d_x
DC-DC Buck-Boost Converter	$\frac{1}{1-d_x}$
DC-DC Boost Converter	$\frac{d_x}{1-d_x}$

Table 3.1: Voltage gain of different basic DC-DC converters.

In contrast, this issue can be easily solved by comparing the characteristics of buck, buck-

boost as well as boost converters, which is based on the voltage gain as obtained in Table 3.1. It is worth noticing that all DC-DC converter's output voltage gain depends on the duty cycle and the DC input voltage. So the output voltage mismatch resulted from sinusoidal duty cycle can be easily mitigated by emerging the voltage gain of utilized DC-DC converters in CMS.

This step modulate the variable duty cycle to make it compatible with the utilized DC-DC converter. This proposal is developed in [86-87 and 89]. The voltage ratio of the utilized buck-boost converters is emerged in the duty cycle calculations. It is applied for five different three-phase buck-boost differential inverter topologies. In [89], it's developed by adding a DC voltage feedforwarding signal at three-phase and single-phase Cuk-based differential inverters.

Therefore, the proposed Modified CMS modulation (MCMS) which have duty cycle calculation of buck-boost SEPIC converter is presented in the proposed SEPIC-BDI. It is derived from the duty-voltage ratio of the SEPIC converter as:

$$d_x = \frac{V_{ox}}{V_{ox} + V_{dc}} \tag{3.1}$$

Fig. 3.3 illustrates the distinction between the conventional CMS and proposed MCMS PWM modulation. Its successfully mitigate the mismatch of the output voltage by modifying the calculated duty cycle. Moreover, it does not need any feedforwarding signals like [89] which reflects its simplicity and feasibility at the proposed SEPIC-BDI.



Figure 3.3: Duty cycle calculation at CMS and MCMS schemes for differential inverters.

3.2.2 Operational Principals of Proposed SEPIC Converters in Differential Inverter

Fig. 3.4 shows the operational waveforms of the proposed SEPIC-BDI for one phase using the proposed MCMS scheme. M represents the static gain of the proposed inverter.

 V_{dc} and $v_{ox}(t)$ present DC input voltage and variable AC output voltage of SEPIC converter x. Similarly, $i_{inx}(t)$ and $i_{sx}(t)$ express the input and output current of the SEPIC converter x. When the instantaneous AC output voltage is larger than the DC input voltage, the DC-DC SEPIC converter process the power of one phase in two directions: forward and reverse power (circulating power) [89, 94].

Fig. 3-5 exhibits the equivalent circuit of one isolated SEPIC converter x and its operational modes at both frequency states (line-frequency and switching-frequency). For proper operation, the SEPIC converter should processes both powers and validate the smooth operation of the differential inverter and the MCMS modulation. The complimentary switching of the main switch S_{mx} and body diode of the synchronous switch S_{rx} processes the forward power as depicted in Fig. 3.5-a. The reverse power, proceeding the circulating power between the parallel SEPIC converters is implemented by using the complementary switching operation of the synchronous switch S_{rx} and the body diode of the main switch S_{mx} . The SEPIC converter achieves both powers according to the passive elements' energy storage and energy release mechanisms [116]. High frequency states of SEPIC converter switches maintain these mechanisms, as demonstrated in Fig. 3.5-b, c for forward power operation.



Figure 3.4: Operational waveforms of the proposed single-stage three-phase isolated grid-tied SEPIC-BDI.



b- Forward power switching state of SEPIC converter for dx

im) 	Cx	1:n		isx
Vdc	Smx dx			336	$1 - d_x \frac{S_{rx}}{C_{ox}}$	+ Vox + _

c- Forward power switching state of SEPIC converter for (1-dx)Figure 3-5: Schematic of proposed SEPIC converter and its operational states for line and switching-frequency.

Due to the symmetrical operation, forward power operation is analyzed in this chapter. During the first interval, $d_x T_s$ shown in Fig. 3.5-b, the main switch S_{mx} is on to charge the inductance L_x of input inductor from the DC source. The stored energy of the coupling capacitor C_x is discharged also at the magnetizing inductance of HFT. At the secondary side of HFT transformer, the output capacitor C_{ox} discharges its energy at load. During the second interval, $(1 - d_x)T_s$ turning off the main switch enables the series connection of input inductor, coupling capacitor and magnetizing inductance of HFT transformer. This connection lets the charged energy of the input inductor L_x and the power of the DC source, to charge the coupling capacitor C_x and the magnetizing inductance moving across HFT transformer to charge the output capacitor C_{ox} and load, benefiting from the on-state of the body diode of the synchronous switch S_{rx} as shown in Fig. 3.5-c.

Applying DC averaging over on complete switching cycle T_s provides the voltage gain of the SEPIC converter that given as [116]:

$$\frac{V_{ox}}{V_{dc}} = \frac{nd_x}{1 - d_x} \tag{3.2}$$

Where n represents the turn's ratio of utilized HFT. The voltage and the current formulas for inductors and capacitors of the SEPIC converter are illustrated at Table. 3.2. The switches, its ripple components and its accompanying stresses, are derived. The voltages and currents of the passive elements are also outlined. It worth noticing that these parameters define the size of the SEPIC converter, its elements ratings and, the whole size of the proposed SEPIC-BDI.

components.							
Component	Voltage	Current	Ripple component				
L_x		$\frac{ni_{sx}d_x}{1-d_x}$	$\frac{d_x(t)V_{dc}}{2L_xF_{sw}}$				
L _{mx}		ni _{sx}	$\frac{d_x(t)V_{dc}}{2L_{mx}F_{sw}}$				
C_x	V _{dc}		$\frac{ni_{sx}d_x(t)}{2C_xF_{sw}}$				
Cox	v _{ox}		$\frac{i_{sx}d_x(t)}{2C_{ox}F_{sw}}$				
S _{mx}	$V_{dc} + \frac{v_{ox}}{n}$	$\frac{ni_{sx}}{1-d_x}$	$\frac{d_x(t)V_{dc}}{2(L_x+L_{mx})F_{sw}}$				
S _{rx}	$V_{dc} + nv_{ox}$	$\frac{ni_{sx}}{1-d_x}$	$\frac{d_x(t)V_{dc}}{2(L_x+L_{mx})F_{sw}}$				

Table 3.2: Isolated SEPIC converter components with its operational voltages, current and ripple

3.2.3 Mathematical Analysis of Proposed SEPIC-BDI

For the convenience of analysis, the balanced three-phase voltages at the grid side, shown in Fig. 3.1, are expressed as:

$$\begin{bmatrix} e_{su}(t) \\ e_{sv}(t) \\ e_{sw}(t) \end{bmatrix} = e_m \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin(\omega t + \frac{2\pi}{3}) \end{bmatrix}$$
(3.3)

Where e_m is the peak phase voltage and, ω is the angular frequency. Time-variant output voltage on each SEPIC converter can be generated using variable duty cycle as portrayed in proposed MCMS, see Fig. 3.4. The duty cycles are displaced with a 120°

phase shift angle to generate three-phase voltages. Therefore, according to equation (3.1), the duty cycle of the SEPIC converters is formulated as:

$$\begin{bmatrix} d_u(t) \\ d_v(t) \\ d_w(t) \end{bmatrix} = \frac{n * M(1+K)}{n * M(1+K)+1}$$
(3.4)

The term k presents the behavior of the grid and given, for three-phase grid as:

$$K = \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix}$$
(3.5)

It's interesting to notice that equation (3.4) considers the static linearization without any feedforward signals and mitigates low-order harmonics which discussed in the previous section.

The output voltages of the SEPIC converters v_{ou} , v_{ov} , v_{ow} , have two different components; a line-frequency sinusoidal component e_m , synchronized with the grid side and a DC offset component v_{do} , which generated from the changeable duty cycle and the SEPIC converters unipolar operation. They are expressed as follows:

$$\begin{bmatrix} v_{ou}(t) \\ v_{ov}(t) \\ v_{ow}(t) \end{bmatrix} = v_{do} + e_m \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix}$$
(3.6)

Owing to the differential connection of SEPIC modules at the grid side as presented in Fig. 3.1. At the line-to-line terminal voltages, these DC offset voltages cancel each other, and in turn, decouple the DC offset voltage from three-phase grid voltages.

Since both components results from SEPIC converters, the output voltages of SEPIC converter, derived at (3.6) can be rewritten as:

$$\begin{bmatrix} v_{ou}(t) \\ v_{ov}(t) \\ v_{ow}(t) \end{bmatrix} = nMV_{dc} + nMV_{dc} \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix}$$
(3.7)

It worth noticing that, according to equation (3.7), v_{do} equals e_m and, the voltage gain of the differential inverter depend on static gain M and the turn's ratio of HFT n.

3.3 Proposed Mathematical Model for NSHC of Differential Inverters

The mathematical analysis presented in the previous section deals with the fundamental equations of the differential inverters. It considers only the ideal waveforms, proposed MCMS scheme, and considers ideal SEPIC converters. However, any buck-boost converters (including SEPIC converters) or boost converters introduce low-order harmonics at the output current due to the paralleled DC-DC converters at differential inverters.

The parallel operation of utilized SEPIC converters introduces a reverse power in between, as depicted in Fig. 3.6. When the output AC voltage of the SEPIC converter larger than DC input voltage, this reverse power does not go back to input DC source. Nevertheless, it circulates between the DC-DC converters as shown in Fig. 3-6. One complete cycle of the duty cycle, output current, input current, and output voltage of three SEPIC converters is presented to show this issue.



Figure 3.6: Mathematical representation of duty cycle, output voltage, output current and input current of the SEPIC-BDI.

For the first SEPIC converter, this cycle is divided into three different periods. In the first period T_1 , the duty cycle d_u is high. Hence, the input DC current and the output grid AC current (i_{inu} and i_{su}) are positive. The converter power flows from the DC source to the grid. For the second period T_2 , the duty cycle d_u is low and the currents
$(i_{inu} \text{ and } i_{su})$ are negative. It means that the power moves from the other two SEPIC converters and circulates into this SEPIC converter. The same issue is for the third period T_3 . However, the difference between T_2 and T_3 is the origin of the current. In T_2 , the current flows from the second SEPIC converter because the duty cycle d_v is high. In T_3 , the current flows from the third SEPIC converter because the duty cycle d_w is high. In the last two periods, the current rotates through first SEPIC converter two times in every cycle. Moreover, the same operation is repeated for the other two SEPIC converters.

Here, two important points should be interpreted:

1- There is a circulating power between three DC-DC converters of the three-phase differential inverter.

2- It is periodically happened two times in every line-cycle.

According to this circulating power, a Second-order Negative Sequence NSHC is imposed on the differential inverters output current, whatever the utilized DC-DC converter.

Looking at literature work, the NSHC of the differential inverters was studied in many papers. It covered many buck-boost DC-DC converters such as SEPIC, Cuk, and flyback converter [86-87, 89, 96 and 106]. In all, the differential inverter's results still have this harmonic component on the output current even in applications rather than grid-connected applications such as standalone applications. Moreover, the most published works state an inverse relationship between NSHC and the passive components of associated DC-DC converters (inductors and capacitors). Although huge passive components damp it a little bit, this solution is not accepted. It adds more size, weight, and cost for the differential inverters and hinders its practical applications. Therefore, all papers solve this issue by adding supplementary control loop to the primary loop (voltage control or current control). The proposed additional control loop mitigate it entirely from the output current without adding complexity to the differential power stage.

The Cuk-based differential inverter proposed in [86] adds a control loop for NSHC mitigation. The main loop, introduced for grid-current control, could not mitigate NSHC even at the proposed static linearization strategies. Therefore, in this loop, a band-path filter was presented to extract NSHC from the output current. The proposed Proportional

Resonance (PR) compensator is then designed to modulate its error signal and add it to the primary control signal. The same approach is implemented for grid-connected differential inverter topologies based on five different DC-DC buck-boost converters, as presented in [87].

Unlike the CMS scheme, which was proposed in [86-88, 91], three-phase and singlephase Cuk-based differential inverters using Discontinuous Modulation Scheme (DMS) was presented in [89, 94] and [104-105], respectively. This modulation disconnects the variable duty cycle for a while (about one-third of the line-cycle). This technique has decreased the circulating currents between different Cuk converters in both topologies, thereby circulating power loss, and NSHC will be decreased. However, the DMS modulation scheme does not completely remove the whole NSHC, especially for threephase differential inverter, but decreases its amplitude and still needs an extra loop.

Also, DMS scheme adds odd harmonic component (especially third-order) that needs further control loop and complicates the overall control performance [94].

Although the previous literature has improved the performance of differential inverter topologies by mitigating NSHC. They successfully investigated the NSHC and proposed the additional loop for its compensation. The following points are common to all:

- There is no general NSHC mathematical model that investigates and accurately detects its real source.
- Detailed design of the utilized DC-DC converters, its active and passive elements are not well covered by including the effect of NSHC.

In this chapter, an in-depth mathematical model is derived. This model investigates the differential inverter practical issues such as variable duty cycle, low-order harmonics, and circulating power. For this purpose, the grid current of differential inverters is assumed as:

$$\begin{bmatrix} i_{su}(t)\\ i_{sv}(t)\\ i_{sw}(t) \end{bmatrix} = i_m \begin{bmatrix} \sin(\omega t)\\ \sin\left(\omega t - \frac{2\pi}{3}\right)\\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix} + i_x \begin{bmatrix} \sin(h\omega t)\\ \sin\left(2\left(ht - \frac{2\pi}{3}\right)\right)\\ \sin\left(h\left(\omega t + \frac{2\pi}{3}\right)\right) \end{bmatrix}$$
(3.8)

Where i_x and h represent the absolute value of the assumed unknown low-order NSHC and its order, respectively. Input current of each SEPIC converter can be formulated by multiplying voltage ratio of SEPIC converter (3.2) and grid current assumed at (3.8) as follows:

$$\begin{bmatrix} i_{inu}(t) \\ i_{inv}(t) \\ i_{inw}(t) \end{bmatrix} = 0.5nMi_m + nMi_m \begin{bmatrix} \sin(wt) \\ \sin\left(wt - \frac{2\pi}{3}\right) \\ \sin\left(wt + \frac{2\pi}{3}\right) \end{bmatrix} - 0.5 \ nMi_m \begin{bmatrix} \cos(2wt) \\ \cos\left(2wt - \frac{4\pi}{3}\right) \\ \cos\left(2wt + \frac{4\pi}{3}\right) \end{bmatrix} + nMi_x \begin{bmatrix} \sin(hwt) \\ \sin\left(h\left(wt - \frac{2\pi}{3}\right)\right) \\ \sin\left(h\left(wt - \frac{2\pi}{3}\right)\right) \\ \sin\left(h\left(wt + \frac{2\pi}{3}\right)\right) \end{bmatrix} + 0.5nMi_x \begin{bmatrix} \cos((1-h)wt) \\ \cos\left((1-h)\left(wt - \frac{2\pi}{3}\right)\right) \\ \cos\left((1-h)\left(wt + \frac{2\pi}{3}\right)\right) \end{bmatrix} - \cos\left((1-h)\left(wt + \frac{2\pi}{3}\right)\right) \end{bmatrix} - 0.5nMi_x \begin{bmatrix} \cos((1+h)wt) \\ \cos\left((1+h)\left(wt - \frac{2\pi}{3}\right)\right) \\ \cos\left((1+h)\left(wt - \frac{2\pi}{3}\right)\right) \end{bmatrix}$$
(3.9)

Based on (3.9), the input power of the proposed SEPIC-BDI can be obtained as follows:

$$P_{in} = V_{dc}I_{dc} = V_{dc}(i_{inu} + i_{inv} + i_{inw}) = \frac{3nMi_mV_{dc}}{2} + nMi_xV_{dc}\left[sin(hwt) + sin\left(h\left(wt - \frac{2\pi}{3}\right)\right) + sin\left(h\left(wt + \frac{2\pi}{3}\right)\right)\right] + 0.5nMi_xV_{dc}\left[cos((1-h)wt) + cos\left((1-h)\left(wt - \frac{2\pi}{3}\right)\right) + cos\left((1-h)\left(wt + \frac{2\pi}{3}\right)\right)\right] - 0.5nMi_xV_{dc}\left[cos((1+h)wt) + cos\left((1+h)\left(wt - \frac{2\pi}{3}\right)\right) + cos\left((1+h)\left(wt + \frac{2\pi}{3}\right)\right)\right]$$
(3.10)

Similarly, the output power of the proposed SEPIC-BDI can be obtained, based on (3.3) and (3.8), as follows:

$$P_{o} = e_{m} \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin(\omega t + \frac{2\pi}{3}) \end{bmatrix} * \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix} + i_{x} \begin{bmatrix} \sin(h\omega t) \\ \sin\left(2\left(ht - \frac{2\pi}{3}\right)\right) \\ \sin\left(h\left(\omega t + \frac{2\pi}{3}\right)\right) \end{bmatrix}$$
(3.11)

The output power in (3.11) can be simplified as follows:

$$P_{o} = \frac{3e_{m}i_{m}}{2} + \frac{e_{m}i_{x}}{2} \left[\cos\left((1-h)wt\right) + \cos\left((1-h)\left(wt - \frac{2\pi}{3}\right)\right) + \cos\left((1-h)\left(wt + \frac{2\pi}{3}\right)\right) \right] - \frac{e_{m}i_{x}}{2} \left[\cos\left((1+h)wt\right) + \cos\left((1+h)\left(wt - \frac{2\pi}{3}\right)\right) + \cos\left((1+h)\left(wt + \frac{2\pi}{3}\right)\right) \right]$$
(3.12)

Assuming lossless DC-DC SEPIC converters, input power and output power are equal that enable the following condition:

$$nMi_{x}V_{dc}\left[sin(hwt) + sin\left(h\left(wt - \frac{2\pi}{3}\right)\right) + sin\left(h\left(wt + \frac{2\pi}{3}\right)\right)\right] = 0$$
(3.13)

It worth noticing that the even number of harmonic orders h satisfy the condition in equation (3.13). For example, second or fourth or sixth.....). This finding is very important because the NSHC is at even order. Moreover, considering only the second-order NSHC simplify the theoretical assumptions and closely matches the previous published works [86-87, 89 and 94]. Using this assumption, the practical input current of the SEPIC converters that equals the inductor current L_x can be expressed as:

$$\begin{bmatrix} i_{inu}(t) \\ i_{inv}(t) \\ i_{inw}(t) \end{bmatrix} = 0.5nMi_m + nMi_m \begin{bmatrix} sin(wt) \\ sin\left(wt - \frac{2\pi}{3}\right) \\ sin\left(wt + \frac{2\pi}{3}\right) \end{bmatrix} + 0.5nMi_x \begin{bmatrix} cos(-wt) \\ cos\left(-wt + \frac{2\pi}{3}\right) \\ cos\left(-wt - \frac{2\pi}{3}\right) \end{bmatrix} - 0.5nMi_m \begin{bmatrix} cos(2wt) \\ cos\left(2wt - \frac{4\pi}{3}\right) \\ cos\left(2wt + \frac{4\pi}{3}\right) \end{bmatrix} + nMi_x \begin{bmatrix} sin(2wt) \\ sin\left(2wt - \frac{4\pi}{3}\right) \\ sin\left(2wt + \frac{4\pi}{3}\right) \end{bmatrix} - 0.5nMi_x \begin{bmatrix} cos(3wt) \\ cos\left(3wt - \frac{6\pi}{3}\right) \\ cos\left(3wt + \frac{6\pi}{3}\right) \end{bmatrix}$$
(3.14)

According to (3.14), the input of the SEPIC converter which integrated at differential inverter considering second-order NSHC includes the following components:

- ✓ DC offset current component, representing the shared average current of each SEPIC module from the DC source.
- ✓ Line-frequency component, representing the fundamental sinusoidal waveform imposed on the DC component and declares the injected current injection in the grid side.
- ✓ Second-order component, representing the circulating currents at reverse power operation and generates the NSHC imposed on grid-current.

✓ Third-order component, representing the component that generates from the current interaction between the reflected NSHC and the circulating currents.

With NSHC, the input and the output current of SEPIC modules are affected according to equations (3.8, 3.14). Here, it's very important to notice that the current of switches, inductors, and HFTs of SEPIC converters depends on them. Moreover, this effect continues to the input current of the DC input source based on the following expression:

$$I_{dc} = i_{inu} + i_{inv} + i_{inw} = \frac{3}{2}nMi_m + \frac{3}{4}nMi_x \cos(3wt)$$
(3.15)

According to (3.15), the DC input current is distorted with a third-order component which has many harmful effects, especially for RES.

Regarding this, the features of the proposed mathematical model are summarized as

- ✓ The real source of NSHC is the modulation scheme (variable duty cycle) and DC-DC converters' unipolar operation such as the SEPIC converter. They generate the circulating power and introduce NSHC on the differential inverter output current. That is why the unfolding Flyback-based differential inverter does not have NSHC because the converter's output is bipolar [108].
- ✓ Equation (3.15) proves that no extra loop is demanded to mitigate the third-order harmonic component because the NSHC is the source of this component. Therefore, the NSHC compensation loop on the grid side is enough. This new finding is very important because it simplifies the control structure of the differential inverters.
- ✓ This model successfully explicates why the DCM modulation scheme does not wholly mitigate NSHC. It interestingly decreases the circulating current between the utilized DC-DC converters. However, it's worth noticing that the DMS modulation scheme moves part of the NSHC from second-order to the fourth-order [89].

3.4 Design Example of the Proposed SEPIC-BDI

The proposed SEPIC-BDI is assumed to convert 1.6kW to a three-phase utility grid of 200V line-line RMS voltage from a 100V DC source. The line-frequency and the switching-frequency is assumed $F_{line} = 60Hz$ and $F_{sw} = 50kHz$. The grid inductance is also presented and assumed as $L_{gx} = 4mH$ as shown in Fig. 3.1. As mentioned before, every SEPIC converter process one-third of the inverter power. Therefore, the power of each SEPIC converter is obtained as:

$$P_{con} = \frac{P_{inv}}{3} = \frac{1600}{3} = 533.33W \tag{3.16}$$

Based on this, the RMS output current of every SEPIC converter is given as:

$$i_{sx} = \frac{P_{con}}{e_m} = \frac{533.33}{200/(\sqrt{3})} = 4.6187A$$
(3.17)

The variable duty cycle of the proposed SEPIC-BDI is changes periodically from $d_x = 0.1 - 0.8$ over the line-frequency cycle to realize the proposed MCMS modulation scheme. From Table 3.2, the maximum input inductor current is obtained as:

$$i_{inx} = i_{Lx} = \frac{ni_{sx}d_x}{1-d_x} = \frac{1*\ 4.6187*\sqrt{2}*0.8}{1-0.8} = 26.1273A$$

(3.18)

Similarly, the maximum current of magnetizing inductance is obtained as:

$$i_{Lmx} = ni_{sx} = 1 * 4.6187 * \sqrt{2} = 6.5318A \tag{3.19}$$

Using the same approach, the coupling capacitor voltage is expressed as:

$$V_{Cx} = V_{dc} = 100V (3.20)$$

In contrast, the maximum voltage of the output capacitor is expressed as:

$$V_{Cox} = v_{ox} = v_{do} + e_m = 2 * \frac{200}{\sqrt{3}} * \sqrt{2} = 326.598V$$
(3.21)

3.4.1 The Parameters Selection and the Design Criteria

The variable duty cycle introduces variable current and voltage ripples. The ripple window for both is expressed by the available duty range. However, the important limit is the high duty $d_x = 0.8$. The percentage of peak to average value of current ripple in input inductor L_x and magnetizing inductance of HFT L_{mx} is assumed 20% and 25%, respectively [116]. Therefore, the input inductance is expressed as:

$$L_x = \frac{d_x V_{dc}}{2\Delta i_{lx} F_{sw}} = \frac{0.8*100}{2*0.20*26.1273*50000} = 153.0965\mu H$$
(3.22)

Similarly, the magnetizing inductance is expressed as

$$L_{mx} = \frac{d_x V_{dc}}{2\Delta i_{lmx} F_{sw}} = \frac{0.8*100}{2*0.25*6.5318*50000} = 489.9109\mu H$$
(3.23)

The percentage of peak to average value of voltage ripple in coupling capacitor C_x and output capacitor C_{ox} is assumed 5% and 3%, respectively. Therefore, the capacitance of coupling capacitor is expressed as:

$$C_{x} = \frac{ni_{sx}d_{x}}{2\Delta V_{Cx}F_{sw}} = \frac{1*6.5318*0.8}{2*0.05*100*50000} = 10.4509\mu F$$
(3.24)

Also, the capacitance of the output capacitors is expressed as:

$$C_{ox} = \frac{ni_{sx}d_x}{2\Delta V_{\text{Cox}}F_{sw}} = \frac{1*6.5318*0.8}{2*0.03*326.598*50000} = 5.333\mu F$$
(3.25)

The final chosen parameters of the proposed SEPIC-BDI are summarized in Table. 3.3. It worth noticing that, the inductances and capacitances is larger than the calculated values in equations (3.22: 3.25). This is important to compensate the voltage drop of the parasitic resistances of the selected passive components. For the switches devices, the maximum current stress of the main switch is expressed as:

$$i_{Smx} = \frac{ni_{Sx}}{1 - d_x} = \frac{1 + 4.6187 + \sqrt{2}}{1 - 0.8} = 32.6591A \tag{3.26}$$

The maximum current ripple of the main switch is also expressed as:

$$\Delta i_{Smx} = \Delta i_{lx} + \Delta i_{lmx} = (0.25 + 0.20) * 4.6187 * \sqrt{2} = 2.9393A$$
(3.27)

This means that the peak current of the main switch is:

$$i_{Smx}(peak) = 32.6591 + 2.9393 = 35.5984A$$
 (3.28)

For the synchronous switch, the maximum current stress is obtained as:

$$i_{Srx} = \frac{ni_{Sx}}{1 - d_x} = \frac{1*4.6187*\sqrt{2}}{1 - 0.8} = 32.6591A$$
(3.29)

Table 3.3: The designated parameters of the proposed isolated SEPIC-BDI.

component	value	Unit
Rated Power, P_{inv}	1600	W
Input DC Voltage, V_{dc}	100	V
Grid Voltage, v_{ll} , ω	200, $2\pi * 60$	V, rad/
		sec
Switching Frequency, F _{sw}	50	kHz
Input inductor, L_x	180	μΗ
HFL transformer inductance, L_{mx}	500	μΗ
HFL transformer turns ratio n	1	
Coupling capacitor, C_x	14	μF
Output capacitor, C_{ox}	14	иF
Grid inductance, L_{gx}	4	mH

The maximum current ripple of the synchronous switch is also expressed as:

$$\Delta i_{Srx} = \Delta i_{lx} + \Delta i_{lmx} = (0.25 + 0.20) * 4.6187 * \sqrt{2} = 2.9393A$$
(3.30)

This means that the peak current of the synchronous switch is:

$$i_{Srx}(peak) = 32.6591 + 2.9393 = 35.5984A \tag{3.31}$$

The voltage stress peak of the main switch is calculated as:

$$V_{Smx} = V_{dc} + \frac{v_{ox}}{n} = 100 + \frac{326.598}{1} = 426.598V$$
(3.32)

The maximum voltage ripple is obtained as:

$$\Delta V_{Smx} = \Delta V_{Cx} + \Delta V_{Cox} = (0.05 * 100) + (0.03 * 326.598) = 14.7979V$$
(3.33)

This means that the peak voltage of main switch is:

$$V_{Smx}(peak) = V_{Smx} + \Delta V_{Smx} = 426.598 + 14.7979 = 441.3959V$$
(3.34)

Following the same approach for synchronous, the maximum voltage stress is:

$$V_{Srx} = V_{dc} + nv_{ox} = 100 + (1 * 326.598) = 426.598V$$
(3.35)

The maximum voltage ripple is obtained as:

$$\Delta V_{Smx} = \Delta V_{Cx} + \Delta V_{Cox} = (0.05 * 100) + (0.03 * 326.598) = 14.7979V$$
(3.36)

This means that the peak voltage of main switch is:

$$V_{Smx}(peak) = V_{Smx} + \Delta V_{Smx} = 426.598 + 14.7979 = 441.3959V$$
(3.37)

By considering the voltage and the current peaks of the main and synchronous switches that derived in previous equations. The maximum switch current and the maximum voltage will be larger than 35.5984*A* and 441.3959*V* because there is a voltage drop introduced from the SEPIC converter elements' parasitic resistances. Therefore, SiC devices with a part number C2M0040120D are selected for both switches of the proposed SEPIC converter. It is worth mentioning that this stress happens at the forward power cycle. Therefore, the main switch and body diode of the synchronous switch should handle this current and voltage stress. Furthermore, SiC devices are preferred because they have low conduction and low switching losses, implying high efficiency. Moreover, they have a high blocking current and voltage capability to withstand the current and voltage strength of the proposed SEPIC-BDI.

Fig. 3.7 illustrates the designated PCP board of the utilized SEPIC DC-DC converter, for reliable operation, film capacitors with part number (C4AEOBW5140A3JJ) are selected for both the coupling and output capacitance. Also, film capacitors is used for the clamp capacitor which is introduced as a part of clamping circuit to mitigate the high voltage

spikes over switches which come from the leakage inductance of HFTs. Also, Heatsink is introduced for stable heat dissipation. Finally, the gate-drives which drive SIC switches is implemented on the same board to validate the modular construction of the proposed SEPIC-BDI.



Figure 3.7: PCB board implementation of one SEPIC converter utilized at the proposed SEPIC-BDI.

In contrast, superior materials suitable for the HFT and input inductor design are monocrystalline, amorphous, and ferrites. For the proposed SEPIC converters, a soft ferrite core (EER-94) is used, for both to achieve low loss and cost [131-132]. The permeability of this core is $\mu_r = 2500$. $A_e = 712mm^2$ and $V_e = 158000mm^3$ are the effective area and effective volume, respectively. The maximum flux density is $B_m = 470mT$. To reduce the magnetic eddy currents and the hysteresis losses, LITZ wire has been used for their windings.

For many years, the basic magnetic theory of the traditional line-frequency transformers was used to design HFTs. The turns of HFT can be expressed using equation (3.38). This equation assumed that the primary and secondary windings of an HFT behave like inductors [133-136], and the magnetic material is composed of one heterogeneous structure. Moreover, it designs HFTs as an independent block by considering only the magnetic elements of an HFT, such as the core geometry and number of turns.

$$N = \sqrt{\frac{L_{mx} * V_e}{\mu_o * \mu_r * A_e}} \tag{3.38}$$

Recently, most isolated DC-DC converters have intricate magnetic structures, consist of many windings and diversified heterogeneous elements (air gaps as well as cores). These

converters require air gaps to store energy during the complementary switching operation [116]. For the proposed SEPIC-BDI, the SEPIC converter has a variable terminal voltage at the HFT due to its variable duty cycle. Therefore, the number of turns can be obtained as follows:

$$N = \frac{V_{on}*(1-D)*T_s}{\Delta B_m*A_e} \tag{3.39}$$

Equation (3.39) considers the circuit parameters of the SEPIC converter, as well as the core geometry. The circuit parameters is introduced by the maximum volt-second V_{on} , on the primary turns of the HFT when the main switch S_{mx} is off. From (3.39), the number of turns is 9. To mitigate the reduction in magnetizing inductance coming from existed air gap, 15 turns is used in the HFT design.

On the other hand, the number of turns for input inductors can be expressed as follows:

$$N = \frac{L_x * I_{pk}}{A_e * \Delta B} \tag{3.40}$$

 I_{pk} is the maximum current of the inductor before saturation. As the maximum current of input inductor is 26.1273*A*. $I_{pk} = 35A$ is selected for the maximum saturation current. From (40), the turn's number for the input inductor is 32. Practically, for higher saturation current, the turn's number is increased to 37 in the proposed SEPIC converter. The implemented HFT and input inductor of the proposed SEPIC module are manifested in Fig. 3.8. Finally, the experimental measurable values of the proposed SEPIC converters including the parasitic components are illustrated at Table. 3.4.



Designed HFT Designed Input inductor **Figure 3.8:** Photographs of implemented passive elements of proposed SEPIC-BDI.

component	specifications		
Input Inductor	L_x	181.9µH	
	R _{dcr}	200mΩ	
	core	EER-94	
	No. of turns	37	
HFL transformer	L_{mx}	493.18µH	
	R_{pri}	50mΩ	
	R _{sec}	$65 \mathrm{m}\Omega$	
	core	EER-94	
	Turns ratio 1:n	1:1	
	No. of turns	15	
	(primary)		
Switches	R _{dson}	40mΩ	
	V_F	1.8v	
	t _{rise}	52ns	
	t_{fall}	34ns	
Capacitors	C_x	C4AEOBW5140A3JJ	
	C_{ox}	C4AEOBW5140A3JJ	
	R _{esr}	5.4mΩ	
Grid inductance,	L _{gx}	4mH	
	R _{gx}	0.2Ω	

Table 3.4: Experimental Parameters of proposed Isolated SEPIC-BDI.

3.4.2 Small-Signal Model of the Proposed SEPIC Converters

The provided SEPIC converter is a buck-boost converter with four passive elements (two inductances and two capacitances) [137]. These passive elements decide the dynamic response of the SEPIC-BDI. The open-loop transfer function of the SEPIC converter generally a fourth-order transfer function [138-142]. A dynamic, accurate model representing the following variables (i_{Lx} , i_{Lmx} , v_{Cx} , v_{ox}) should be derived to design a reasonable control for SEPIC-BDI, shown in Fig. 3.1.

A linearization strategy of the SEPIC converter dynamic response using the AC Modeling approach is obtained to solve this issue [116]. It is based on the SEPIC operating modes at switching frequency that is illustrated in Fig. 3.5. This approach is the simplest one

between other methods such as switch averaging and the state-space averaging. It is worth mentioning that the dynamic response in forward and reverse power is similar because the voltage gain and operating modes are identical. Averaging the previous variables over one complete cycle of switching operation (at forward power) and linearizing it over the quiescent operating point result the following equations:

$$L_{x}\frac{d}{dt}i_{Lx}^{\infty}(t) = V_{dc}^{\infty}(t) - (1 - d_{x})v_{Cx}^{\infty}(t) - (1 - d_{x})\frac{v_{ox}^{\infty}(t)}{n} + d_{x}^{\infty}(t)[V_{Cx} + \frac{V_{ox}}{n}]$$

$$L_{mx}\frac{d}{dt}i_{Lmx}^{\infty}(t) = -d_{x}v_{Cx}^{\infty}(t) + (1 - d_{x})\frac{v_{ox}^{\infty}(t)}{n} - d_{x}^{\infty}(t)[V_{Cx} + \frac{V_{ox}}{n}]$$

$$C_{x}\frac{d}{dt}v_{Cx}^{\infty}(t) = d_{x}i_{Lmx}^{\infty}(t) + (1 - d_{x})i_{Lx}^{\infty}(t) + d_{x}^{\infty}(t)[I_{Lmx} - I_{Lx}]$$

$$C_{ox}\frac{d}{dt}v_{ox}^{\infty}(t) = -i_{Sx}^{\infty}(t) + (1 - d_{x})\frac{i_{Lx}^{\infty}(t)}{n} - (1 - d_{x})\frac{i_{Lmx}(t)}{n}$$

$$+ d_{x}^{\infty}(t)[\frac{I_{Lmx} - I_{Lx}}{n}]$$
(3.41)

Where $i_{Lx}(t)$ is the small-signal component of inductor current $i_{lx}(t)$. $v_{Cx}(t)$ represents the small-signal component of the coupling capacitor voltage $v_{cx}(t)$. $V_{dc}(t)$ is the small signal of DC input voltage V_{dc} . The small-signal components for the rest of the passive elements are also the same. The small-signal AC equivalent circuit model based on the previous equation (3.41) was constructed, as shown in Fig. 3.9. It's assumed that each variable has two distinct components, DC component and AC small-signal component around it and neglect nonlinear terms. The output voltage dynamics is based on the duty cycle and DC input voltage.

To control this converter, the control-to-output transfer equation is derived as:

$$G_{vd}(s) = \frac{v_{ox}(s)}{d_x(s)} = G_0 \frac{a_3 s^3 + a_2 s^2 + a_1 s + a_0}{s^4 + b_3 s^3 + b_2 s^2 + b_1 s + b_0}$$
(3.42)

Where, G_o is the dc gain of SEPIC converter, $a_o - a_3$ represent the constants of zeros, and $b_o - b_3$ are the constants of its poles. The coefficients of these constants which depend on circuit parameters are presented in Table 3.5.

SEPIC-BDI.		
coefficient	value	
G _o	$\frac{nd_x}{1-d_x}$	
<i>a</i> ₃	$\frac{-i_{sx}}{v_{ox}C_{ox}}$	
<i>a</i> ₂	$\frac{d_x(L_{mx}L_x)}{C_{ox}(L_{mx}+L_x)}$	
<i>a</i> ₁	$\frac{d_x i_{sx}}{v_{ox} C_x C_{ox} L_{mx}}$	
<i>a</i> ₀	$\frac{d_x}{C_x C_{ox} L_{mx} L_x}$	
b_3	$rac{i_{sx}}{v_{ox}C_{ox}}$	
<i>b</i> ₂	$\frac{k_a - k_b}{nC_x C_{ox} L_x L_{mx}}$	
k _a	$d_x^2(\mathcal{C}_x + n^2 \mathcal{C}_{ox})(L_{mx} + L_x)$	
k _b	$(2d_x-1)(C_xL_{mx}+n^2C_{ox}L_{mx}+C_xL_x)$	
b_1	$\frac{i_{sx}}{v_{ox}C_{ox}C_{x}} \left[\frac{(1-d_{x})^{2}}{L_{x}} + \frac{d_{x}^{2}}{L_{mx}}\right]$	
b_0	$\frac{(1-d_x)^2}{n^2 C_x C_{ox} L_x L_{mx}}$	

 Table 3.5: Coefficients of the open-loop transfer function of SEPIC converter utilized at proposed

The selected parameters of the utilized SEPIC converter, derived in previous section, is used to draw the open-loop transfer function and define the exact location of zeros and poles and also decide the stability range of the converter. Using high switching frequency enhances the overall size of the SEPIC-BDI by implementing small passive elements. It worth noticing that, the high switching-frequency also improves the performance of the closed-loop control because its effect on the line-frequency response is neglected, especially over the entire range of the variable duty cycle. Fig. 3.10 illustrates the bode plot of the open-loop transfer function of the SEPIC converter.



Figure 3-9: AC equivalent circuit of the SEPIC converter utilized at proposed SEPIC-BDI.

It has been seen that the system is unstable at some values of the duty cycles, which makes the differential inverter prone to instability. Moreover, the phase delay of the SEPIC converter increases at a higher duty cycle. Therefore, the closed-loop control should compensate for the phase delay and stabilizes the inverter transfer function. Regarding this, the control design should be designed according to the worst status of control to the output transfer function, which was at the highest duty cycle $d_x = 0.8$.



Figure 3.10: Bode plot of $G_{vd}(s)$ of SEPIC converter at $(d_x = 0.1 - 0.8 \text{ from bottom to above})$.

3.4.3 Proposed Control Technique

Fig. 3.11 shows the proposed control strategy based on synchronous d-q control, with the proposed harmonic compensator loop. Existing RHP zeros of the SEPIC converters that introduces the phase delay of the converter places limits the maximum allowable phase

margin. Phase lead controller based on PID compensator is proposed to solve this issue [143-147]. The proportional part of the PID compensator accelerates the dynamics of the system. The transfer function of PID compensator is expressed as:

$$G_c(s) = K_p + \frac{K_i}{s} + k_d s \tag{3.43}$$

The proportional gain of the PID K_p increases the DC gain, which enhances the sensitivity of the controller to change and react more quickly. The integral term K_i improves the accuracy of the system by reducing the steady state error. Moreover, the derivative gain decrease its error by comparing it with previous gains. At final, PID control is easy to design, which simplifies the design process and ensures stable operation of the main controller.



Figure 3.11: Proposed control of the SEPIC-BDI.

On the other hand, the harmonic compensator loop is realized by using three cascaded first-order Low Pass Filter (LPFs). The cascaded connection of three LPFs can accurately detect the NSHC from the grid current. As shown in Fig. 3.11, the three-phase grid-current is mutated into the stationary d-q axis frame based on the grid voltage phase angle. The Clark transformation is expressed as:

$$\begin{bmatrix} i_d \\ i_q \end{bmatrix} = \frac{\sqrt{2}}{\sqrt{3}} \begin{bmatrix} \cos(\omega t) & \cos\left(\omega t - \frac{2\pi}{3}\right) & \cos\left(\omega t + \frac{2\pi}{3}\right) \\ \cos(\omega t) & -\sin\left(\omega t - \frac{2\pi}{3}\right) & -\sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix} \begin{bmatrix} i_{su} \\ i_{sv} \\ i_{sw} \end{bmatrix}$$
(3.44)

It worth mentioning that the angular frequency of the line component (positive

component) in the d-q frame is (ω), where the angular frequency of NSHC in the same frame is (-2ω). The three LPFs will assure NSHC attenuation from the grid current. Then the output of LPFs is subtracted from the actual grid current to obtain only the accurate component of NSHC. Finally, the conventional PI controller is used to eliminate the NSHC by modulating each converter's duty cycle. The attenuation gain of the cascaded LPFs is 1. It is worthy to notice that the dynamic action of PI controllers is slow in the second loop. It prevents any distortion on the first loop and, at the same time, does not affect the whole system's bandwidth. Therefore, the output of main loop is transferred to ABC frame using inverse Clark transformation as follows:

$$\begin{bmatrix} M_u \\ M_v \\ M_w \end{bmatrix} = \frac{\sqrt{2}}{\sqrt{3}} \begin{bmatrix} \cos(\omega t) & -\sin(\omega t) \\ \cos\left(\omega t - \frac{2\pi}{3}\right) & -\sin\left(\omega t - \frac{2\pi}{3}\right) \\ \cos\left(\omega t + \frac{2\pi}{3}\right) & -\sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix} \begin{bmatrix} M \\ M_q \end{bmatrix}$$
(3.45)

The same transformation is applied to the NSHC compensation loop. Finally, the odd harmonics resulting from SEPIC converters' nonlinear dynamics have been eliminated using the proposed MCMS modulation.

3.5 Results of the Proposed SEPIC-BDI

3.5.1 Simulation Results

To verify the designated synchronous d-q control and NSHC compensation technique. The SEPIC-BDI circuit and its control technique have been tested using PSIM 11.1.3 software. The simulation parameters is presented in Table 3.3. The control dynamics is verified by varying reference current from 13*A* to 8*A* (2.6 - 1.6kW) before and after NSHC compensation as demonstrated in Fig. 3.12. The grid-current is distorted at a positive peak, which means an NSHC component.

Moreover, the distortion also appears on the controller output by making significant oscillation as shown in Fig. 3.12-a. Fig. 3.12-b reveals the compensation technique which cleared NSHC from the grid output current and control output and optimizes control dynamics.



B- After NSHC compensation.

Figure 3.12: Simulation results of grid phase voltage, output grid current and control parameters.

Fig. 3.13 provides the duty cycle, the input inductor, and the SEPIC converters' output voltages. The distorted waveforms at the input inductor current and the output voltage of every SEPIC module effectively appear before NSHC compensation. It results from the circulating current and the SEPIC converters' unipolar operation, as illustrated in Fig. 3.13-a. The proposed compensation technique effectively cleared this distortion by narrowing the periods of T_2 and T_3 , as illustrated in Fig. 3.13-b.

Finally, the proposed compensation technique's feature over the coupling capacitor voltage, HFT current, and switches current is illustrated in Fig. 3.14.



A-Before NSHC compensation.



B- After NSHC compensation.

Figure 3.13: Simulation results of duty cycle, input inductor currents and output voltages.

3.5.2 Experimental Results

The feasibility of the proposed SEPIC-BDI with the suggested NSHC compensation is verified using a 1.6 KW system setup, as offered in Fig. 3.15. The parameters are provided in Table 3.4. The system parameters, components, and measurement tools are depending on the proposed inverter, which depicted in Fig. 3.1. The variable DC source, which emulates the input DC source, is implemented by variable AC supply, diode bridge rectifier, and input capacitor. The 3-phase AC supply (S3P-240-30) manufactured by Yamabishi company is used in this setup with 0-240 V and 10.4kVA ratings. This power supply is connected to a six-diode bridge-rectifier (75L6P41) manufactured by Toshiba and cascaded with 6000uF to provide pure variable DC

voltage. Another variable 3-phase AC supply (S3P-240-15) with 0-240 V and 5.2kVA rated power is included to simulate the grid-side.



B- After NSHC compensation.

The voltage, current, and power analysis is recorded experimentally by Yokogawa DL850 Scope and Yokogawa WT 1800 power analyzer, respectively. LA 55-P and LV25-P are used to sense three phase currents as well as two grid line-to-line voltages, introducing only five sensors in total. The isolated DC-DC converter (ISE0524A) from XP power is integrated in the designated gate drive to form gate isolation and provide cost effective solution in driving low-side (main switch) and high-side (synchronous switches) using the same driver chip.

The MWPE3 C6713A Expert III DSP board develops the control task of the proposed

Figure 3.14: Simulation results of coupling capacitor voltage, HFT current and switches currents.

control technique. The proposed control is realized using PE-Expert 3 digital control system. It is embedded with TI high-speed floating-point DSP TMS320C6713. Moreover, the MWPE3 Xilinx FPGA board with part number XC3S500E is linked with the DSP board for pulses generation with a carrier frequency of 50 kHz. The integration of previous boards with PE-View9 computer-based interface features easy modifications for the control variables and high performance.



Figure 3.15: Experimental Setup of the proposed single-stage SEPIC-BDI.

Fig. 3.16 illustrates the experimental verifications of grid voltage, grid current, and output voltages of the SEPIC converters utilized at proposed SEPIC-BDI. Suppose the leakage inductance of HFTs is effective. In that case, this means that the percentage of NSHC is high because of the circulating current increases with it. After NSHC compensation, the proposed control technique modulates the duty cycle to clear NSHC and decreases the circulating current even at high leakage inductance.

Fig. 3.17 appears the input inductor currents, DC voltage, and DC current before and after harmonic compensation. In particular, inductor currents and DC current is also effected by circulating current and the existed NSHC. On the other hand, after NSHC, the waveforms is free from distortion.



A-Before NSHC compensation. B- After NSHC compensation. **Figure 3.16:** Experimental results of grid voltage, output grid-current and output voltages of SEPIC-BDI.

Fig. 3.18 shows the FFT harmonic spectrum of the grid current before and after NSHC compensation, respectively. Without harmonic compensation, the grid current is totally distorted, especially in the positive cycle, with NSHC component, which is 41.6% of the RMS line current. Otherwise, the proposed harmonic compensation technique minimized the NSHC to be 1.93% of the total RMS line current, which results in a pure sinusoidal waveform at the grid-current.



Figure 3.17: Experimental results of input currents, DC voltage and DC current of SEPIC-BDI.

The THD of the grid-current is 48.5% without NSHC compensation due to the circulating current and the high leakage inductance, whereas THD of the grid-current is 5.7 % with the proposed NSHC compensation technique.



Figure 3.18: FFT Harmonic Spectrum of grid current before and after NSHC for proposed SEPIC-

BDI.

Chapter 4: SEPIC-BDI: Improved Control with High Performance

4.1 Introduction

The simple circuit based on the DC-DC SEPIC DC-DC converters is the prime attractive feature of the propositioned differential inverter. It has a single-stage which provides great power density. The NSHC introduced at the output of the differential inverter may suppress its practicality. Therefore, the proposed mathematical model successfully investigates NSHC and its issues for the proposed SEPIC-BDI. Moreover, a two-loop-based grid current control is proposed to mitigate NSHC and achieve grid current control effectively. The results of the proposed SEPIC-BDI has two important notices:

- ✓ The THD of the proposed inverter is still more extensive than the standards of the recent grid codes (>5%).
- ✓ The DC input current of the proposed inverter has significant switching harmonics, disturbing the smooth operation of MPPT controllers and it's tracking efficiency and deteriorating PV modules during inverter applications at RES [148-151].

This chapter, motivated by this, presents two modified control techniques (Control-I, Control-II) to enhance the inverter performance and decrease THD under acceptable limits.

- In Control-I, a Type-II compensator, which enhances SEPIC converters' stability, is used instead of the PID controller. The proposed Type-II compensator effectively damp the resonance of the SEPIC converter [152-154]. This resonance comes from the passive elements of the proposed SEPIC converter and increases THD's previous results. In the second loop, the NSHC detection circuit depending on the LPFs is also used.
- In Control-II, the proposed control is redesigned based on the parasitic components of SEPIC converters. According to [155-156], The Right half-Plane (RHP) zeros may moves to the Left Half-Plane (LHP) because of the parasitic resistances of the SEPIC converter. This occurrence dramatically reduces the phase delay of proposed SEPIC converters shown in the open-loop transfer function in Fig. 3.10 in the previous section. Finally, LPF still used for NSHC compensation in the second loop.

Furthermore, this chapter presents the proposed SEPIC-BDI results when integrating a small film capacitor at the DC input side. It clears the aforementioned high switching harmonics. The voltage and the current range of the proposed SEPIC-BDI is also investigated. Finally, this chapter presents the proposed mathematical losses model to check the proposed inverter's efficiency profile at different conditions.

4.2 Proposed Control Technique (Control-I)

Fig. 4.1 illustrates the bode-diagram of the open-loop transfer function at a high duty cycle $d_x = 0.8$. There is a resonance between the inductors and the capacitors of the proposed SEPIC-BDI. This resonance should be damped by resistances of the passive elements and parasitic resistances of the proposed SEPIC converter. However, the dynamic response makes some subharmonics that increase THD. The previous proposed PID compensator did not completely solve this issue. To solve this issue, a Type-II compensator is proposed in this chapter. Type-II compensator replaces the derivative part of the PID compensator with a pole. This pole has -20DB/decade which decrease the gain of SEPIC converter before resonance frequency. Therefore, this pole damps effectively the previously mentioned resonance.



Figure 4.1: Open Loop transfer function of the bode blot of isolated SEPIC-BDI.

This compensator is realized as follows [116, 152-154]:

$$G_c(s) = G_{co} \frac{(1 + \frac{s}{w_{21}})}{(1 + \frac{s}{w_{p1}})(1 + \frac{s}{w_{p2}})}$$
(4.1)

The gain of Type-II, G_{co} increases the zero-frequency gain of the open-loop transfer function of the SEPIC, and improves the steady-state error. Zero of PID, w_{z1} boosts the phase delay of the SEPIC and enhance the overall inverter stability by expanding the system bandwidth. The first pole w_{p1} that put at low frequency flattens the DC gain and increases the control accuracy. The second pole frequency w_{p2} is high, located between resonance frequencies and switching frequency, to damp the switching harmonics. It eliminates the resonance effects from the control path. As shown in Fig. 4.2, the compensator outputs (M, M_q) , indicate the differential inverter's static gain in the rotating frame. Then, the static gain is transferred to the ABC frame using inverse Clark transformation, representing the first component of each SEPIC's modulation index.



Figure 4.2: Proposed control diagram (Control-I) with NSHC compensation circuit of isolated SEPIC-BDI.

In the second loop, the proposed detection circuit of NSHC is based on three LPFs. The output sensors of the grid current have delivered to the proposed LPFs. These LPFs accurately mitigate all low-order harmonics and obtain fundamental components of grid currents. Then, the output of LPFs is subtracted from the actual grid currents to detect the second-order NSHC, as shown in Fig. 4.2. The dynamic transfer function of the LPF has formulated as:

$$G_s(s) = \frac{1}{(1 + \frac{s}{w_l})^3}$$
(4.2)

This transfer function of LPFs has -60DB per decade response due to the located three poles. In addition to pole of Type-II, there are four poles in the system and, all damp the switching harmonics and the resonance produced between capacitors and inductors of

SEPIC converter. The Clark transformation achieves the operation between ABC frame and d-q frame at the angular frequency (-2ω) . Then, the cascaded integrator attenuates the errors of the NSHC loop by modulating the duty cycle of each SEPIC converter. Here, an integrator is used instead of conventional PI to simplify the control function. Finally, the dynamic behavior of integrators in the second loop is slow to prevent any distortion on the first loop.

The closed-loop transfer function of SEPIC-based differential inverter that uses the proposed control is then expressed as follows:

$$T(s) = G_{vd}(s) * G_{q}(s) * G_{c2}(s) * d_{x}(s)$$
(4.3)

Here $G_{c2}(s)$ is the total transfer function of both loops. $G_g(s)$ Is the transfer function of grid-side inductance. $d_x(s)$ represents the transfer function of static linearization function based on equation (3). At frequency domain, it is expressed as:

$$d_{\chi}(s) \cong \frac{m(s) + m(s)K}{m(s) + m(s)K + 1}$$

$$\tag{4.4}$$

Where, m(s) is the small signal component of static gain of differential inverter. According to the proposed control, the bandwidth of the full-system transfer function of the SEPIC converters is 590Hz. The DC gain and the phase margin are 52DB and 26.2, respectively, as illustrated in Fig. 4.3.



Figure 4.3: Bode diagram of closed-loop of modified grid-current control (Control-I).

4.3 System Results of Proposed Control Technique (Control-I)

Simulated-software and experimental outcomes have been conducted in this section to validate the feasibility of the proposed control technique and the theoretical assumption. To consider the mathematical analysis of NSHC, two case studies have been provided as:

- Case 1: This case discusses the effect of NSHC on the SEPIC converters, DC source, and grid. The control task includes Type-II compensators in the first loop. The second loop is disconnected.
- Case 2: This case shows the features of the proposed NSHC strategy using the proposed LPFs. The control task includes the Type-II compensator in the first loop as well as LPFs and integrator in the second loop.

Both cases have performed using PSIM 11.1.3 software and MWPE3 C6713A Expert III DSP board which utilized to implement Control-I.

Fig. 4.4 and 4.5, give both simulation and experimental results of the proposed isolated SEPIC-BDI at case 1. The proposed Type-II compensator has achieved grid-current control and injected active power. However, the grid current in both have distorted with NSHC even at applying static linearization developed by proposed MCMS scheme. These findings of both results match the theoretical assumptions as well as the literature work of differential inverters.

This NSHC component is reflected in SEPIC converters and DC voltage source, as shown in the input currents and the DC current. Again, these results are matched with our assumptions derived at proposed mathematical model at Ch. 3. Moreover, the reflection of NSHC on the DC source current generates a third-order component that also complies with the proposed mathematical model.

It is worth noticing that the NSHC distortion, in experimental results, is worse than simulation results. The additional parasitic of SEPIC converters, such as DC resistances of the PCB boards and connecting wires, are the source of this difference. This difference also introduced at the FFT harmonic spectrum of the grid-current (simulation and experimental results), as provided at Fig. 4.6.



Figure 4.4: Simulation results of proposed SEPIC-BDI at case 1.



Figure 4.5: Experimental results of proposed SEPIC-BDI at case 1.



b- Experimental results Figure 4.6: FFT Harmonic Spectrum of grid-current at case 1.

It shows that the percentage of second-order NSHC is about 25.81% and 40.6% for simulations and experiments. For the THD, the proposed inverter has 23.45% and 35.61% from simulations and experiments. Practically, Table 4.1 illustrates the percentage of other low-order harmonics. It's shown that the proposed static linearization strategy eliminates the other components, and there is a reflection of NSHC on input current of the SEPIC converter. This harmonic violates the IEC-61000 grid standards [157] and hinders the differential-based inverter's practical applications, as shown in Fig. 4.7. Finally, all previous results have shown the serious deformation of NSHC on the proposed differential inverter and the ultimate need for additional loop for NSHC mitigation.

Fig. 4.8 and 4.9 depict the simulation and experimental waveforms of case 2 that consider the proposed NSHC circuit. In both, the proposed control technique injects sinusoidal grid-current at unity power factor. Compared with case 1, the proposed control technique successfully eliminates the NSHC from the grid current. Moreover, it reduces the distortion of the three input inductor currents of SEPIC converters and their mismatch. It eliminates the third-order component from the DC source due to the successful elimination of the NSHC of grid-current.

Low-order harmonics from Simulation				
Current/ Harmonic order	2	3	4	5
Grid current <i>i_{sw}</i>	25.81	0.204	0.605	0.235
Input Current iinw	27.32	10.55	0.902	0.312
Low-order harmonics from Experiments				
Current	2	3	4	5
Grid current <i>i_{sw}</i>	40.6	1.127	3.4	0.923
Input Current <i>i</i> _{inw}	33.64	17.54	1.54	1.93

TABLE 4.1: Low-order harmonic components percentages of proposed inverter at case 1.



Figure 4.7: IEC-61000 standards compared with FFT Spectrum of experimental grid-current at case 1.

Fig. 4.10 demonstrates the FFT harmonic spectrum of the injected grid current at case 2 for simulation and experimental grid-currents. The percentage of second-order NSHC reduced from 25.81% to 0.278% in the simulations. Also, it reduced from 40.6% to 1.6143% in the experiments. It worth mentioning that the percentage of NSHC at proposed Control-I is less than the previously proposed PID control, which has 1.93%. It is backed to the damping effect of the proposed Type-II compensator. The successful mitigation of NSHC forces the third-order harmonies at SEPIC converters' input to reduce from 17.54% to 2.13% in the experimental results. The simulations results also have the same improvements, as illustrated in Table 4.2.



Figure 4.8: Simulation results of proposed SEPIC-BDI at case 2.



Figure 4.9: Experimental results of proposed SEPIC-BDI at case 2.



b- Experimental results

Figure 4.10: FFT Harmonic Spectrum of grid-current at case 2.

Low-order harmonics from Simulation				
Current/ Harmonic order	2	3	4	5
Grid current i_{sw}	0.278	0.575	0.06	0.169
Input Current <i>i</i> _{inw}	35.05	1.08	0.175	0.191
Low-order harmonics from Experiments				
Current	2	3	4	5
Grid current i_{sw}	1.614	1.572	2.385	3.278
Input Current <i>i</i> _{inw}	33.64	2.13	1.03	1.77

Table 4.2: Low-order harmonic components percentages of proposed inverter at case 2.

The maximum regulation limit of the harmonic components in case 2 is reduced from 150% to 70% as shown in Fig. 4.11. It confirms that the SEPIC-BDI using proposed control meets the standard limits of IEC-61000. Finally, THD is reduced from 35.61% to 4.087% and satisfy the recent grid codes (<5%).



Figure 4.11: IEC-61000 standards compared with FFT Spectrum of experimental grid-current at case 2.

4.4 Proposed Control Technique (Control-II)

It worth noticing that the control loops of Control-I are symmetrical as follows:

- ✓ In the first loop, the errors e_d , e_q between fundamental values of actual and reference currents $(i_d, i_q \text{ and } i_{dref}, i_{qref})$ is the input of Type-II compensator.
- ✓ The compensator outputs (M and M_q) are then transferred to ABC frame using inverse Clark transformation, representing the first component of the modulation index of each SEPIC module.

High-order compensators, such as PID and Type-II, provide good performance, as shown in the previous section and Ch. 3. Also, nonlinear control techniques perform good performance for SEPIC converter at many applications, especially differential inverter [88]. However, it still needs excessive computational burden and mighty DSP boards and increases the cost of the proposed SEPIC-DBI. Therefore, using a low computational burden decreases the required DSP board's capability and facilitates an inexpensive one. We are motivated by this; decreasing the order of differential inverter compensator effectively simplifies the control programming code and reduces the calculation time. Low order compensators like conventional PI can be used instead of Type-II compensator by noticing the following important points

Ref [155] states that the parasitic resistances of the SEPIC converter can move the complex RHP zeros to the left half-plane and reduce the phase delay of the SEPIC converter. The phase-delay of the open-loop SEPIC converter is 630, as shown in Fig. 4.1.

✓ LPFs and cascaded integrators in the second loop produces the second component of the static gain. The sum of both represents the final static gain of each SEPIC converter. In Contrast, the transfer function of LPFs has -60DB per decade response due to the located three poles, as presented in (4.2). These poles can be used to damp the switching harmonics of resonance frequencies.

This modified control, employed in Control-II, shown in Fig. 4.12, has the following advantages:

- ✓ The order of compensator, in the first loop, decreases from Type-II to PI and gives the same performance.
- ✓ The modified control-II decreases the execution burden of the controller by using lower execution time.
- \checkmark The modified control-II enhances the THD especially at low static gain M.



Figure 4.12: Proposed control diagram (Control-II) with NSHC compensation circuit of isolated SEPIC-BDI.

4.5 System Results of Proposed Control Technique (Control-II)

As discussed before, the high frequency switching harmonics can be mitigated by using small film capacitors. We add input capacitor to solve the high frequency switching harmonics shown in the previous results. Capacitance of the proposed SEPIC-BDI is 120uF with four parallel connected film capacitors (C4AEHBW5300A3JJ). Moreover, Table 4.3 illustrates the control variables which utilized in the proposed control. Fig. 4.13 illustrates the proposed SEPIC-BDI's experimental validations based on three SEPIC

converters at Vdc = 120V. The input DC current is continuous with a minimal switching ripple, which suits PV modules characteristics and enhances the utilized MPPT control tracking efficiency. As the proposed differential inverter is derived by a variable duty cycle, the input currents and output currents of utilized SEPIC converters change periodically with grid-frequency. The grid injected current is synchronized with grid voltage and introduce pure active power with unity power factor operation.

	Parameters	Value
PI	K_p	0.004
	K _i	$1.34 * 10^{-6}$
LPF	Gain	0.5
	Cut off frequency	120 Hz
Integrator	Gain	1.67
	Frequency	$1 * 10^{-4}$

 Table 4.3: Control parameters for the proposed Control-II.

Fig. 4.14 shows the three SEPIC converters' duty cycles and static gain of the differential inverter at Vdc = 120V and P = 1.6kW. The duty cycle is adjusted to compare with (-1:1) saw-tooth signal with a carrier frequency of 50kHz to generate switch pulses. It is worth noticing that the duty cycles are symmetrical. The static gain is constant and larger than one indicating the boosted voltage gain. Its value is greater than the ideal value due to the voltage drop of the SEPIC converter parasitic.



Figure 4.13: Experimental results of proposed differential inverter.

Fig. 4.15 illustrates the d-q reference and actual values. The reactive power equal to zero, and the steady-state error for both components is small that validates the stability of the proposed control system.



Figure 4.14: Duty cycle and Static gain M of proposed differential inverter.



Figure 4.15: Actual and reference d-q values of proposed differential inverter.

Fig. 4.16 shows the FFT for the experimented grid-current of the SEPIC-BDI at full power 1.6kW. The proposed control technique enhances the waveforms by reducing the NSHC to diminutive values. The percentage of NSHC is 0.624%.



Figure 4.16: FFT analysis of the differential inverter's injected grid-current at full power.
Fig. 4.17 shows the THD values of the output grid-current for this power injected range. At full power, the THD is 3.86%. The THD is lower than 5% in the most different input DC voltage and different load conditions with maximum value 4.99% at 200W.



Figure 4.17: THD analysis of differential inverter's injected grid-current for internal power range.

4.6 Loss Analysis of the Proposed Inverter

The proposed SEPIC-BDI has three identical SEPIC converters with equal power (onethird of total inverter power). Although there is a 120-degree phase shift between them, they have similar duty cycle modulation over one complete cycle with the same peak, average, and valley value. Therefore, power loss distribution is also the same. This section shows the loss distribution of one SEPIC converter. It has split among the following components.

4.6.1 MOSFET Devices

The main switch S_{mx} and body-diode of the synchronous switch S_{rx} process the inverter power in the first half of line-frequency cycle $T_L/2$ or $0 - \pi$ and produce conduction and switching loss. On the other hand, the synchronous switch S_{rx} and the body diode of the main switch S_{mx} process the power in the second half $T_L/2$ or $\pi - 2\pi$. To ease the power loss calculation, the conduction and switching loss are calculated depend on the switching-frequency T_s . Then the power losses are averaged over the line-frequency T_L . Therefore, the conduction loss over one switching cycle T_s of the MOSFET switch is obtained from equation (4.5) and can be given as [158]:

$$P_{cond}(t) = R_{dson} d_x(t) [I_D^2 + \frac{\Delta i_D^2}{12}]$$
(4.5)

Where, R_{dson} is the switch on-resistance, $d_x(t)$ is the duty cycle at this instant. I_D is average drain current over one cycle, and, Δi_D is the ripple component of switch current. Also, the switching loss can be obtained as [59]:

$$P_{swon}(t) = P_{swon} + P_{swof} + P_{coss}$$

$$P_{swon}(t) = \frac{1}{6} t_{rise} F_{sw} V_{ds} \left[I_D - \frac{\Delta i_D}{2} \right]$$

$$P_{swof}(t) = \frac{1}{6} t_{fall} F_{sw} V_{ds} \left[I_D + \frac{\Delta i_D}{2} \right]$$

$$P_{coss}(t) = 0.5 C_{oss} V_{ds}^{2}$$

$$(4.6)$$

In (4.6), t_{rise} , t_{fall} , C_{oss} , are the rise-time, fall-time and drain-source capacitance of the switch, respectively. All previous parameters were found in the datasheet of the switch. Moreover, the conduction loss of the body diode can be obtained as:

$$P_{cond}(t) = d(t)I_F V_F \tag{4.7}$$

The switching loss of body diode can be obtained as:

$$P_{sw}(t) = P_{swon} + P_{swof} + P_{Qrr}$$

$$P_{swon}(t) = t_{dead} F_{sw} V_F \left[I_F - \frac{\Delta i_F}{2} \right]$$

$$P_{swof}(t) = t_{dead} F_{sw} V_F \left[I_F + \frac{\Delta i_F}{2} \right]$$

$$P_{coss}(t) = 0.5 \ V_{ds} F_{sw} Q_{rr}$$

$$(4.8)$$

Table 4.4: Calculated average voltages and currents of the SEPIC converter.

Component	I _d or	ΔI_d or ΔI_F	d(t)	Conduction
Component	I_F			period
Main Switch	ni _{sx}	$d_x(t)V_{dc}$	$d_x(t)$	0-π
S_{mx}	1-d(t)	$(L_x + L_{mx})F_{sv}$		
Synchronous	ni _{sx}	$d_x(t)V_{dc}$	1	$\pi - 2\pi$
Switch	1-d(t)	$(L_x + L_{mx})F_{sv}$	$-d_x(t)$	
S_{rx}				
Body diode of Main	ni _{sx}	$d_x(t)V_{dc}$	1	$\pi - 2\pi$
switch	1-d(t)	$(L_x + L_{mx})F_{sv}$	$-d_x(t)$	
Body diode of	ni _{sx}	$d_x(t)V_{dc}$	$d_x(t)$	0-π
Synchronous switch	1-d(t)	$(L_x + L_{mx})F_{sv}$		

In (4.7) and (4.8), I_F is average forward current over T_s, V_F is the forward-voltage of the body-diode, Δi_F is the ripple component of diode current, t_{dead} is the dead time between main and synchronous switch of the SEPIC converter, and Q_{rr} is the reverse recovery charge of the diode. Table 4.4 states the previously calculated parameters of switches and their body diodes and their connected operating period.

4.6.2 Input Inductor

The inductor losses are generated from magnetic core losses and winding conduction losses. The winding conduction loss is a function of DC resistance of inductor wire R_{dcr} and AC resistance R_{acr} that indicting the proximity effect of inductor wire at switching-frequency. This loss can be obtained as [159]:

$$P_{cond} = I_{lx}^2 R_{dcr} + \frac{1}{12} \Delta i_{lx}^2 R_{acr}$$
(4.9)

Generally, R_{dcr} and R_{acr} of inductor can be determined using inductor test at switching frequency.

The magnetic core loss of inductor P_{core} depends on the maximum ac flux density B_m and the used switching frequency F_{sw} , and the volume of the core and usually obtained by Steinmetz equation to be [116]:

$$P_{core} = k f^{\alpha} B^{\beta} A_c l_m \tag{4.10}$$

Where, A_c is the cross-sectional area of the core, l_m is the core mean magnetic path length. α, β are constants and obtained from the datasheet of the utilized core.

4.6.3 HFT

The losses of the HFT transformers are separated into core losses and conduction losses. Conduction losses introduced from the primary and secondary resistances of the HFT transformer wires R_{pri} , R_{sec} [158]. They are calculated as

$$P_{cond} = \left[\frac{V_{pri}d_x(t)}{(L_{mx} + L_{pri})F_{sw}}\right]^2 R_{pri} + \left[\frac{V_{sec}d_x(t)}{(L_{mx} + L_{sec})F_{sw}}\right]^2 R_{sec}$$
(4.11)

Where, V_{pri} , V_{sec} are the primary and secondary voltages of HFT, L_{pri} , L_{sec} are the primary and secondary winding inductances. Moreover, the core losses can be obtained using equation (4.10).

4.6.4 Coupling and Output Capacitors

From (4.12) and (4.13), the power losses of both capacitors are obtained as:

$$P_{closs} = \frac{1}{12} \left[\frac{n i_{sx} d_x(t)}{c_x F_{sw}} \right]^2 R_{esr}$$

$$\tag{4.12}$$

$$P_{closs} = \frac{1}{12} \left[\frac{i_{sx} d_x(t)}{c_{ox} F_{sw}} \right]^2 R_{esr}$$

$$\tag{4.13}$$

Where, R_{esr} is the ESR resistance of the capacitor. Finally, the losses of other components such as control circuit, snubber circuit of HFL transformers and connecting wires can be obtained from [59, 158-162].

4.7 Efficiency Profile of the Proposed Inverter

The power loss distribution for the proposed inverter topology has been carried out based on the theoretical calculation derived in previous section. The semiconductor devices, inductors, HFT transformer, and capacitors are modeled using their datasheets. Table 4.5 gives the power loss of all components for one SEPIC converter together with the calculated efficiency with an output power of 533W. Then the power of proposed inverter equal to 1.6kW using one SEPIC converter per phase.

Moreover, the variation of efficiency against injected output power with loss calculation using mathematical, simulation and experimental measurements are shown in Fig. 4.18. The maximum experimental efficiency of the SEPIC-BDI is 89.744% at the output power of 700W. As the output power increases, the efficiency decreases to 86.4% at full rating of the proposed inverter 1.6kW. Finally, the developed mathematical analysis of powerloss with simulation and experimental measurements have the following important notices:

- ✓ The experimental efficiency is the lowest due to parasitic components of SEPIC-BDI.
- ✓ These parasitic is hard to calculate it or even measures it. However, the theoretical calculation is still beneficial because it gives us an in-depth investigation of the power-loss distribution and behavior of proposed inverter.
- ✓ Finally, the mismatch between simulation, experimental and theoretical calculations is small and acceptable. And the slope of theoretical calculation is matched with experimental measurements.

			-		
component	P _{cond}	P_{Sw}	total		
Main Switch S_{mx}	6.351	3.666	10.017		
Body diode of main Switch	2.310 1.335		3.645		
S_{mx}					
Synchronous Switch S_{rx}	1.558	1.429	2.987		
Body diode of Synchronous	4.739	1.884	6.623		
Switch S_{rx}					
component	P _{cond}	P _{core}	total		
Input Inductor L_x	10.250	8.000	18.250		
HFL transformer	11.235	8.184	19.437		
Coupling capacitor C_x	0.007				
output capacitor C_{ox}	0.002				
Snubber circuit		2.2	2.2		
Grid inductance	3.07				
Control and gate drives	0.883				
Total power loss and output	67.121 533		533		
Power W					
Efficiency		87.406%			

 Table 4.5: Calculated Power loss distribution of one SEPIC converter at Proposed Inverter topology.

 \checkmark

This small drop is due to the accuracy of loss modeling derived at previous section.



Figure 4.18: Efficiency of the SEPIC-BDI at different output power and Vdc = 100V.

Chapter 5: Modular Differential Inverter: Analysis and Experimentation

5.1 Introduction

The recent increase in PV modules at PV module architecture is normal to expand the system's power generation, especially at grid-connecting PV residential applications. Therefore, many inverter topologies have been widely investigated to process more power through power scalability without any additional complexity for the existing PV architecture. Regarding this, only modular inverter topologies are capable of doing this function efficiently. Moreover, the micro-inverter's most important feature is the power flexibility between MII and String inverter, as discussed in many literary works.

The design and control technique of the proposed SEPIC-BDI and its modular technique at Modular Differential Inverter (MDI) for high power application is presented in this chapter. The operation of the MDI has been described mathematically, considering inputparallel output-parallel SEPIC modules in each phase. A design example based on a threephase MDI is then carried out in PSIM simulation software. It considers a $\pm 20\%$ mismatch between the parallel modules of each phase to investigate system performance in the case of unsymmetrical modules. Finally, a comparison study of the proposed inverter with similar topologies are also presented.

5.2 Proposed Modular Differential Inverter MDI

Generally, a grid-connected MDI is composed of n DC-DC modules (It's called modules in this chapter to differentiate between MDI and traditional differential inverter) per phase, as illustrated in Fig. 5.1. These modules are connected in an Input-Parallel Output-Parallel (IPOP) configuration. For a three-phase grid, the phases are connected differentially. More modules (n > 1) are required when the processing power at each phase is larger than the rated power of the utilized modules. In other words, the MDI processes high power using 3 * n small DC-DC modules.

Fundamental operation of a MDI depends on driving DC-DC modules with a variable duty cycle like the normal circuit of differential inverter discussed in previous chapters (3 and 4). This duty cycle produces an AC output voltage with a common DC offset. It is

worth noting that this offset will appear at the common point, C, as shown in Fig. 5.1. Unlike conventional DC-AC circuits, this DC offset comes from the unipolar operation of DC-DC modules and was cancelled out by the three-phase differential connection at the grid side. The derived DC-DC modules of the same phase have a 0° phase shift, whereas the DC-DC modules of different phases have a 120° phase shift.

According to Fig 1, the phase voltages and phase currents of a balanced three-phase grid is given as follows:

$$\begin{bmatrix} e_{su}(t) \\ e_{sv}(t) \\ e_{sw}(t) \end{bmatrix} = e_m \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin(\omega t + \frac{2\pi}{3}) \end{bmatrix}$$
(5.1)

$$\begin{bmatrix} i_{su}(t) \\ i_{sv}(t) \\ i_{sw}(t) \end{bmatrix} = i_m \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin\left(\omega t + \frac{2\pi}{3}\right) \end{bmatrix}$$
(5.2)



Figure 5.1: Main Circuit of proposed single-stage three-phase MDI.

Where e_m presents the maximum phase voltage, i_m presents the maximum phase current, and ω indicates the line-frequency. The simplest version of a MDI has one DC-DC module per phase for processing the power. The power of each module changes

instantaneously with line-frequency. However, all phases process the same average power, and all modules have the same peak and valley power, and work in a symmetrical operation. The only difference comes from the introduced phase shift of 1200.

To obtain a voltage step-up, any bidirectional boost or buck-boost DC-DC module can be utilized. The voltage gain and current gain of a DC-DC boost converter is obtained as:

$$\frac{v_{mx}}{v_{DC}} = \frac{i_{mx}}{i_{omx}} = \frac{1}{1 - d_x}$$
(5.3)

The voltage and current gain of the DC-DC buck-boost converter is:

$$\frac{v_{mx}}{v_{DC}} = \frac{i_{mx}}{i_{omx}} = \frac{d_x}{1 - d_x}$$
(5.4)

As demonstrated in Fig. 5.1, v_{mx} represents the instantaneous output voltage of the associated DC-DC module, x, in Phase m. Here, x indicates the number of DC-DC modules in each phase (x = 1,2,3,...n), while m indicates the associated phase (m = u, v, and w). For example, v_{u1} presents the instantaneous output voltage of the first DC-DC module at phase u. The same scenario can be applied for input and output currents, e.g., i_{u1}, i_{ou1} represent the input and output currents of the first DC-DC module at phase u, respectively. V_{DC} is the DC input voltage and d_x is the duty cycle of this module.

For MDI, the total input current of one phase can be expressed as follows:

$$i_{inm} = \sum_{x=1}^{n} i_{mx} \tag{5.5}$$

For all phases, the total input current can be expressed as follows:

$$I_{DC} = i_{inu} + i_{inv} + i_{inw} \tag{5.6}$$

On the other hand, the output voltages of the modules of one phase are equal and can be expressed as follows:

$$v_{m1} = v_{m2} = \dots \cdot v_{mn} = v_{om} \tag{5.7}$$

For all phases, it worth mentioning that the RMS value of the output voltages are similar due to IPOP configuration. The phase current of grid can then be expressed as follows:

$$i_{sm} = \sum_{x=1}^{n} i_{omx} \tag{5.8}$$

Many modulation strategies were presented for the control of differential based inverters. It worth noticing that a simple modulation strategy, without additional feedforward sensing and complicated calculations, is required for an MDI. Therefore, the proposed modulation MCMS employed in Chapter 3 is used in this work, as it provides static linearization and suits modularity. Moreover, it can be applied to boost and buck-boost DC-DC modules. From (5.4), the duty the duty cycle can be expressed as follows:

$$d_x = \frac{v_{mx}}{v_{mx} + V_{DC}} \tag{5.9}$$

As presented in Fig. 5.1, the differential inverter produces two voltage components at the output terminal of the DC-DC module: the DC common voltage v_{Do} and an AC voltage. For ideal operation, it is worth noticing that this common voltage is equal at all modules, despite the connected phase. The output voltage of the utilized module can then be simplified to

$$v_{mx}(t) = v_{Do} + e_{sm}(t)$$
(5.10)

Where $e_{sm}(t)$ is the associated phase voltage connected to the module (5.1). Since both components come from the same module, they have the same voltage gain. The output voltage from (5.10) can then be expressed as follows:

$$v_{mx}(t) = MV_{DC} + MV_{DC}K = MV_{DC}(1+K)$$
(5.11)

At (5.11), M represents the accurate static gain of the differential inverter. Term K is the sinusoidal differential constant of the output voltage that introduce the behavior of grid and can then be expressed as follows:

$$K = \begin{bmatrix} \sin(\omega t) \\ \sin\left(\omega t - \frac{2\pi}{3}\right) \\ \sin(\omega t + \frac{2\pi}{3}) \end{bmatrix}$$
(5.12)

By comparing (5.3), (5.4), (5.10) and (5.11), the actual static voltage gain of MDI can be outlined as follows:

$$M = 0.5 \frac{v_{mx}(max)}{v_{DC}}$$
(5.13)

The maximum value of v_{mx} is defined at K = 1. From (5.13), MDI gain, M, presents the half of the maximum voltage gain of the utilized DC-DC module, e.g., if the maximum voltage gain of the utilized DC-DC module is 4, then the maximum possible static gain M is 2. Practically, the voltage gain is reduced by the voltage drop of the parasitic components of the DC-DC modules. Therefore, it is imperative to use DC-DC modules with small parasitic components. By substituting equations (5.10), (5.12) into (5.9), the duty cycle of the utilized DC-DC module can be given as:

$$d_x(t) = \frac{M(1+K)}{M(1+K)+1}$$
(5.14)

It is worth noticing that MDI's duty cycle modulation is typical of the conventional differential inverter's duty cycle modulation (expressed in Ch. 3 and 4). This feature is critical because it facilitates power expansion without any extra complication for the modulation circuit. From (5.14), the duty cycle of the associated DC-DC module x depends on two different variables; the gain of the MDI, M, and the associated connected phase, K. Thus, this modulation supports the modularity of the MDI and each of the utilized DC-DC modules will work independently. Moreover, the control structure does not require major changing if using the MDI at a greater power.

Since the boost DC-DC modules working at MDI, their input current, according to (5.3) and (5.14), can be derived as follows:

$$i_{mx}(t) = 0.5Mi_{omx} + i_{omx}(M+1)K - 0.5Mi_{omx}K_2$$
(5.15)

According to (5.4) and (5.14), the input current of buck-boost DC-DC modules working at MDI is outlined as:

$$i_{mx}(t) = 0.5Mi_{omx} + i_{omx}MK - 0.5Mi_{omx}K_2$$
(5.16)

In both, term, K_2 represents the sinusoidal differential constant of input current for associated DC-DC modules and can be expressed as follows:

$$K_{2} = \begin{bmatrix} \cos[2(\omega t)] \\ \cos[2(\omega t - \frac{2\pi}{3})] \\ \cos[2(\omega t + \frac{2\pi}{3})] \end{bmatrix}$$
(5.17)

The term, K_2 , accurately describes the circulating current between different phases at the differential inverter. This circulating current is evaluated at the double frequency because a reverse current circulates instantaneously from the other two phases in each cycle of grid frequency, as shown in Fig. 5.1. This constant should be considered during the design process of DC-DC modules in order to achieve proper operation of the MDI. Based on (5.15), the input power of the boost DC-DC modules used at the MDI is obtained as follows:

$$p_{inmx}(t) = 0.5MV_{DC}i_{omx} + V_{DC}i_{omx}(M+1)K - 0.5MV_{DC}i_{omx}K_2$$
(5.18)

Using the same process for buck-boost based DC-DC modules, according to (5.16), the input power is expressed as follows:

 $p_{inmx}(t) = 0.5MV_{DC}i_{omx} + V_{DC}i_{omx}MK - 0.5MV_{DC}i_{omx}K_2$ (5.19)

5.3 Performance of proposed SEPIC modules at MDI

Since the MDI converts power between a DC source and the grid, its parallel DC-DC modules should maintain the objectives of micro-inverters which discussed in Ch. 2. DC-DC modules having a two-switch structure are preferred for achieving low component counts and a high efficiency and can work within the maximum possible duty cycle range of $0.0 < d_x < 0.9$. From (5.3), (5.4), and (5.13), Fig. 5.2 shows the voltage gain and possible maximum static gain, M of the MDI, utilizing boost and buck-boost modules, respectively. It worth mentioning that both of them have an approximate static gain.

In perspective of voltage and current stress, according (5.11) and (5.14), and assuming $V_{Dc} = 100V$, 200V line-line voltage and 1.6kW. Fig. 5.3 illustrates duty cycle, output voltage, switches' voltage and current stress of boost and buck-boost modules. Here, the static gain M and equivalent output resistance R_{mx} are assumed 2.0 and100 Ω , respectively. S_{mx} , S_{rx} are the complementary switches of these modules. Notice that the voltage stress is equal for both types. However, the current stress of both switches is lower in buck-boost modules.

The continuous input current is an important factor for renewable energy DC sources, such as PVs. This will add two important features. Firstly, the input inductor of these modules prevents issues related to the short circuit current.



Figure 5.2: Voltage gain of boost and buck-boost modules and MDI static gain at possible duty cycle range.



Figure 5.3: Duty cycle, output voltage, voltage and current stress of boost and buck-boost DC-DC modules at one cycle of line-frequency.

Secondly, it eliminates the large electrolyte input capacitor integrated between the DC source and the MDI, required to decrease the switching ripples [19, 30, 33, 36,87 and 124].

From this point of view, boost converters and other buck-boost converters, such as Cuk and SEPIC, are preferred due to their continuous input current. From (5.15) and (5.16), input current stress on the DC source for both modules is illustrated in Fig. 5.4 at $i_{omx} = 8A$. Obviously, buck-boost converters have a lower input current stress.



Figure 5.4: Input current of boost and buck-boost modules at differential inverter on one cycle of line-frequency.

According to the grid, voltage isolation is important for CMV and EMI. This feature is accomplished by using HFT-based DC-DC converters such as SEPIC, Cuk, and flyback. Moreover, HFT based topologies are better for low DC current injection. Finally, the THD of the grid-current should be within acceptable limits, which depends on the MDI control rather than the utilized DC-DC modules.

Based on the previous discussion, SEPIC and Cuk converters are preferred. Furthermore, the SEPIC converter is better in this MDI due to its low component count and the low energy storage of its components [87, 91].

5.4 Design Example of MDI based on Proposed SEPIC Modules

The design process of the proposed SEPIC module is shown in Fig. 5.5. In the case of more than one module per phase, the process should be repeated in the following pattern $n = 1, 2, \dots$. The boundary conditions are defined by the following points:

- (Maximum n) is restricted by the efficiency of the utilized SEPIC. Efficiency is generally low at low rated power due to high power losses; in particular, switching losses [59, 158-162]. The DC-DC modules in the 100 V range are most efficient when operated at several hundred watts (300 W ≤ Power ≤ 700 W). Similar converters are implemented in modular systems such as data centers, UPSs, and smart grids [163-171].
- (Minimum n ≥ 1), is restricted by the current standing of the utilized SEPIC. Generally, the voltage gain of any DC-DC module is inversely proportional to the rated current. At high current, the voltage drop will increase and will not provide a sufficient voltage gain for MDI [116, 129-130].

Table 5.1 demonstrates the circuit variables of the presented MDI. The power of, one SEPIC module, per phase can be expressed as follows:

$$P_{module} = \frac{P_{MDI}}{3*n} = \frac{1600}{3*1} = 533.3w \tag{5.20}$$

Form grid specifications, the phase voltage and phase current of each module are $e_m = 163.29v$ and $i_m = 6.532A$, respectively. From (5.13), the static gain of MDI can be obtained as follows:

$$M = \frac{163.29 + 163.29}{2 \times 100} = 1.63 \tag{5.21}$$

The voltage gain of SEPIC Module should be larger than 2 * M as shown in Fig 2. Therefore, the design is conducted based on gain M = 2 to consider the voltage drop of parasitic components.



Figure 5.5: Flowchart of the design steps for the proposed SEPIC module utilized

in MDI.

Table 5.1: Circuit parameters of MDI.

Rated Power P _{MDI}	4.8kW
Input voltage V _{DC}	100v
Line voltage V_{ll}	200v
Line-frequency F_{line}	60Hz
Switching-frequency F_{sw}	50kHz

5.4.1 Simulation Results

To prove the validity of the MDI proposed control technique, a circuit comprising of nine SEPIC modules is simulated in PSIM software. System parameters of the MDI used in the simulation are recorded in Table 5.1. Three modules of the SEPIC are connected in IPOP in each phase in order to process a triple high power 4.8 kW. A mismatch in components is introduced to investigate the performance of the MDI.

A $\pm 20\%$ parameter mismatch is considered in the DC resistance of the input inductor

for the first and third SEPIC modules in phase u. Fig. 5.6 illustrates the grid voltage, duty cycle (d_u, d_v, d_w) , injected grid current (i_{su}, i_{sv}, i_{sw}) , reference current, and actual current. The grid current is synchronized with grid voltage. The current is free from NSHC and the DC offset, even at the existing mismatch between phase u and the other phases (phases v and w).



Figure 5.6: Grid voltage, duty cycles, injected grid-current, reference and actual current of MDI.

Fig. 5.7 demonstrates the duty cycles, output voltages and input currents of the SEPIC modules connected at phase u. The existed mismatch does not affect the output voltage due to the grid voltage effect.

The mismatch slightly affected input currents as shown in Fig 5.8. The maximum effect of mismatch results at high duty cycle. The average input-currents are 26.4A, 26.7A and 27.0A for I_{u1} , I_{u2} , I_{u3} respectively. Therefore, the parameter mismatch of 20% resulted in a 0.3A difference of the input current. Otherwise, Table 5.2 shows the maximum mismatch of the input-current and its percentage at ($d_u = 0.8$) due to $\pm 20\%$ disturbance in the input inductance HFT magnetizing inductance, Coupling capacitor, and output capacitors. In all cases, the three-phase output currents were synchronized with the grid voltage and hence generated low THD.



Figure 5.7: Duty cycles, output voltages, input current of SEPIC modules in



phase u with $\pm 20\%$.

Figure 5.8: Zoomed View of Fig. 5.7 at high duty cycle of SEPIC modules connected in phase u.

5.4.2 Experimental Results

A simplified MDI based on three SEPIC modules is implemented to validate the proposed modules. In this system, only one SEPIC module is used to process the power in each phase. This because the following two reasons

- There is no effect of mismatched modules in the same phase. This finding is validated by simulation results
- The mismatch of modules at different phases can be checked in the simplified MDI based on three SEPIC modules
- The system setup of MDI needs many SEPIC modules and the laboratory experimentation is limited.

Motivated by this, the experimental results, shown in this chapter, concentrated on

mismatch SEPIC modules and voltage and current stress.

Component	Module mismatch element	Average	Maximum
		Input	Mismatch &Percentage
		current	
Input	$L_x(u1) = 144 \mu \text{H}$	27.8	1A
Inductor	$L_x(u2) = 180 \ \mu \text{H}$	26.8	3.73%
	$L_x(u3) = 216 \mu H$	25.9	
Magnetizing	$L_{mx}(u1) = 400 \ \mu H$	29.7	0.1A
inductance	$L_{mx}(u2) = 500 \ \mu H$	29.8	0.34%
	$L_{mx}(u3) = 600 \ \mu H$	29.9	
Coupling	$C_x(u1)=11.2 \ \mu F$	29.8	0.04A
Capacitor	$C_x(u2)=14 \ \mu F$	29.76	0.13%
	$C_x(u3)=16.8 \ \mu F$	29.75	
output	$C_{ox}(u1)=11.2 \ \mu F$	29.8	0.04A
Capacitor	$C_{ox}(u2)=14 \ \mu F$	29.76	0.13%
	$C_{ox}(u3)=16.8 \ \mu F$	29.75	

Table 5.2: The maximum mismatch in PSIM at different mismatch cases at $d_u =$

0.8.

As previously discussed, the voltage and current switches' ratings should be designed for the worst-case conditions that existed at the peak voltages and currents. This is very important for MDI and SEPIC-BDI presented in Ch. 4 and this chapter. The voltage and current of the primary and synchronous switch are measured experimentally to confirm this effect, as illustrated in Fig. 5.9-a. It is worth noticing that both switches operate simultaneously and achieve two power operations. The inverter manages forward and circulating power efficaciously. The maximum peak voltage of switches has existed at the main switch and equal 700V. This value includes the drain-source voltage, voltage ripple component, and voltage spikes of leakage inductance HFT. On the other hand, the maximum current has obtained at the synchronous switch's body diode and equal 40A. Both voltage and current peaks are limited successfully under intentioned limits because of the snubber circuit, which damps it under the utilized switches' safe operation zones. Fig. 5.9-b shows the switches operation at forward power by zooming Fig 5.9-a (Zoom A). The instant duty cycle at this voltage is 0.75. It has validated that the main switch and body-diode process this forward power. Fig 5.9-c shows the switches operation at reverse power by zooming Fig 5.9-a (Zoom B). The instant duty cycle at this voltage is 0.32. The reverse power is diverted by the synchronous switch and body-diode of the main switch. Finally, Fig 5.9-b, c illustrates the effective operation of both switches at switching frequency, which proves the excellent performance and the designated gate drives of the proposed SEPIC-BDI.



a. Switching performance at grid frequency range.



b. Zoom A: switching frequency at forward power.

```
c. Zoom B: switching frequency at reverse power.
```

Figure 5.9: Experimental voltage and current stress of utilized MOSFET switches at proposed differential inverter.

Fig. 5.10 displays the voltage and current of input inductor and HFT at grid-frequency and switching-frequency. Both magnetic elements operate simultaneously (forward and reverse power). They manage forward and reverse power by charging and discharging states. The maximum peak voltage and the maximum peak current of the input inductor are 666V and 28.7A. The maximum peak voltage and the maximum peak current of HFT are 567V and 29A.



a. Magnetic elements performance at grid frequency range.





b. Zoom A: switching frequency at forward power (Mode-A).



Figure 5.10: Experimental voltage and current stress of utilized inductor and HFT at proposed differential inverter.

It is worth noticing that these worst-case of voltages and current conditions that include the average values, ripple components, and switching voltage and current spikes are under the saturation limits set during these magnetic elements' design process, as presented in Ch. 3.

Fig 5.10-b exhibits the voltage and current waveforms at forward power by zooming Fig 5.10-a (Zoom A). The instant duty cycle at this voltage is 0.75. It has validated that the

input inductor's voltage equal to the DC input voltage and the primary voltage of HFT at charging and discharging states, respectively. Fig 5.10-c displays the voltage and current at reverse power by zooming Fig 5.10-a (Zoom B). The instant duty cycle at this voltage is 0.355. The inductor current in both charging and discharging states is negative, which indicate the reverse power processing.

Finally, Fig 5.10-b and c show the tight operation of input inductor and HFT at switching frequency. Again, these findings prove the accuracy of the design criteria proposed in Ch.3 and 4 for the proposed differential inverter.

5.5 Discussion, Comparison and Evaluation

In Table 5.3, comparative evaluation between the proposed inverter and state of the art topologies have been carried out in terms of topology, voltage gain, voltage step-up/down ability, different components count and power rating. Also, it includes the feasibility of the topologies for modularity, galvanic isolation, and power efficiency. The proposed inverter requires low number of switches (six switches like the traditional VSI) which is less than the proposed topologies in [67-70], and [78-84]. Furthermore, it does not need any extra diodes like topology proposed in [73-77]. Another exciting benefit of the proposed topology is the galvanic isolation using HFT without any additional switches. Finally, it has an excellent potency to be modular and scale the power for over extension. The switched-capacitors inverters still need improvements in terms of switches, where topologies in [81] and [82] have 8 and 27 switches with many gate drive circuits. It does not have a modular configuration, and the HFT integration is still not investigated.

For Z source inverter topologies, the buck-boosting ability is excellent. However, the Z network's size is still substantial (they have inductors in mHrange) [72-76]. Again, it needs additional components to implement galvanic isolation and modularity. For multi-level inverters, the efficiency is generally high. However, the number of switches is very high and needs large filters due to low switching-frequency operation.

In contrast, Table 5.3 gives a quantitative comparison between the proposed inverter and similar differential inverter topologies. The proposed topology based on SEPIC converters has only three inductors, which is lower than the Cuk topologies (six inductors). Furthermore, the utilized inductors have small values that improve the inverter size.

Although the measured efficiency of the proposed topology is close to proposed one in [89], it has many advantages to enhance it in the future.

	Topology	Voltage	Powe	Capacitors	Switch	Modula	HFL	Applicati	Measure
T		gain	r	%	es	r	&	on	d
Topology			(KW)	inductors	Diodes		CMV		efficienc
									у
Ref [81]	Switched	Buck-	NA	2	8	No	No	Three-	97.4
	Capacitor	Boost		0	2		yes	phase	
Ref [82]	Switched	Buck-	NA	2	27	No	No	Three-	NA
	Capacitor	Boost		0	0		NA	phase	
Ref [76]	Z source	Buck-	0.45	2	6	No	No	Three-	NA
		Boost		2	1		NA	phase	
Ref[172]	Z source	Buck-	1	2	6	No	No	Three-	97.1
		Boost		2	1			phase	
Ref[173]	Multi-level	Buck	1.1	5	8	Yes	No	Single-	96.3
				0	6		yes	phase	
Ref[85]	Differential	Boost	1.4	3	6	Yes	No	Three-	NA
				3	0		NA	phase	
Ref[101]	Differential	Buck-	0.5	3	4	Yes	No	Single-	97
		Boost		4	2		NA	phase	
Ref[86]	Differential	CUK	2.5	6	6	Yes	No	Three-	NA
				6	0		NA	phase	
Ref[87]	Differential	CUK	2.5	6	6	Yes	yes	Three-	NA
				6	0		yes	phase	
Ref[99]	Differential	Boost	0.5	2	4	NO	No	Single-	NA
				2	0		NA	phase	
Ref[89]	Differential	CUK	0.5	6	6	Yes	yes	Three-	91%
				6	0		yes	phase	
Ref[88]	Differential	Buck& 2	0.4	3	4	Yes	No	Three-	NA
		SEPIC		3	0		NA	phase	
proposed	Differential	SEPIC	1.6	6	6	Yes	yes	Three-	89.744
				3	0		yes	phase	

 Table 5.3: Comparison between proposed inverter and similar topologies.

Table 5.4 summaries the literature work, which eliminates NSHC at differential inverters. The proposed control design provides grid-current control and excludes NSHC by using two control loops. NSHC can be decreased partially by using DCM operation, as reported at [89]. However, this will add higher-order harmonic components (i.e., third, fourth, and extra) that disturb THD, and need extra loops. The DCM Cuk-based differential inverter at [89] needs more loops to decrease THD to acceptable values, as outlined at [94]. Unlike [87], the proposed control eliminates the third-order components using the second loop from the perspective of input current distortion. It is imperative to exclude the third loop, which adds complexity and a computational burden for utilized DSP board. Moreover, it provides accurate mathematical modeling for its association with NSHC, mentioned in Ch. 3. Also, this behavior has been verified using simulations and experiments.

Work		This work Ref [87] Ref [89]		Ref [89]	Ref [94]
DC-DC Converter		SEPIC Cuk Cuk		Cuk	Cuk
Modulation Scheme		CMS	CMS CMS DMS		DMS
Control	SLB	Yes	Yes	Yes	Yes
Feature	NSHC	Yes	Yes	No	Yes
5	No	2	3	1	5
	loops				
THD		4.087	1.2	5.1	4
F _{sw} [kHz]		50	25	125	100

Table 5.4: Comparison between proposed inverter and similar control techniques.

Table 5.5 demonstrates the performance of proposed control techniques in this thesis. It worth noticing that all controllers provide high performance. Moreover, the proposed improved controls (Control-I and Control-II) have low execution time that improves the computational burden of the DSP board. The NSHC is less than 1.6143% that enhances

the THD (less than 5%). Finally, these findings achieve the standards of recent grid codes.

Control/	Proposed	Proposed	Proposed
Parameters	PID control	Control-I	Control-II
First Loop	PID	Type-II	PI
Second Loop	LPFs	LPFs	LPFs
NSHC (%)	1.93	1.6143	0.613
Execution time (us)	21.35	19.95	18.04
THD (%)	5.7	4.087	3.86
Static gain M	1.63	1.63	1.36

 Table 5.5: Comparison between the three proposed control techniques.

Chapter 6: Conclusions

6.1 Contribution of This Work

The work outlined in this thesis has made some important contribution in the area of power-electronics in DC-AC micro-inverter circuits, especially, for single-stage three-phase inverter topologies utilized in advanced PV module architectures. The major contribution of this thesis can be highlighted as follows:

- Detailed classification of the published three-phase inverter topologies is presented. This classification is depend on the requirements and standards of the advanced PV module architectures.
- State the advantages of three-phase micro-inverter against single-phase micro-inverter topologies in the recent PV module architectures.
- The interesting topologies for three-phase micro-inverters such as Two-Stage VSI inverters, Single-Stage Z Source Inverters, Single-Stage Switched-Capacitor Inverters and Single-Stage Differential Inverters have also reviewed.
- A three-phase single-stage SEPIC-Based Differential Inverter SEPIC-BDI is proposed. It has three small HFT based SEPIC converters which enhance the power density and power scalability and modularity.
- The modified modulation (MCMS) which provides static linearization for the duty cycle of the SEPIC-BDI is presented with details. It produces a grid output current free form low-order harmonics without any extra harmonic compensation loops.
- A novel mathematical model is proposed for NSHC of the differential inverters. It investigates the NSHC source and its associations with the differential inverters characteristics, such as variable duty cycle, unipolar operation of DC-DC converters, and circulating currents.
- The proposed SEPIC-BDI, design and implementation are also described. It includes the parameters selection, its design criteria, small-signal representation, and the proposed control technique.
- The proposed inverter's simulation and experimental outcomes have been discussed to show the high performance and effectiveness of the proposed inverter.
- A novel two modified control techniques (Control-I, Control-II) has been illustrated to enhance the inverter performance and decrease THD under acceptable

limits.

- In Control-I, a Type-II compensator, which enhances SEPIC converters' stability, is used instead of the PID controller. The proposed Type-II compensator effectively damp the resonance of the SEPIC converter. In the second loop, the NSHC detection circuit added the proposed LPFs is also utilized.
- In Control-II, the proposed Control-I is redesigned to use SEPIC converters' parasitic components to move the right half-plane zeros from the right half-plane to the left half-plane. This critical feature at SEPIC converters reduces the proposed Type-II compensator's order to a conventional PI controller, especially at low static gain *M*. The new results show tremendous improvements with a small computational burden.
- The proposed SEPIC-BDI results when integrating a small film capacitor at the input DC side is implemented. It clears the high switching harmonics. The power range of the proposed SEPIC-BDI is also investigated to suit PV modules characteristics.
- The proposed inverter efficiency profile is presented with power loss distinction. In this direction, an accurate mathematical losses model is provided to check the proposed inverter's efficiency profile at different conditions.
- The design and control technique of the proposed SEPIC-BDI and its modular technique at Modular Differential Inverter (MDI) for high power application is presented.
- The operation of the MDI has been described mathematically, considering inputparallel output-parallel SEPIC modules in each phase.
- A design example based on a three-phase MDI is then carried out in PSIM simulation software. It considers a ±20% mismatch between the parallel modules of each phase to investigate system performance in the case of unsymmetrical modules. Finally, a comparison study of the proposed inverter with similar topologies are also presented.

6.2 Suggestions for Future Work

In continuation of this proposal, the following points are some suggestions for future work

- The proposed mathematical model reveals that the stress of switches devices and magnetic elements can be enhanced by using novel modulation schemes. This scheme is an improved version of MCMS and DMS modulation scheme.
- To validate the proposed inverter at PV module architecture, implementing MPPT control with inherent partial shading condition circuits should be developed. Also, the voltage and current range of the SEPIC-BDI ought to be tested.
- Check the capability of the proposed SEPIC-BDI for bidirectional power flow should be investigated. Also, the capability to control reactive power control.
- Future work can also be recommended for HFT design and making accurate mathematical analysis for variable duty cycle which include line-frequency variation.

References

Chapter 1:

- US Energy Information Administration (EIA). "International Energy Outlook 2013 with Projection to 2040". Available online: http://www.eia.gov/forecasts/ieo/ (accessed on 1 July 2013).
- [2] M. Nadeau, "World Energy Scenarios World Energy Council. 2016" Available online: www.worldenergy.org (Accessed on 1 January 2016).
- [3] M. Hasanuzzaman, U. S. Zubir, N. I. Ilham, H. S. Che, "Global electricity demand, generation, grid system, and renewable energy polices: A review" Wiley Inter-disciple Rev. Energy Environ. 2016, 6, E222.
- [4] V. Moutinho, M. Robaina, "Is the share of renewable energy sources determining the CO2 kWh and income relation in electricity generation" Renew. Sustain. Energy Rev. 2016, 65, 902–914.
- [5] European Photovoltaic Industry Association (EPIA), "Global Market Outlook for Photovoltaic until 2016; European" Photovoltaic Industry association (EPIA): Brussels, Belgium, May 2012.
- [6] "Economic and Energy Outlook of Japan through FY2019: Decelerating growth and growing disquiets" the Institute of Energy Economics of Japan, July 2018.
- [7] K. Ardian, "Solar Technologies Market Report; National Renewable Energy Laboratory (NREL)" Golden, CO, USA, 2010.
- [8] W. Hoffmann, "PV on the way from a few lead markets to a world market," 2006 IEEE 4th World Conference on Photovoltaic Energy Conference, Waikoloa, HI, 2006, pp. 2454-2456, doi: 10.1109/WCPEC.2006.279706.
- [9] D. Feldman, G. Barbose, R. Margolis, R. Wiser, N. Darghouth, A. Goodrich1, "Photovoltaic (PV) Pricing Trends: Historical, Recent, and Near-Term Projections" Report, Sun Shot; US Department of Energy: Washington, DC, USA, 2012.
- [10] F. Petrakopoulou, "On the economics of stand-alone renewable hybrid power plants in remote regions" Energy Conversion and Management. 2016, 118, pp. 63–74.
- [11] J. P. Torreglosa, P. Garcia-Trivino, L. M. Fernandez-Ramirez, F. Jurado, "Control based on technoeconomic optimization of renewable hybrid energy system for stand-alone applications" Expert Syst. Appl. 2016, 51, 59–75.
- [12] I.R. Cristóbal-Monreal, R. Dufo-López, "Optimization of photovoltaic-diesel-battery stand-alone systems minimizing system weight" Energy Conversion and Management. 2016, 119, pp. 279–288.
- [13] N. Sujitha, S. Krithiga, "RES based EV battery charging system: A review" Renew. Sustain. Energy Rev. 2017, 75, pp. 978–988.
- [14] T. Ackermann, G. Andersson, L. Söder, "Distributed generation: A definition" Electric. Power Syst. Res. 2001, 57, 195–204.
- [15] Z. Zhao, L. Yin, X. Sun, L. Yuan and T. Lu, "Recent development of technology and market of grid-

connected PV system in China," 2010 International Conference on Electrical Machines and Systems, Incheon, 2010, pp. 1-6.

- [16] S. Kouro, J. I. Leon, D. Vinnikov and L. G. Franquelo, "Grid-Connected Photovoltaic Systems: An Overview of Recent Research and Emerging PV Converter Technology," in *IEEE Industrial Electronics Magazine*, vol. 9, no. 1, pp. 47-61, March 2015, doi: 10.1109/MIE.2014.2376976.
- [17] R. Shah, N. Mithulananthan, R.C. Bansal, V.K. Ramachandaramurthy, "A review of key power system stability challenges for large-scale PV integration. Renew" Sustain. Energy Rev. 2015, 41, pp. 1423–1436.
- [18] K. Alluhaybi, I. Batarseh and H. Hu, "Comprehensive Review and Comparison of Single-Phase Grid-Tied Photovoltaic Micro-inverters," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 8, no. 2, pp. 1310-1329, June 2020, doi: 10.1109/JESTPE.2019.2900413.
- [19] A. Shawky, "Development of Power Management Systems for Advanced Photovoltaic Architectures" Master's Thesis, Aswan University, Aswan, Egypt, May 2014. Chapter 1.
- [20] Q. Li and P. Wolfs, "A Review of the Single Phase Photovoltaic Module Integrated Converter Topologies With Three Different DC Link Configurations," in *IEEE Transactions on Power Electronics*, vol. 23, no. 3, pp. 1320-1333, May 2008, doi: 10.1109/TPEL.2008.920883.
- [21] J. Feng, B. Zeng, D. Zhao, G. Wu, Z. Liu and J. Zhang, "Evaluating Demand Response Impacts on Capacity Credit of Renewable Distributed Generation in Smart Distribution Systems," in IEEE Access, vol. 6, pp. 14307-14317, 2018, doi: 10.1109/ACCESS.2017.2745198.
- [22] A. Shawky, M. Orabi, J. A. Qahouq and M. Youssef, "Development and comparative evaluation of power management systems for advanced photovoltaic architectures," 2014 IEEE 36th International Telecommunications Energy Conference (INTELEC), Vancouver, BC, 2014, pp. 1-8, doi: 10.1109/INTLEC.2014.6972141.
- [23] P. Mazumdar, P. N. Enjeti and R. S. Balog, "Analysis and Design of Smart PV Modules," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 2, no. 3, pp. 451-459, Sept. 2014, doi: 10.1109/JESTPE.2013.2294640.
- [24] J. M. Bing, "Advances in commercial applications of photovoltaic technology in North America: 2009 update," *IEEE PES General Meeting*, Providence, RI, 2010, pp. 1-5, doi: 10.1109/PES.2010.5589631.
- [25] S. Macalpine, C. Deline, "Modeling Micro inverters and DC Power Optimizers in PV Watts", Technical Report NREL/TP-5J00-63463; National Renewable Energy Laboratory (NREL): Golden, CO, USA, 2015.
- [26] A. Reinders, P. Verlinden, van, W. Sark, A. Freundlich, "Photovoltaic Solar Energy: From Fundamentals to Applications. In Inverters, Power Optimizers and Micro inverters" 1st ed.; John Wiley & Sons, Ltd.: Hoboken, NJ, USA, 2017; Chapter 11.
- [27] H. Zhou, J. Zhao and Y. Han, "PV Balancers: Concept, Architectures, and Realization," in *IEEE Transactions on Power Electronics*, vol. 30, no. 7, pp. 3479-3487, July 2015, doi:

10.1109/TPEL.2014.2343615.

- [28] A. Shawky, F. Hilmy, M. Orabi, E. M. Ahmed and J. A. A. Qahouq, "A Single Cell Integrated Photovoltaic Converter based on Buck-Boost Topology with RCC MPPT Control," *Intelec 2013; 35th International Telecommunications Energy Conference, SMART POWER AND EFFICIENCY*, Hamburg, Germany, 2013, pp. 1-6.
- [29] M. Orabi, F. Hilmy, A. Shawky, J. A. Abu Qahouq, E. S. Hasaneen, E. Gomaa. "On-chip integrated power management MPPT controller utilizing cell-level architecture for PV solar system" Sol. Energy. 2015; 117: 10–28.
- [30] Femia, N. and Petrone, G. and Spagnuolo, G. and Vitelli, M., "Power Electronics and Control Techniques for Maximum Energy Harvesting in Photovoltaic Systems", CRC Press, 1st edition December 2012.
- [31] SMA Solar Inverters "SB5000TL21 SMA 5 kW Single Phase TL Solar Inverter". Available online: https://www.sma-america.com/products/solarinverters.html.
- [32] Enphase Energy. Enphase Energy Commercial PV Using the Enphase M-Series Micro inverter System; Design Guide: California CA, USA, 2015.
- [33] H. Hu, S. Harb, N. Kutkut, I. Batarseh and Z. J. Shen, "A Review of Power Decoupling Techniques for Microinverters With Three Different Decoupling Capacitor Locations in PV Systems," in *IEEE Transactions on Power Electronics*, vol. 28, no. 6, pp. 2711-2726, June 2013, doi: 10.1109/TPEL.2012.2221482.
- [34] Y. Sun, Y. Liu, M. Su, W. Xiong and J. Yang, "Review of Active Power Decoupling Topologies in Single-Phase Systems," in *IEEE Transactions on Power Electronics*, vol. 31, no. 7, pp. 4778-4794, July 2016, doi: 10.1109/TPEL.2015.2477882.
- [35] R. Hasan, S. Mekhilef, M. S. Seyedmahmoudian, and B. Horan, "Grid connected isolated PV micro inverters: A review," Renew. Sustain. Energy Rev., vol. 67, pp. 1065–1080, Jan. 2017.
- [36] M.Y. Ali Khan, H. Liu, Z. Yang, X. Yuan, "A Comprehensive Review on Grid Connected Photovoltaic Inverters, Their Modulation Techniques, and Control Strategies" *Energies* 2020, 13, 4185.
- [37] A. R. Gautam, D. M. Fulwani, R. R. Makineni, A. K. Rathore and D. Singh, "Control Strategies and Power Decoupling Topologies to Mitigate 2ω-Ripple in Single-Phase Inverters: A Review and Open Challenges," in *IEEE Access*, vol. 8, pp. 147533-147559, 2020, doi: 10.1109/ACCESS.2020.3015315.
- [38] C. R. Bush and B. Wang, "A single-phase current source solar inverter with reduced-size DC link," 2009 IEEE Energy Conversion Congress and Exposition, San Jose, CA, 2009, pp. 54-59, doi: 10.1109/ECCE.2009.5316285.
- [39] U. Somani, "Design Optimization of LLC Topology and Phase Skipping Control of Three-phase Inverter for PV Applications" Master's Thesis, University of Central Florida, Orlando, FL, USA, September 2013.

- [40] Q. Zhang *et al.*, "A Center Point Iteration MPPT Method With Application on the Frequency-Modulated LLC Microinverter," in *IEEE Transactions on Power Electronics*, vol. 29, no. 3, pp. 1262-1274, March 2014, doi: 10.1109/TPEL.2013.2262806.
- [41] A. Shawky, F. Helmy, M. Orabi, J. A. Qahouq and Z. Dang, "On-chip integrated cell-level power management architecture with MPPT for PV solar system," 2014 IEEE Applied Power Electronics Conference and Exposition - APEC 2014, Fort Worth, TX, 2014, pp. 572-579, doi: 10.1109/APEC.2014.6803366.
- [42] A. Shawky, H. Radwan, M. Orabi and M. Z. Youssef, "A novel platform for an accurate modeling and precise control of photovoltaic modules with maximum operating efficiency," 2015 IEEE Applied Power Electronics Conference and Exposition (APEC), Charlotte, NC, 2015, pp. 205-212, doi: 10.1109/APEC.2015.7104353.
- [43] T. Esram and P. L. Chapman, "Comparison of Photovoltaic Array Maximum Power Point Tracking Techniques," in *IEEE Transactions on Energy Conversion*, vol. 22, no. 2, pp. 439-449, June 2007, doi: 10.1109/TEC.2006.874230.
- [44] S. H. Hanzaei, S. A. Gorji and M. Ektesabi, "A Scheme-Based Review of MPPT Techniques with Respect to Input Variables Including Solar Irradiance and PV Arrays' Temperature," in *IEEE Access*, vol. 8, pp. 182229-182239, 2020, doi: 10.1109/ACCESS.2020.3028580.
- [45] D. Meneses, F. Blaabjerg, Ó. García and J. A. Cobos, "Review and Comparison of Step-Up Transformerless Topologies for Photovoltaic AC-Module Application," in *IEEE Transactions on Power Electronics*, vol. 28, no. 6, pp. 2649-2663, June 2013, doi: 10.1109/TPEL.2012.2227820.
- [46] T. T. Nguyen, N. Nguyen and N. R. Prasad, "Eliminated common-mode voltage pulsewidth modulation to reduce output current ripple for multilevel inverters," in *IEEE Transactions on Power Electronics*, vol. 31, no. 8, pp. 5952-5966, Aug. 2016, doi: 10.1109/TPEL.2015.2489560.
- [47] A. Videt, M. Messaoudi, N. Idir, H. Boulharts and H. Vang, "PWM Strategy for the Cancellation of Common-Mode Voltage Generated by Three-Phase Back-to-Back Inverters," in *IEEE Transactions* on Power Electronics, vol. 32, no. 4, pp. 2675-2686, April 2017, doi: 10.1109/TPEL.2016.2573831.
- [48] Q. A. Le and D. Lee, "Elimination of Common-Mode Voltages Based on Modified SVPWM in Five-Level ANPC Inverters," in *IEEE Transactions on Power Electronics*, vol. 34, no. 1, pp. 173-183, Jan. 2019, doi: 10.1109/TPEL.2018.2825230.
- [49] K. Basu, A. C. Umarikar, K. K. Mohapatra and N. Mohan, "High-frequency transformer-link threelevel inverter drive with common-mode voltage elimination," 2008 IEEE Power Electronics Specialists Conference, Rhodes, 2008, pp. 4413-4418, doi: 10.1109/PESC.2008.4592657.
- [50] Y. Ji, D. Jung, J. Kim, J. Kim, T. Lee and C. Won, "A Real Maximum Power Point Tracking Method for Mismatching Compensation in PV Array Under Partially Shaded Conditions," in *IEEE Transactions on Power Electronics*, vol. 26, no. 4, pp. 1001-1009, April 2011, doi: 10.1109/TPEL.2010.2089537.

- [51] M. Sidrach-de-Cardona, J. Carretero, "Analysis of the current total harmonic distortion for different single-phase inverters for grid- connected PV-systems" Solar Energy Materials and Solar Cells, 2005, 87, 529-540.
- [52] M. Ahmed *et al.*, "General Mathematical Solution for Selective Harmonic Elimination," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 8, no. 4, pp. 4440-4456, Dec. 2020, doi: 10.1109/JESTPE.2019.2932933.
- [53] Y. Hu *et al.*, "Ultrahigh Step-up DC–DC Converter for Distributed Generation by Three Degrees of Freedom (3DoF) Approach," in *IEEE Transactions on Power Electronics*, vol. 31, no. 7, pp. 4930-4941, July 2016, doi: 10.1109/TPEL.2015.2487821.
- [54] H. Liu, H. Hu, H. Wu, Y. Xing and I. Batarseh, "Overview of High-Step-Up Coupled-Inductor Boost Converters," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 4, no. 2, pp. 689-704, June 2016, doi: 10.1109/JESTPE.2016.2532930.
- [55] A. Shawky, A. Shier, M. Orabi, J. A. A. Qahouq and M. Youssef, "A High Efficiency Single-stage Current Source Inverter for Photovoltaic Applications," *Intelec 2013; 35th International Telecommunications Energy Conference, SMART POWER AND EFFICIENCY*, Hamburg, Germany, 2013, pp. 1-6.
- [56] H. D. Paulino, P. J. M. Menegáz and D. S. L. Simonetti, "A review of the main inverter topologies applied on the integration of renewable energy resources to the grid," XI Brazilian Power Electronics Conference, Praiamar, 2011, pp. 963-969, doi: 10.1109/COBEP.2011.6085334.
- [57] X. She, A. Q. Huang, Ó. Lucía and B. Ozpineci, "Review of Silicon Carbide Power Devices and Their Applications," in *IEEE Transactions on Industrial Electronics*, vol. 64, no. 10, pp. 8193-8205, Oct. 2017, doi: 10.1109/TIE.2017.2652401.
- [58] A. Merkert, T. Krone and A. Mertens, "Characterization and Scalable Modeling of Power Semiconductors for Optimized Design of Traction Inverters with Si- and SiC-Devices," in *IEEE Transactions on Power Electronics*, vol. 29, no. 5, pp. 2238-2245, May 2014, doi: 10.1109/TPEL.2013.2294682.
- [59] M. Orabi and A. Shawky, "Proposed Switching Losses Model for Integrated Point-of-Load Synchronous Buck Converters," in *IEEE Transactions on Power Electronics*, vol. 30, no. 9, pp. 5136-5150, Sept. 2015, doi: 10.1109/TPEL.2014.2363760.
- [60] D. Zhang *et al.*, "A comparison of soft and hard-switching losses in three phase micro-inverters," 2011 IEEE Energy Conversion Congress and Exposition, Phoenix, AZ, 2011, pp. 1076-1082, doi: 10.1109/ECCE.2011.6063893.
- [61] Han Ju Cha and P. N. Enjeti, "A new soft switching direct converter for residential fuel cell power system," *Conference Record of the 2004 IEEE Industry Applications Conference, 2004. 39th IAS Annual Meeting.*, Seattle, WA, USA, 2004, pp. 1172-1177 vol.2, doi: 10.1109/IAS.2004.1348561.

Chapter 2:

- [62] S. Shuvo, E. Hossain and Z. R. Khan, "Fixed Point Implementation of Grid Tied Inverter in Digital Signal Processing Controller," in *IEEE Access*, vol. 8, pp. 89215-89227, 2020, doi: 10.1109/ACCESS.2020.2993985.
- [63] T. Sen, N. Pragallapati, V. Agarwal and R. Kumar, "Global maximum power point tracking of PV arrays under partial shading conditions using a modified particle velocity-based PSO technique," in *IET Renewable Power Generation*, vol. 12, no. 5, pp. 555-564, 9 4 2018, doi: 10.1049/ietrpg.2016.0838.
- [64] A. Hassan, A. Shawky, M. Orabi, J. A. A. Qahouq and M. Z. Youssef, "Optimization of integrated power conditioning PV parameters," *Intelec 2012*, Scottsdale, AZ, 2012, pp. 1-7, doi: 10.1109/INTLEC.2012.6374550.
- [65] H. Li, D. Yang, W. Su, J. Lü and X. Yu, "An Overall Distribution Particle Swarm Optimization MPPT Algorithm for Photovoltaic System Under Partial Shading," in *IEEE Transactions on Industrial Electronics*, vol. 66, no. 1, pp. 265-275, Jan. 2019, doi: 10.1109/TIE.2018.2829668.
- [66] B. Sahan, S. V. Araújo, C. Nöding and P. Zacharias, "Comparative Evaluation of Three-Phase Current Source Inverters for Grid Interfacing of Distributed and Renewable Energy Systems," in *IEEE Transactions on Power Electronics*, vol. 26, no. 8, pp. 2304-2318, Aug. 2011, doi: 10.1109/TPEL.2010.2096827.
- [67] C. Jain and B. Singh, "A Three-Phase Grid Tied SPV System With Adaptive DC Link Voltage for CPI Voltage Variations," in *IEEE Transactions on Sustainable Energy*, vol. 7, no. 1, pp. 337-344, Jan. 2016, doi: 10.1109/TSTE.2015.2496297.
- [68] L. Chen, A. Amirahmadi, Q. Zhang, N. Kutkut and I. Batarseh, "Design and Implementation of Three-Phase Two-Stage Grid-Connected Module Integrated Converter," in *IEEE Transactions on Power Electronics*, vol. 29, no. 8, pp. 3881-3892, Aug. 2014, doi: 10.1109/TPEL.2013.2294933.
- [69] L. Chen, C. Hu, Q. Zhang, K. Zhang and I. Batarseh, "Modeling and Triple-Loop Control of ZVS Grid-Connected DC/AC Converters for Three-Phase Balanced Microinverter Application," in IEEE Transactions on Power Electronics, vol. 30, no. 4, pp. 2010-2023, April 2015, doi: 10.1109/TPEL.2014.2329278.
- [70] R. Huang and S. K. Mazumder, "A Soft-Switching Scheme for an Isolated DC/DC Converter With Pulsating DC Output for a Three-Phase High-Frequency-Link PWM Converter," in *IEEE Transactions on Power Electronics*, vol. 24, no. 10, pp. 2276-2288, Oct. 2009, doi: 10.1109/TPEL.2009.2022755.
- [71] F. C. Melo, L. S. Garcia, L. C. de Freitas, E. A. A. Coelho, V. J. Farias and L. C. G. de Freitas, "Proposal of a Photovoltaic AC-Module With a Single-Stage Transformerless Grid-Connected Boost Microinverter," in *IEEE Transactions on Industrial Electronics*, vol. 65, no. 3, pp. 2289-2301, March 2018, doi: 10.1109/TIE.2017.2750611.

- [72] Fang Zheng Peng, "Z-source inverter," in *IEEE Transactions on Industry Applications*, vol. 39, no. 2, pp. 504-510, March-April 2003, doi: 10.1109/TIA.2003.808920.
- [73] T. -D. Duong, M. -K. Nguyen, T. -T. Tran, Y. -C. Lim, J. -H. Choi and C. Wang, "Modulation Techniques for a Modified Three-Phase Quasi-Switched Boost Inverter With Common-Mode Voltage Reduction," in *IEEE Access*, vol. 8, pp. 160670-160683, 2020, doi: 10.1109/ACCESS.2020.3020635.
- [74] P. Liu, J. Xu, Y. Yang, H. Wang and F. Blaabjerg, "Impact of Modulation Strategies on the Reliability and Harmonics of Impedance-Source Inverters," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 8, no. 4, pp. 3968-3981, Dec. 2020, doi: 10.1109/JESTPE.2019.2933575.
- [75] B. Barathy, A. Kavitha and T. Viswanathan, "Effective space vector modulation switching sequence for three phase Z source inverters," in *IET Power Electronics*, vol. 7, no. 11, pp. 2695-2703, 11 2014, doi: 10.1049/iet-pel.2013.0903.
- [76] A. Bakeer, M. A. Ismeil and M. Orabi, "A Powerful Finite Control Set-Model Predictive Control Algorithm for Quasi Z-Source Inverter," in *IEEE Transactions on Industrial Informatics*, vol. 12, no. 4, pp. 1371-1379, Aug. 2016, doi: 10.1109/TII.2016.2569527.
- [77] L. Pan, "L-Z-Source Inverter," in *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp. 6534-6543, Dec. 2014, doi: 10.1109/TPEL.2014.2303978.
- [78] W. Qian, F. Z. Peng and H. Cha, "Trans-Z-Source Inverters," in *IEEE Transactions on Power Electronics*, vol. 26, no. 12, pp. 3453-3463, Dec. 2011, doi: 10.1109/TPEL.2011.2122309.
- [79] S. Jiang, D. Cao and F. Z. Peng, "High frequency transformer isolated Z-source inverters," 2011 Twenty-Sixth Annual IEEE Applied Power Electronics Conference and Exposition (APEC), Fort Worth, TX, 2011, pp. 442-449, doi: 10.1109/APEC.2011.5744634.
- [80] S. Bayhan, H. Abu-Rub and R. S. Balog, "Model Predictive Control of Quasi-Z-Source Four-Leg Inverter," in *IEEE Transactions on Industrial Electronics*, vol. 63, no. 7, pp. 4506-4516, July 2016, doi: 10.1109/TIE.2016.2535981.
- [81] T. Tran, M. Nguyen, T. Duong, J. Choi, Y. Lim and F. Zare, "A Switched-Capacitor-Voltage-Doubler Based Boost Inverter for Common-Mode Voltage Reduction," in IEEE Access, vol. 7, pp. 98618-98629, 2019, doi: 10.1109/ACCESS.2019.2930122.
- [82] J. Zeng, W. Lin and J. Liu, "Switched-Capacitor-Based Active-Neutral-Point-Clamped Seven-Level Inverter With Natural Balance and Boost Ability," in IEEE Access, vol. 7, pp. 126889-126896, 2019, doi: 10.1109/ACCESS.2019.2927351.
- [83] C. Daolian, "Single-Stage Three-Phase Voltage Source Inverter with a Cascaded Magnetic Integrated Switching Inductor-Capacitor Network" United States Patent Application 20200106370.
- [84] A. Ahmad, R. K. Singh and A. R. Beig, "Switched-Capacitor Based Modified Extended High Gain Switched Boost Z-Source Inverters," in *IEEE Access*, vol. 7, pp. 179918-179928, 2019, doi: 10.1109/ACCESS.2019.2959136.
- [85] C. Cecati, A. Dell'Aquila and M. Liserre, "A novel three-phase single-stage distributed power

inverter," in *IEEE Transactions on Power Electronics*, vol. 19, no. 5, pp. 1226-1233, Sept. 2004, doi: 10.1109/TPEL.2004.835112.

- [86] A. Darwish, D. Holliday, S. Ahmed, A. M. Massoud and B. W. Williams, "A Single-Stage Three-Phase Inverter Based on Cuk Converters for PV Applications," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 2, no. 4, pp. 797-807, Dec. 2014, doi: 10.1109/JESTPE.2014.2313185.
- [87] A. Darwish, A. M. Massoud, D. Holliday, S. Ahmed and B. W. Williams, "Single-Stage Three-Phase Differential-Mode Buck-Boost Inverters With Continuous Input Current for PV Applications," in *IEEE Transactions on Power Electronics*, vol. 31, no. 12, pp. 8218-8236, Dec. 2016, doi: 10.1109/TPEL.2016.2516255.
- [88] M. S. Diab, A. Elserougi, A. M. Massoud, A. S. Abdel-Khalik and S. Ahmed, "A Four-Switch Three-Phase SEPIC-Based Inverter," in *IEEE Transactions on Power Electronics*, vol. 30, no. 9, pp. 4891-4905, Sept. 2015, doi: 10.1109/TPEL.2014.2363853.
- [89] S. Mehrnami, S. K. Mazumder and H. Soni, "Modulation Scheme for Three-Phase Differential-Mode Ćuk Inverter," in *IEEE Transactions on Power Electronics*, vol. 31, no. 3, pp. 2654-2668, March 2016, doi: 10.1109/TPEL.2015.2442157.
- [90] K. Basu and N. Mohan, "A High-Frequency Link Single-Stage PWM Inverter With Common-Mode Voltage Suppression and Source-Based Commutation of Leakage Energy," in *IEEE Transactions on Power Electronics*, vol. 29, no. 8, pp. 3907-3918, Aug. 2014, doi: 10.1109/TPEL.2013.2280600.
- [91] A. Shawky, M. E. Ahmed and M. Orabi, "Performance analysis of isolated DC-DC converters utilized in Three-phase differential inverter," 2016 Eighteenth International Middle East Power Systems Conference (MEPCON), Cairo, 2016, pp. 821-826, doi: 10.1109/MEPCON.2016.7836989.
- [92] B. Koushki, A. Safaee, P. Jain and A. Bakhshai, "Zero voltage switching differential inverters," 2015 IEEE Applied Power Electronics Conference and Exposition (APEC), Charlotte, NC, 2015, pp. 1905-1910, doi: 10.1109/APEC.2015.7104606.
- [93] B. Koushki, S. A. Khajehoddin, S. M. Saghaian-Nejad, J. Ghaisari, P. Jain and A. Bakhshai, "A voltage reference design for three-phase differential inverters," *IECON 2014 40th Annual Conference of the IEEE Industrial Electronics Society*, Dallas, TX, 2014, pp. 1167-1173, doi: 10.1109/IECON.2014.7048650.
- [94] A. Kulkarni, A. Gupta and S. K. Mazumder, "Resolving Practical Design Issues in a Single-Phase Grid-Connected GaN-FET-Based Differential-Mode Inverter," in *IEEE Transactions on Power Electronics*, vol. 33, no. 5, pp. 3734-3751, May 2018, doi: 10.1109/TPEL.2017.2767572.
- [95] N. Kumar, M. Mohamadi and S. K. Mazumder, "Passive Damping Optimization of the Integrated-Magnetics-Based Differential-Mode Ćuk Rectifier," in *IEEE Transactions on Power Electronics*, vol. 35, no. 10, pp. 10008-10012, Oct. 2020, doi: 10.1109/TPEL.2020.2981918.
- [96] A. Kulkarni and S. K. Mazumder, "Dead time Elimination in a GaN-Based Grid-Connected

Differential-Mode Ćuk Inverter," in *IEEE Transactions on Industrial Electronics*, vol. 66, no. 4, pp. 3296-3302, April 2019, doi: 10.1109/TIE.2018.2836906.

- [97] H. Soni, S. K. Mazumder, A. Gupta, D. Chatterjee and A. Kulkarni, "Control of Isolated Differential-Mode Single- and Three-Phase Cuk Inverters at Module Level," in *IEEE Transactions on Power Electronics*, vol. 33, no. 10, pp. 8872-8886, Oct. 2018, doi: 10.1109/TPEL.2017.2779408.
- [98] P. T. Sivasubramanian, S. K. Mazumder, H. Soni, A. Gupta and N. Kumar, "A DC/DC Modular Current-Source Differential-Mode Inverter," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 4, no. 2, pp. 489-503, June 2016, doi: 10.1109/JESTPE.2015.2506584.
- [99] R. O. Caceres and I. Barbi, "A boost DC-AC converter: analysis, design, and experimentation," in *IEEE Transactions on Power Electronics*, vol. 14, no. 1, pp. 134-141, Jan. 1999, doi: 10.1109/63.737601
- [100] G. L. Piazza and I. Barbi, "New Step-Up/Step-Down DC–AC Converter," in IEEE Transactions on Power Electronics, vol. 29, no. 9, pp. 4512-4520, Sept. 2014, doi: 10.1109/TPEL.2013.2277961.
- [101] A. A. Khan, Y. W. Lu, W. Eberle, L. Wang, U. A. Khan and H. Cha, "Single-Phase Split-Inductor Differential Boost Inverters," in *IEEE Transactions on Power Electronics*, vol. 35, no. 1, pp. 107-120, Jan. 2020, doi: 10.1109/TPEL.2019.2913750.
- [102] S. Huang, F. Tang, Z. Xin, Q. Xiao and P. C. Loh, "Grid-Current Control of a Differential Boost Inverter With Hidden LCL Filters," in *IEEE Transactions on Power Electronics*, vol. 34, no. 1, pp. 889-903, Jan. 2019, doi: 10.1109/TPEL.2018.2817640.
- [103] W. Yao, Y. Xu, Y. Tang, P. C. Loh, X. Zhang and F. Blaabjerg, "Generalized Power Decoupling Control for Single-Phase Differential Inverters With Nonlinear Loads," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 7, no. 2, pp. 1137-1151, June 2019, doi: 10.1109/JESTPE.2018.2844098.
- [104] A. K. Pati and N. C. Sahoo, "A novel power quality enhancement scheme for three-phase differential boost inverter-based grid-connected photovoltaic system with repetitive and feedback linearizing control" international Transactions on Electrical and Energy Systems, Wiley Online Library, 2019.
- [105] M. S. Diab, A. Elserougi, A. S. Abdel-Khalik, A. M. Massoud and S. Ahmed, "Modified modulation scheme for photovoltaic fed grid-connected three-phase boost inverter," *IECON 2013 -39th Annual Conference of the IEEE Industrial Electronics Society*, Vienna, 2013, pp. 1735-1740, doi: 10.1109/IECON.2013.6699394.
- [106] Y. Wang, A. Darwish, D. Holliday and B. W. Williams, "Plug-In Repetitive Control Strategy for High-Order Wide-Output Range Impedance-Source Converters," in *IEEE Transactions on Power Electronics*, vol. 32, no. 8, pp. 6510-6522, Aug. 2017, doi: 10.1109/TPEL.2016.2615689.
- [107] P. Killeen and D. C. Ludois, "Three-Phase Bidirectional-Flyback Differential-Inverter for Synchronous Electrostatic Machines," 2020 IEEE Energy Conversion Congress and Exposition

(ECCE), Detroit, MI, USA, 2020, pp. 2364-2371, doi: 10.1109/ECCE44975.2020.9236257.

- [108] R. T. Pogulaguntla and N. R. Tummuru, "Three Phase Differential Flyback Based Inverter for Photovoltaic Grid Connected Applications," 2020 IEEE International Conference on Power Electronics, Smart Grid and Renewable Energy (PESGRE2020), Cochin, India, 2020, pp. 1-6, doi: 10.1109/PESGRE45664.2020.9070272.
- [109] M. Jang and V. G. Agelidis, "A Boost-Inverter-Based, Battery-Supported, Fuel-Cell Sourced Three-Phase Stand-Alone Power Supply," in *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp. 6472-6480, Dec. 2014, doi: 10.1109/TPEL.2014.2303994.
- [110] B. Eskandari and M. T. Bina, "Support vector regression-based distortion compensator for threephase dc-ac boost-inverters: analysis and experiments," in *IET Power Electronics*, vol. 7, no. 2, pp. 251-258, February 2014, doi: 10.1049/iet-pel.2013.0046.
- [111] M. Saad, A. Shawky and M. Orabi, "Design of integrated POL DC-DC converters based on twostage architectures," 2013 Twenty-Eighth Annual IEEE Applied Power Electronics Conference and Exposition (APEC), Long Beach, CA, 2013, pp. 2098-2105, doi: 10.1109/APEC.2013.6520585.
- [112] M. Saad, Y. Nour, A. Shawky, M. Orabi and E. M. Ahmed, "A Two-Stage Boost-Buck Converter with Intermediate Bus Voltage Control Scheme for Li-Ion Battery Powered Applications," *Intelec* 2013; 35th International Telecommunications Energy Conference, SMART POWER AND EFFICIENCY, Hamburg, Germany, 2013, pp. 1-6.
- [113] M. Khodabandeh, E. Afshari and M. Amirabadi, "A Family of Ćuk, Zeta, and SEPIC Based Soft-Switching DC–DC Converters," in *IEEE Transactions on Power Electronics*, vol. 34, no. 10, pp. 9503-9519, Oct. 2019, doi: 10.1109/TPEL.2019.2891563.
- [114] K. Park, G. Moon and M. Youn, "Non isolated High Step-up Boost Converter Integrated With Sepic Converter," in *IEEE Transactions on Power Electronics*, vol. 25, no. 9, pp. 2266-2275, Sept. 2010, doi: 10.1109/TPEL.2010.2046650.
- [115] J. Yang and H. Do, "Bridgeless SEPIC Converter with a Ripple-Free Input Current," in *IEEE Transactions on Power Electronics*, vol. 28, no. 7, pp. 3388-3394, July 2013, doi: 10.1109/TPEL.2012.2226607.

Chapter 3:

- [116] R. W. Erickson and D. Maksimovic, Fundamentals of Power Electronics, 2nd ed. Norwell, MA: Kluwer, 2001.
- [117] R. Zhao and A. Kwasinski, "Multiple-input single ended primary inductor converter (SEPIC) converter for distributed generation applications," 2009 IEEE Energy Conversion Congress and Exposition, San Jose, CA, 2009, pp. 1847-1854, doi: 10.1109/ECCE.2009.5316295.
- [118] H. Y. Kanaan, K. Al-Haddad and F. Fnaiech, "Switching-function-based modeling and control of a SEPIC power factor correction circuit operating in continuous and discontinuous current modes," 2004 IEEE International Conference on Industrial Technology, 2004. IEEE ICIT '04.
Hammamet, Tunisia, 2004, pp. 431-437 Vol. 1, doi: 10.1109/ICIT.2004.1490329.

- [119] J. N. Hemalatha, S. A. Hariprasad and G. S. Anitha, "Performance evaluation of single ended primary inductance converter for photo voltaic applications," 2014 Annual International Conference on Emerging Research Areas: Magnetics, Machines and Drives (AICERA/iCMMD), Kottayam, 2014, pp. 1-8, doi: 10.1109/AICERA.2014.6908235.
- [120] Y. P. Siwakoti *et al.*, "High-Voltage Gain Quasi-SEPIC DC–DC Converter," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 7, no. 2, pp. 1243-1257, June 2019, doi: 10.1109/JESTPE.2018.2859425.
- [121] S. A. Ansari and J. S. Moghani, "A Novel High Voltage Gain Non coupled Inductor SEPIC Converter," in *IEEE Transactions on Industrial Electronics*, vol. 66, no. 9, pp. 7099-7108, Sept. 2019, doi: 10.1109/TIE.2018.2878127.
- [122] P. K. Maroti, S. Padmanaban, J. B. Holm-Nielsen, M. Sagar Bhaskar, M. Meraj and A. Iqbal, "A New Structure of High Voltage Gain SEPIC Converter for Renewable Energy Applications," in IEEE Access, vol. 7, pp. 89857-89868, 2019, doi: 10.1109/ACCESS.2019.2925564.
- [123] Z. Zhang, J. Lin, Y. Zhou and X. Ren, "Analysis and Decoupling Design of a 30 MHz Resonant SEPIC Converter," in *IEEE Transactions on Power Electronics*, vol. 31, no. 6, pp. 4536-4548, June 2016, doi: 10.1109/TPEL.2015.2472479.
- [124] S. Lee and H. Do, "Zero-Ripple Input-Current High-Step-Up Boost–SEPIC DC–DC Converter with Reduced Switch-Voltage Stress," in *IEEE Transactions on Power Electronics*, vol. 32, no. 8, pp. 6170-6177, Aug. 2017, doi: 10.1109/TPEL.2016.2615303.
- [125] J. J. Jozwik and M. K. Kazimierczuk, "Dual sepic PWM switching-mode DC/DC power converter," in *IEEE Transactions on Industrial Electronics*, vol. 36, no. 1, pp. 64-70, Feb. 1989, doi: 10.1109/41.20346.
- [126] M. R. Banaei and S. G. Sani, "Analysis and Implementation of a New SEPIC-Based Single-Switch Buck–Boost DC–DC Converter With Continuous Input Current," in *IEEE Transactions on Power Electronics*, vol. 33, no. 12, pp. 10317-10325, Dec. 2018, doi: 10.1109/TPEL.2018.2799876.
- [127] C. Shi, A. Khaligh and H. Wang, "Interleaved SEPIC Power Factor Pre regulator Using Coupled Inductors In Discontinuous Conduction Mode With Wide Output Voltage," in *IEEE Transactions on Industry Applications*, vol. 52, no. 4, pp. 3461-3471, July-Aug. 2016, doi: 10.1109/TIA.2016.2553650.
- [128] R. Pandey and B. Singh, "PFC-SEPIC converter-fed half-bridge LLC resonant converter for ebike charging applications," in *IET Electrical Systems in Transportation*, vol. 10, no. 3, pp. 225-233, 9 2020, doi: 10.1049/iet-est.2019.0001.
- [129] M. A. Al-Saffar, E. H. Ismail, A. J. Sabzali and A. A. Fardoun, "An Improved Topology of SEPIC Converter With Reduced Output Voltage Ripple," in *IEEE Transactions on Power Electronics*, vol. 23, no. 5, pp. 2377-2386, Sept. 2008, doi: 10.1109/TPEL.2008.2001916.
- [130] H. Ardi and A. Ajami, "Study on a High Voltage Gain SEPIC-Based DC-DC Converter With

Continuous Input Current for Sustainable Energy Applications," in *IEEE Transactions on Power Electronics*, vol. 33, no. 12, pp. 10403-10409, Dec. 2018, doi: 10.1109/TPEL.2018.2811123.

- [131] S. Gotoh, T. Otake, Y. Fukuda, and J. Togawa, "High Performance MnZn Ferrites for Transformer Core Used in Forward Mode Switching Power Supply" JFE Techn. Rep. No.16 (2011).
- [132] M. Kacki, M. S. Rylko, J. G. Hayes and C. R. Sullivan, "A Practical Method to Define High Frequency Electrical Properties of MnZn Ferrites," 2020 IEEE Applied Power Electronics Conference and Exposition (APEC), New Orleans, LA, USA, 2020, pp. 216-222, doi: 10.1109/APEC39645.2020.9124101.
- [133] N. Hirose, Y. Matsui, and T. Takeshita: "Isolated AC/DC Converter Using Simple PWM Strategy", International Power Electronics Conference, IPEC (2018).
- [134] M. A. Sayed, K. Suzuki, T. Takeshita and W. Kitagawa, "PWM Switching Technique for Three-Phase Bidirectional Grid-Tie DC–AC–AC Converter With High-Frequency Isolation," in *IEEE Transactions on Power Electronics*, vol. 33, no. 1, pp. 845-858, Jan. 2018, doi: 10.1109/TPEL.2017.2668441.
- [135] M. A. Sayed, K. Suzuki, T. Takeshita and W. Kitagawa, "Soft-Switching PWM Technique for Grid-Tie Isolated Bidirectional DC–AC Converter With SiC Device," in *IEEE Transactions on Industry Applications*, vol. 53, no. 6, pp. 5602-5614, Nov.-Dec. 2017, doi: 10.1109/TIA.2017.2731738.
- [136] O. Aldosari, L.A. Garcia Rodriguez, J.C. Balda, and S.K. Mazumder: "Design Trade-Offs for Medium- and High-Frequency Transformers for Isolated Power Converters in Distribution System Applications", 2018 9th IEEE PEDG, Charlotte, NC, USA, pp.1–7 (2018).
- [137] A. Hren and P. Slibar, "Full order dynamic model of SEPIC converter," Proceedings of the IEEE International Symposium on Industrial Electronics, 2005. ISIE 2005. Dubrovnik, Croatia, 2005, pp. 553-558 vol. 2.
- [138] V. Gali and P. B. Amrutha, "Fast dynamic response of SEPIC converter based photovoltaic DC motor drive for water pumping system," 2016 International Conference on Circuit, Power and Computing Technologies (ICCPCT), Nagercoil, 2016, pp. 1-5, doi: 10.1109/ICCPCT.2016.7530298.
- [139] E. Vuthchhay and C. Bunlaksananusorn, "Dynamic Modeling of a Zeta Converter with State-Space Averaging Technique," 2008 5th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology, Krabi, 2008, pp. 969-972, doi: 10.1109/ECTICON.2008.4600593.
- [140] M. Veerachary and M. V. Reddy, "Robust digital controller for fourth order converter," 2010 Joint International Conference on Power Electronics, Drives and Energy Systems & 2010 Power India, New Delhi, 2010, pp. 1-5, doi: 10.1109/PEDES.2010.5712508.
- [141] P. Athalye, D. Maksimovic and R. Erickson, "Averaged switch modeling of active-clamped converters," *IECON'01. 27th Annual Conference of the IEEE Industrial Electronics Society (Cat.*

No.37243), Denver, CO, USA, 2001, pp. 1078-1083 vol.2, doi: 10.1109/IECON.2001.975930.

- [142] A. Cantillo, A. De Nardo, N. Femia and W. Zamboni, "Stability Issues in Peak-Current-Controlled SEPIC," in *IEEE Transactions on Power Electronics*, vol. 26, no. 2, pp. 551-562, Feb. 2011, doi: 10.1109/TPEL.2010.2066288.
- [143] T. Kobaku, R. Jeyasenthil, S. Sahoo and T. Dragicevic, "Experimental verification of robust PID controller under feedforward framework for a Non-minimum phase DC-DC Boost converter," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, doi: 10.1109/JESTPE.2020.2999649.
- [144] A. binti Alias, M. binti Azri, A. bin Jidin and W. binti Abdul Halim, "A modified PID controller of SEPIC converter for excellent dynamic performance," 2016 IEEE International Conference on Power and Energy (PECon), Melaka, 2016, pp. 423-427, doi: 10.1109/PECON.2016.7951599.
- [145] S. Yazdani, M. Ferdowsi and P. Shamsi, "Internal Model based PID Tuning of A Phase-Shift Control In A Single-Phase Bidirectional Dual Active Bridge DC-DC Converter," 2020 IEEE Energy Conversion Congress and Exposition (ECCE), Detroit, MI, USA, 2020, pp. 5966-5970, doi: 10.1109/ECCE44975.2020.9236216.
- [146] T. Kobaku, R. Jeyasenthil, S. Sahoo, R. Ramchand and T. Dragicevic, "Quantitative Feedback Design Based Robust PID Control of Voltage Mode Controlled DC-DC Boost Converter," in *IEEE Transactions on Circuits and Systems II: Express Briefs*, doi: 10.1109/TCSII.2020.2988319.
- [147] B. Saleh, A. Teirelbar and A. Wasfi, "A DC/DC buck-boost converter control using sliding surface mode controller and adaptive PID controller," 2020 22nd European Conference on Power Electronics and Applications (EPE'20 ECCE Europe), Lyon, France, 2020, pp. P.1-P.8, Doi: 10.23919/EPE20ECCEEurope43536.2020.9215853.

Chapter 4

- [148] N. C. Swanepoel, C. G. Richards, A. F. Nnachi and P. J. Ehlers, "A comparative study of maximum power point tracking algorithms for PV arrays," 2020 6th IEEE International Energy Conference (ENERGYCon), Gammarth, Tunis, Tunisia, 2020, pp. 819-823, doi: 10.1109/ENERGYCon48941.2020.9236562.
- [149] J. Ahmad *et al.*, "Performance Analysis and Hardware-In-the-Loop (HIL) Validation of Single Switch High Voltage Gain DC-DC Converters for MPP Tracking in Solar PV System," in *IEEE Access*, doi: 10.1109/ACCESS.2020.3034310.
- [150] N.E. Zakzouk, M.A. Elsaharty, A.K. Abdelsalam, A.A. Helal, "Improved performance low-cost incremental conductance PV MPPT technique" IET Renewable Power Generation 2016, 10(4): 561-574.
- [151] H. Rezk, A. Fathy, "Simulation of global MPPT based on teaching-learning-based optimization technique for partially shaded PV system" Electrical Engineering 2017, 99: 847-859.
- [152] "Demystifying Type II and Type III Compensators Using OpAmp and OTA for DC/DC Converters" Texas instrument application report, July 2014.

- [153] N. Rana, A. Ghosh and S. Banerjee, "Development of an Improved Tristate Buck–Boost Converter With Optimized Type-3 Controller," in IEEE Journal of Emerging and Selected Topics in Power Electronics, vol. 6, no. 1, pp. 400-415, March 2018.
- [154] P. Modesto, R. Cordero and A. D. Estefani, "Design of an Angle Tracking Observer for Resolver Sensors using a Type-II System and Root Locus Technique," 2020 IEEE XXVII International Conference on Electronics, Electrical Engineering and Computing (INTERCON), Lima, Peru, 2020, pp. 1-4, doi: 10.1109/INTERCON50315.2020.9220224.
- [155] V. Vorpérian, "Analysis of the Sepic Converter by Dr. Vatché Vorpérian,"Ridley Engineering Inc, available online: www.switchingpowermagazine.com, 2006.
- [156] V. Vorperian, "Fast analytical techniques for electrical and electronic circuits," Cambridge University Press, 2002.
- [157] Harmonic Current Emissions Guidelines to the standard EN 61000-3-2" available online at http://www.epsma.org/PFCver100406_b.pdf.
- [158] A.V. Kulkarni, W. Chen, A.M. Bazzi, "Implementation of Rapid Prototyping Tools for Power Loss and Cost Minimization of DC-DC Converters" Energies 2016, 9, 509.
- [159] G. Z. Abdelmessih and J. M. Alonso, "Loss analysis for efficiency improvement of the integrated buck-flyback converter for LED driving applications," 2017 IEEE Industry Applications Society Annual Meeting, Cincinnati, OH, 2017, pp. 1-8, doi: 10.1109/IAS.2017.8101801.
- [160] Y. Xiong, S. Sun, H. Jia, P. Shea and Z. John Shen, "New Physical Insights on Power MOSFET Switching Losses," in *IEEE Transactions on Power Electronics*, vol. 24, no. 2, pp. 525-531, Feb. 2009, doi: 10.1109/TPEL.2008.2006567.
- [161] S. K. Roy and K. Basu, "Analytical Model to Study Hard Turn-off Switching Dynamics of SiC mosfet and Schottky Diode Pair," in *IEEE Transactions on Power Electronics*, vol. 36, no. 1, pp. 861-875, Jan. 2021, doi: 10.1109/TPEL.2020.2998873.
- [162] Z. Tong, J. Roig-Guitart, T. Neyer, J. D. Plummer and J. M. Rivas-Davila, "Origins of Softswitching Coss Losses in SiC Power MOSFETs and Diodes for Resonant Converter Applications," in *IEEE Journal of Emerging and Selected Topics in Power Electronics*, doi: 10.1109/JESTPE.2020.3034345.

Chapter 5

- [163] Y. Hayashi and M. Mino: "Series-Parallel Connected 10W/cm3 Dc-dc Converter for Advanced High-Density Converter Design", Proceedings of the International Telecommunications Energy Conference (INTELEC), Amsterdam, Netherland (2011)
- [164] Z. Li, S. Kai, X. Yan, F. Lanlan, and G. Hongjuan: "A modular grid connected photovoltaic generation system based on DC bus", IEEE Trans. Power Electron., Vol.26, No.2, pp.523–531 (2011)
- [165] M. Salato, A. Zolj, D.J. Becker, and B.J. Sonnenberg: "Power System Architectures for 380V DC Distribution in Telecom Datacenters", IEEE Intelec (2012)

- [166] V. Jaikrishna, S. Dash, L. Alex, and R. Sridhar: "Investigation on modular flyback converters using PI and fuzzy logic controllers", International Journal of Ambient Energy, pp.1–9 (2017)
- [167] R. Giri, R. Ayyana, and N. Mohan: "Common duty ratio control of input series connected modular dc-dc converters with active input voltage and load current sharing", in Proc. 18th Annu. IEEE Applied Power Electronics Conf. Expo. (APEC'03), Vol.1, pp.322–326 (2003).
- [168] M. A. Pagliosa, T. B. Lazzarin and I. Barbi, "Modular Two-Switch Flyback Converter and Analysis of Voltage-Balancing Mechanism for Input-Series and Output-Series Connection," in *IEEE Transactions on Power Electronics*, vol. 34, no. 9, pp. 8317-8328, Sept. 2019, doi: 10.1109/TPEL.2018.2886072.
- [169] D. Ma, W. Chen and X. Ruan, "A Review of Voltage/Current Sharing Techniques for Series– Parallel-Connected Modular Power Conversion Systems," in *IEEE Transactions on Power Electronics*, vol. 35, no. 11, pp. 12383-12400, Nov. 2020, doi: 10.1109/TPEL.2020.2984714.
- [170] J. -S. Bae, T. -H. Kim, S. -H. Son, H. -S. Kim, C. -H. Yu and S. -R. Jang, "Series Stacked Modular DC–DC Converter Using Simple Voltage Balancing Method," in *IEEE Transactions on Power Electronics*, vol. 36, no. 3, pp. 2471-2475, March 2021, doi: 10.1109/TPEL.2020.3015103.
- [171] J. Duan, D. Zhang, L. Wang, Z. Zhou and Y. Gu, "A Building Block Method for Input-Series-Connected DC/DC Converters," in *IEEE Transactions on Power Electronics*, vol. 36, no. 3, pp. 3063-3077, March 2021, doi: 10.1109/TPEL.2020.3014505.
- [172] M. Nguyen, T. Duong, Y. Lim, J. Choi, D. M. Vilathgamuwa and G. R. Walker, "DC-Link Quasi-Switched Boost Inverter With Improved PWM Strategy and its Comparative Evaluation," in IEEE Access, vol. 8, pp. 53857-53867, 2020, doi: 10.1109/ACCESS.2020.2981126.
- [173] A. H. Sabry, Z. M. Mohammed, F. H. Nordin, N. H. Nik Ali and A. S. Al-Ogaili, "Single-Phase Grid-Tied Transformer less Inverter of Zero Leakage Current for PV System," in IEEE Access, vol. 8, pp. 4361-4371, 2020,

List of publication

Journal Papers

- A. Shawky, M. A. Sayed, T. Takeshita, "Analysis and Experimentation of Modular Differential Inverter Utilizing SEPIC Modules and SIC Devices for Grid Connected Application" IEEJ Journal of industry Applications, Vol 9, no 5, pp. 573-583, Sept 2020.
- A. Shawky, T. Takeshita, M. A. Sayed, "Single-Stage Three-phase Grid-Tied Isolated SEPIC-Based Differential Inverter With Improved Control and Selective Harmonic Compensation" IEEE access, Vol 8, pp. 147407-147421, 2020.
- A. Shawky, M. Ahmed, M. Orabi, and A. E. Aroudi, "Classification of three-phase gridtied micro inverters in photovoltaic applications," Energies, vol. 13, no. 11, p. 2929, Jun. 2020.

Conference Proceeding Papers

- A. Shawky, M. A. Sayed, and T. Takeshita, "Selective harmonic compensation of three phase grid tied SEPIC based differential inverter," 2019 IEEE Applied Power Electronics Conference and Exposition (APEC), Anaheim, CA, USA, 2019, pp. 396-403, 2019.
- A. I. M. Ali, M. A. Sayed, A. Shawky and T. Takeshita, "Efficient Single-Stage Three-Phase Isolated Differential-Based Flyback Inverter with Selective Harmonic Compensation Strategy for Grid-Tied Applications," 2020 IEEE Applied Power Electronics Conference and Exposition (APEC), New Orleans, LA, USA, 2020, pp. 1778-1785, doi: 10.1109/APEC39645.2020.9124255.